Macaachusette institute of Technology, Castleitige, Macaachus Laboratory for Castleitiges



A.

21 October 1987

This work has been generatively extended to get by the Advanced Research Reducts Apericy of the Department of Colores and the second s

This empty page was substituted for a blank page in the original document.

Table of Contents

I CALLANDER AS MANDAGE

1. Overview	3
1.1. Objects and Variables	3
1.2. Assignment and Calls	4
1.3. Type Correctness	4
1.4. Rules and Guidelines	4
1.5. Program Structure	5
2. Concepts for Distributed Programs	7
2.1. Guardians	7
2.2. Actions	8
2.2.1. Nested Actions	8
2.2.2. Atomic Objects and Atomic Types	9
2.2.3. Netted Topactions	11
2.3. Henry: 2.4. Transmissible Types	11
2.4. Hananapana Types 2.5. Ombane	12
2.6. Deedlocke	13
2 Environment	15
	15
3.1. T he Library 2.2. Independence of Guardian Images	10
2.2. Riceptingence of Guerdian Integral 2.3. Guerdian Creation	15
3.4. The Catalog	15
4. Notation	17
5 i exicel Considerations	19
5.1 Beenved Words	10
5.2. kientifiere	19
5.3. Literais	20
5.4. Operators and Punctuation Tokens	20
5.5. Comments and Other Separators	20
6. Types, Type Generators, and Type Specifications	21
6.1. Type Inclusion	22
6.2. The Sequential Built-in Types and Type-generators	22
6.2.1. Null	22
6.2.2. Bool	22
6.2.3. Int	22
	23
0.2.3. UNM 6.3.6. String	40 24
6.2.6. Suary 6.2.7 Any	24
6.2.8. Sequence Types	25
6.2.9. Array Types	25
6.2.10. Structure Types	26
6.2.11. Record Types	27
6.2.12. Oneof Types	28
6.2.13. Variant Types	28
6.2.14. Procedure and iterator Types	29
6.4. Outputtion Theorem	- 30
0.4. Gueranan i ypes 6.5. Handler and Creater Types	31
	34

I

A MARKET AND A DATE OF A D

6.7. Mutex 33 6.8. Node 34 6.9. Other Type Specifications 34 7. Scopes, Declarations, and Equates 35 7.1. Scopes, Declarations, and Equates 35 7.1. Variables 35 7.1. Variables and Constants 36 7.2. Declarations 36 7.2. Locisantics 37 7.2. Constant Expressions 38 8. Assignment and Calls 39 8.1. Assignment 39 8.1. Semperations 39 8.1. Semperatics of Handler Calls 41 8.3. Semantics of Creator Calls 44 8.4. Creator Calls 44 8.4. Sector Calls 47 9. Expressions 47 9. Expressions 47 9. Expressions 47 9. Expressions 47 9. Procedure, Iterator, and Creator Names 48 9. Procedure, Iterator, and Creator Names 48 9. Procedure Calls 50 9.1. Uterate Module References 50 9.1. Creator Calls		6.6. Image	32
6.8. Node 34 6.9. Other Type Specifications 34 7. Scopies, Declarations, and Equates 35 7.1. Scoping Units 35 7.1.1. Variables 36 7.2. Declarations 36 7.2. Declarations 36 7.2. Declarations 37 7.2. Abdreviations for Types 38 7.2.2. Constant Expressions 38 8. Assignment and Calls 39 8.1. Assignment 39 8.1. Signment 39 8.1. Sequence Calls 40 8.1. Semantics of Handler Calls 40 8.2. Local Calls 40 8.3. Handler Calls 41 8.4. Tester Calls 43 8.4. Tester Calls 44 8.4. Tester Calls 47 9. Expressions 47 9. Expressions 47 9. Procedure, Iterator, and Creator Names 48 9. Procedure, Iterator, and Creator Names 49 9. Procedure, Resord, and Atomic Array Constructors 51 <		6.7. Mutex	33
6.9. Other Type Specifications 34 7. Scopes, Declarations, and Equates 35 7.1. Scoping Units 35 7.1. Variables 36 7.1. Variables 36 7.2. Declarations 36 7.2. Lociarations 36 7.2. Lociarations 37 7.2. Abbreviations for Types 38 7.2. Constant Expressions 38 8. Assignment and Calls 39 8.1. Assignment 39 8.1. Assignment 39 8.1. Semantics of Handler Calls 41 8.3. Handler Calls 41 8.4. Creator Calls 44 8.4. Testor Calls 43 8.4. Creator Calls 44 8.4. Testor Calls 47 9.3. Handler Calls 47 9.4. Literals 47 9.5. Equate Module Reference 47 9.6. Sold 46 9.7. Procedure, Reference 47 9.8. Procedure Calls 50 9.9. Procedure Calls 50 9.1. Literals 50 9.3. Procedure Calls 50 9.4. Equated Identifiers 50 9.5. Sold 50 9.1. Creater Calls 50 9.1. Creater Calls <		6.8. Node	34
7. Scopes, Declarations, and Equates 35 7.1. Scoping Units 35 7.1. Scoping Units 36 7.1.1. Variables 36 7.1.2. Declarations 36 7.1.2. Declarations 36 7.2.1. Abbraviations for Types 38 7.2.1. Abbraviations for Types 38 7.2.2. Constant Expressions 39 8.1. Assignment 39 8.1.1. Simple Assignment 39 8.1.1. Simple Assignment 39 8.2. Local Calls 41 8.3.1. Semantics of Handler Calls 41 8.4. Creator Calls 44 8.4. Testion Calls 44 8.4. Testion Calls 47 9. Expressions 47 9. Procedure, Iterator, and Creator Names 48 9. Procedure, Iterator, and Creator Names 48 9. Procedure, Iterator, and Creator Names 49 9. Procedure Calls 50 9.1. Creator Calls 50 9.1. Creator Calls 50 9.1. Creator Calls 50 9.1. Expression 51 9.1. Structure, Neco		6.9. Other Type Specifications	34
7.1. Scoping Units 35 7.1.1. Variables 36 7.1.2. Declarations 36 7.2.Equates and Constants 37 7.2.1. Abbreviations for Types 38 7.2.2. Constant Expressions 38 8. Assignment and Calls 39 8.1. Assignment 39 8.1.1. Simple Assignment 39 8.1.2. Multiple Assignment 39 8.1.2. Multiple Assignment 39 8.1.3. Bemantics of Handler Calls 40 8.3. Handler Calls 40 8.3. Handler Calls 41 8.3.1. Semantics of Creator Calls 44 8.4. Creator Calls 44 8.4.1. Semantics of Creator Calls 47 9. Expressions 47 9.1. Literals 47 9.2. Variables 47 9.3. Equate Module Reference 47 9.4. Equate Module Reference 47 9.5. Sold 48 9.7. Procedure, Narator, and Creator Names 48 9.8. Bind 48 9.9. 10. Intentifiere 49 9.11. Creator Calls 50	7.	Scopes, Declarations, and Equates	35
7.11. Variables 36 7.12. Declarations 36 7.2. Equates and Constants 37 7.2.1. Abbreviations for Types 38 7.2.2. Constant Expressions 38 8. Assignment and Calls 39 8.1. Assignment 39 8.1.1. Simple Assignment 39 8.1.2. Multiple Assignment 39 8.1.3. Multiple Assignment 39 8.1.4. Semantics of Handler Calls 41 8.3.1. Semantics of Handler Calls 43 8.4. Creator Calls 43 8.4. Creator Calls 44 8.4.1. Semantics of Creator Calls 44 9. Expressions 47 9.1. Literais 47 9.2. Variables 47 9.3. Parametera 47 9.4. Equated Identifiere 47 9.5. Solt 48 9.6. Procedure, iterator, and Creator Names 48 9.8. Bind 48 9.9. Procedure Calls 50 9.10. Handler Calls 50 9.11. Creator Calls 50 9.12. Literment Selection 51		7.1. Scoping Units	35
7.12. Declarations 36 7.2. Equates and Constants 37 7.2.1. Abbreviations for Types 38 7.2.2. Constant Expressions 38 8. Assignment and Calls 39 8.1. Assignment 39 8.1.1. Simple Assignment 39 8.1.1. Simple Assignment 39 8.1.2. Local Calls 40 8.3. Handler Calls 40 8.3. Handler Calls 41 8.3.1. Semantics of Creator Calls 44 8.4. Creator Calls 44 8.4. Creator Calls 47 9. Expressions 47 9.1. Literats 47 9.2. Variables 47 9.3. Parametera 47 9.4. Equated identifiers 47 9.5. Equate Module References 48 9.7. Procedure, Iterator, and Creator Names 48 9.8. Ind 48 9.9. Procedure Calls 50 9.10. Hendler Calls 50 9.11. Creator Calls 50 9.12. Expressions 51 9.13. Sequence Constructors 52 9.13.1. Sequ		7.1.1. Variables	36
7.2. Equates and Constants 37 7.2.1. Abbreviations for Types 38 7.2.2. Constant Expressions 39 8. Assignment and Calls 39 8.1. Assignment 39 8.1. Assignment 39 8.1.1. Simple Assignment 39 8.1.2. Multiple Assignment 39 8.1.2. Multiple Assignment 39 8.3. Handler Calls 40 8.3. Handler Calls 41 8.3.1. Semantics of Handler Calls 43 8.4.1. Semantics of Creator Calls 44 9. Expressions 47 9. Expressions 47 9.1. Literais 47 9.2. Variables 47 9.3. Parameters 47 9.4. Equate Module References 47 9.5. Self 48 9.6. Procedure, Restor, and Creator Names 48 9.8. Bind 48 9.9. Procedure Calls 50 9.10. Handler Calls 50 9.11. Creator Calls 50 9.12.1. Element Selection 51 9.13.2. Array and Atomic Array Constructors 52 <		7.1.2. Declarations	- 36
7.2.1. Abbreviations for Types 38 7.2.2. Constant Expressions 38 8. Assignment and Calls 39 8.1. Assignment 39 8.1.1. Simple Assignment 39 8.1.2. Multiple Assignment 39 8.1.3. Bernantics of Handler Calls 40 8.3. Handler Calls 41 8.3.1. Semantics of Handler Calls 43 8.4. Creator Calls 44 8.4.1. Semantics of Creator Calls 47 9. Expressions 47 9. Self		7.2. Equates and Constants	37
7.2.2. Constant Expressions 38 8. Assignment and Calls 39 8.1. Assignment 39 8.1.1. Simple Assignment 39 8.1.2. Multiple Assignment 39 8.1.2. Multiple Assignment 39 8.1.3. Semantics of Handler Calls 41 8.3. Handler Calls 41 8.3. Semantics of Handler Calls 44 8.4. Creator Calls 44 8.4.1. Semantics of Creator Calls 44 8.4.1. Semantics of Creator Calls 47 9. Expressions 47 9.1. Literals 47 9.2. Variables 47 9.3. Parameters 47 9.4. Equate Identifiere 47 9.5. Equate Module References 48 9.8. Bind 48 9.9. Procedure Calls 50 9.10. Handler Calls 50 9.11. Creator Calls 50 9.12.1. Element Selection 51 9.12.2. Component Selection 51 9.13.3. Structure, Record, and Atomic Record Constructors 52 9.13.4. Prefix and Inftx Operatore 53 9.15. Cand a		7.2.1. Abbreviations for Types	38
8. Assignment and Calls 39 8.1. Assignment 39 8.1.1. Simple Assignment 39 8.1.2. Multiple Assignment 39 8.3. Local Calls 40 8.3. Handler Calls 41 8.3.1. Semantics of Handler Calls 43 8.4. Creator Calls 43 8.4. Creator Calls 44 9. Expressions 47 9.1. Literals 47 9.2. Variables 47 9.3. Equated Identifiers 47 9.4. Equated Identifiers 47 9.5. Equate Module References 48 9.8. Bind 48 9.9. Procedure Calls 50 9.10. Handler Calls 50 9.11. Creator Calls 50 9.12. Creator Calls 50 9.13. Sequence Constructors 52 9.13. Structure, Record, and Atomic Record Constructors 52 9.13. Structure, Record, and Atomic Record Constructors 53 9.13. Structure, Record, and Atomic Record Constructors 53 9.13. Cand and Cor 54 9.14. Preix and Infix Operatore 55 9		7.2.2. Constant Expressions	38
8.1. Assignment 39 8.1.1. Simple Assignment 39 8.1.2. Multiple Assignment 39 8.2. Local Calls 40 8.3. Handler Calls 40 8.3.1. Semantics of Handler Calls 41 8.4.1. Semantics of Creator Calls 44 8.4. Creator Calls 44 8.4.1. Semantics of Creator Calls 47 9. Expressions 47 9.1. Literals 47 9.2. Variables 47 9.3. Parameters 47 9.4. Equate Module References 47 9.5. Self 48 9.7. Procedure, Iterator, and Creator Names 48 9.8. Dind 48 9.9. Procedure Calls 50 9.10. Handler Calls 50 9.11. Creator Calle 51 9.12. Element Selection 51 9.13.1. Sequence Constructors 52 9.13.1. Sequence Constructors 52 9.13.2. Array and Atomic Array Constructors 52 9.13.3. Structure, Record, and Atomic Record Constructors 52 9.13.2. Array and Atomic Array Constructors 53	8.	Assignment and Calls	39
8.1.1. Simple Assignment 39 8.1.2. Multiple Assignment 39 8.1.2. Multiple Assignment 39 8.2. Local Calls 40 8.3.1. Semantics of Handler Calls 41 8.3.1. Semantics of Creator Calls 44 8.4. Creator Calls 43 8.4.1. Semantics of Creator Calls 44 8.4.1. Semantics of Creator Calls 47 9. Expressions 47 9.1. Literals 47 9.2. Variables 47 9.3. Parameters 47 9.4. Equated Identifiers 47 9.5. Self 48 9.7. Procedure, iterator, and Creator Names 48 9.8. Bind 48 9.9. Procedure Calls 50 9.10. Handler Calls 50 9.11. Creator Calls 51 9.12. Demonst Selection 51 9.13. Sequence Constructors 52 9.13.1. Sequence Constructors 52 9.13.2. Array and Atomic Array Constructors 52 9.13.3. Structure, Record, and Atomic Record Constructors 52 9.14. Prefix and Infut Operators 53		8.1. Assignment	39
8.1.2. Multiple Assignment 39 8.2. Local Calls 40 8.3. Handler Calls 41 8.3.1. Semantics of Handler Calls 43 8.4. Creator Calls 43 8.4.1. Semantics of Creator Calls 44 9. Expressions 47 9.1. Literals 47 9.2. Variables 47 9.3. Parameters 47 9.4. Equated Identifiers 47 9.5. Equate Module References 47 9.5. Equate Module References 47 9.6. Self 48 9.7. Procedure, Iterator, and Creator Names 48 9.8. Bind 48 9.9. Procedure Calls 50 9.10. Handler Calls 50 9.11. Creator Calls 51 9.12. Element Selection 51 9.12. Element Selection 51 9.13.1. Sequence Constructors 52 9.13.2. Array and Atomic Array Constructors </td <td></td> <td>8.1.1. Simple Assignment</td> <td>39</td>		8.1.1. Simple Assignment	39
8.2. Local Calis 40 8.3. Handler Calis 41 8.3.1. Semantics of Handler Calis 43 8.4. Creetor Calis 44 8.4.1. Semantics of Creator Calis 44 8.4.1. Semantics of Creator Calis 44 9. Expressions 47 9.1. Literals 47 9.2. Variables 47 9.3. Parameters 47 9.4. Equated Identifiers 47 9.5. Self 48 9.7. Procedure, Iterator, and Creator Names 48 9.8. Bind 48 9.9. Procedure Calis 50 9.10. Handler Calis 50 9.11. Creator Calis 50 9.12.1. Element Selection 51 9.12.2. Component Selection 51 9.13.1. Sequence Constructors 52 9.13.1. Sequence Constructors 52 9.13.1. Sequence 54 9.15. Cand and Cor 54 9.16. Creator and Atomic Record Constructors 52 9.13.1. Sequence 54 9.13.1. Secondates 57 10. Statements 55 <t< td=""><td></td><td>8.1.2. Multiple Assignment</td><td>39</td></t<>		8.1.2. Multiple Assignment	39
8.3. Handler Calls 41 8.3. I. Semantics of Handler Calls 43 8.4. Creator Calls 44 8.4. I. Semantics of Creator Calls 44 9. Expressions 47 9. Expressions 47 9.1. Literals 47 9.2. Variables 47 9.3. Parameters 47 9.4. Equated Identifiers 47 9.5. Equate Module References 47 9.6. Self 48 9.7. Procedure, iterator, and Creator Names 48 9.8. Bind 48 9.9. Procedure Calls 50 9.10. Handler Calls 50 9.11. Creator Calls 50 9.12. Selection Operations 51 9.12. Element Selection 51 9.12. Element Selection 51 9.13. Sequence Constructors 52 9.13.1. Sequence Constructors 52 9.13.2. Array and Atomic Array Constructors 52 9.13.3. Structure, Record, and Atomic Record Constructors 52 9.14. Prefix and Infix Operators 53 9.15. Cand and Cor 54 9.16. Precede		8.2. Local Calls	40
8.3.1. Semantics of Handler Calls438.4. Creator Calls448.4.1. Semantics of Creator Calls449. Expressions479.1. Literals479.2. Variables479.3. Parameters479.4. Equated Identifiers479.5. Equate Module References479.6. Self489.7. Procedure, Iterator, and Creator Names489.8. Bind489.9. Procedure Calls509.10. Handler Calls509.11. Creator Calls509.12. Element Selection519.12.1. Element Selection519.13.2. Component Selection519.13.3. Structure, Record, and Atomic Record Constructors529.13.4. Prefix and Infix Operators529.13.5. Cand and Cor549.17. Up and Down5510. Statements5710.2. Update Statements5810.3. Block Statement5810.4. Fork Statement5810.4. Fork Statement58		8.3. Handler Calls	41
8.4. Creator Calls 44 8.4.1. Semantics of Creator Calls 44 8.4.1. Semantics of Creator Calls 47 9. Expressions 47 9.1. Literals 47 9.2. Variables 47 9.3. Parameters 47 9.4. Equated Identifiers 47 9.5. Self 48 9.7. Procedure, Iterator, and Creator Names 48 9.8. Bind 48 9.9. Procedure Calls 50 9.10. Handler Calls 50 9.10. Handler Calls 50 9.11. Creator Calls 50 9.12. Selection Operations 51 9.12. Element Selection 51 9.12. Component Selection 51 9.13.1. Sequence Constructors 52 9.13.2. Array and Atomic Array Constructors 52 9.13.3. Structure, Record, and Atomic Record Constructors 52 9.13.4. Prefix and Infix Operators 52 9.13.5. Cand and Cor 54 9.16. Precedence 54 9.17. Up and Down 55 10.2.1. Element Update 58 10.2.2. Component Updat		8.3.1. Semantics of Handler Calls	43
8.4.1. Semantics of Creator Calls 44 9. Expressions 47 9.1. Literals 47 9.2. Variables 47 9.3. Parameters 47 9.4. Equated Identifiers 47 9.5. Equate Module References 47 9.5. Solf 48 9.7. Procedure, iterator, and Creator Names 48 9.8. Bind 48 9.9. Procedure Calls 50 9.10. Handler Calls 50 9.11. Creator Calls 50 9.12. Selection Operations 51 9.12.2. Component Selection 51 9.13.1. Sequence Constructors 52 9.13.3. Structure, Record, and Atomic Record Constructors 52 9.13.4. Prefix and Infix Operators 52 9.13.5. Structure, Record, and Atomic Record Constructors 52 9.14. Prefix and Infix Operators 52 9.15. Precedence 54 9.17. Up and Down 55 10.1. Calls 57 10.2.1. Element Update 58 10.2.2. Component Update 58 10.2.3. Elorement 58 10.4. F		8.4. Creator Calls	- 44
9. Expressions 47 9.1. Literals 47 9.2. Variables 47 9.2. Variables 47 9.3. Parameters 47 9.4. Equated identifiers 47 9.5. Equate Module References 47 9.5. Seif 48 9.7. Procedure, iterator, and Creator Names 48 9.8. Bind 48 9.9. Procedure Calls 50 9.10. Handler Calls 50 9.11. Creator Calls 50 9.12. Selection Operations 51 9.12. Selection Operations 51 9.12.1. Element Selection 51 9.13.2. Component Selection 51 9.13.1. Sequence Constructors 52 9.13.2. Array and Atomic Array Constructors 52 9.13.3. Structure, Record, and Atomic Record Constructors 53 9.15. Cand and Cor 54 9.17. Up and Down 55 10. Statements 57 10.2. Update Statements 58 10.2.1. Element Update 58 10.2.2. component Update 58 10.3.8 lock Statement 58		8.4.1. Semantics of Creator Calls	44
9.1. Literais479.2. Variables479.3. Parameters479.4. Equated Identifiers479.5. Equate Module References479.6. Solf489.7. Procedure, Iterator, and Creator Names489.8. Bind489.9. Procedure Calls509.10. Handler Calls509.11. Creator Calls509.12. Selection Operations519.12.2. Component Selection519.13.2. Component Selection519.13.3. Structure, Record, and Atomic Record Constructors529.13.3. Structure, Record, and Atomic Record Constructors529.13.4. Cond and Cor549.15. Cand and Cor549.17. Up and Down5510. Statements5710.2. Update Statements5810.2.1. Element Update5810.2.2. Component Update5810.3. Block Statement5810.4. Fork Statement58	9.	Expressions	47
9.2. Variables479.3. Parameters479.4. Equated Identifiers479.5. Equate Module References479.5. Self489.7. Procedure, Iterator, and Creator Names489.8. Bind489.9. Procedure Calls509.10. Handler Calls509.11. Creator Calls519.12.1. Element Selection519.13.2. Component Selection519.13.2. Array and Atomic Array Constructors529.13.3. Structure, Record, and Atomic Record Constructors529.14. Prefix and Infix Operators529.15. Cand and Cor549.16. Precedence549.17. Up and Down5510. Statements5710.1. Calls5710.2. Update Statements5810.2.1. Element Update5810.3. Block Statement5810.4. Fork Statement5810.4. Fork Statement58		9.1. Literals	47
9.3. Parameters479.4. Equated Identifiers479.5. Equate Module References479.5. Self489.7. Procedure, Iterator, and Creator Names489.8. Bind489.9. Procedure Calls509.10. Handler Calls509.11. Creator Calls509.11. Creator Calls519.12. Selection Operations519.12.1. Element Selection519.13.1. Sequence Constructors529.13.1. Sequence Constructors529.13.2. Array and Atomic Array Constructors529.14. Prefix and Infix Operators539.15. Cand and Cor549.16. Precedence549.17. Up and Down5510. Statements5710.1. Calls5710.2. Update Statements5810.2.1. Element Update5810.3. Block Statement5810.4. Fork Statement58		9.2. Variables	47
9.4. Equated identifiers479.5. Equate Module References479.6. Self489.7. Procedure, Iterator, and Creator Names489.8. Bind489.9. Procedure Calls509.10. Handler Calls509.11. Creator Calls509.12. Selection Operations519.12.1. Element Selection519.12.2. Component Selection519.13.1. Sequence Constructors529.13.2. Array and Atomic Array Constructors529.13.3. Structure, Record, and Atomic Record Constructors539.14. Prefix and Infix Operators539.15. Cand and Cor549.17. Up and Down5510. Statements5710.1. Calls5710.2. Update Statements5810.2.1. Element Update5810.3. Block Statement5810.4. Fork Statement58		9.3. Parametera	47
9.5. Equate Module References479.6. Self489.7. Procedure, Iterator, and Creator Names489.8. Bind489.9. Procedure Calls509.10. Handler Calls509.11. Creator Calls509.11. Creator Calls519.12. Selection Operations519.12. Element Selection519.13. Constructors529.13.1. Sequence Constructors529.13.2. Array and Atomic Array Constructors529.13.3. Structure, Record, and Atomic Record Constructors539.15. Cand and Cor549.17. Up and Down5510. Statements5710.1. Calle5710.2. Update Statements5810.2.1. Element Update5810.3. Block Statement5810.4. Fork Statement5810.4. Fork Statement58		9.4. Equated Identifiers	47
9.5. Self489.7. Procedure, Iterator, and Creator Names489.8. Bind489.9. Procedure Calls509.10. Handler Calls509.11. Creator Calls509.11. Creator Calls519.12. Selection Operations519.12. Component Selection519.13. Constructors529.13.1. Sequence Constructors529.13.2. Array and Atomic Array Constructors529.13.3. Structure, Record, and Atomic Record Constructors539.14. Prefix and Infix Operators539.15. Cand and Cor549.17. Up and Down5510. Statements5710.1. Calls5710.2. Update Statements5810.2.1. Element Update5810.3. Block Statement5810.4. Fork Statement5810.4. Fork Statement58		9.5. Equite Module References	47
9.7. Procedure, terator, and Creator Names409.8. Bind489.9. Procedure Calls509.10. Handler Calls509.11. Creator Calls519.12. Selection Operations519.12.1. Element Selection519.12.2. Component Selection519.13.1. Sequence Constructors529.13.2. Array and Atomic Array Constructors529.13.3. Structure, Record, and Atomic Record Constructors529.13.4. Prefix and Infix Operators539.15. Cand and Cor549.17. Up and Down5510. Statements5710.1. Calls5710.2. Update Statements5810.2.1. Element Update5810.3. Block Statement5810.4. Fork Statement5810.4. Fork Statement5810.4. Fork Statement58		9.5. Self	45
9.0. DruceTecesoryTecesory9.9. Procedure Calls509.10. Handler Calls509.11. Creator Calls519.12. Selection Operations519.12.1. Element Selection519.12.2. Component Selection519.13.1. Sequence Constructors529.13.1. Sequence Constructors529.13.2. Array and Atomic Array Constructors529.13.3. Structure, Record, and Atomic Record Constructors529.14. Prefix and Infix Operators539.15. Cand and Cor549.17. Up and Down5510. Statements5710.1. Calls5710.2. Update Statements5810.2.1. Element Update5810.2.2. Component Update5810.4. Fork Statement5810.4. Fork Statement5810.4. Fork Statement58		9.7. Procedure, Refator, and Greator Names	40
9.10. Handler Cells509.11. Creator Cells519.12. Selection Operations519.12.1. Element Selection519.12.2. Component Selection519.13.1. Sequence Constructors529.13.1. Sequence Constructors529.13.2. Array and Atomic Array Constructors529.13.3. Structure, Record, and Atomic Record Constructors529.14. Prefix and Infix Operators539.15. Cand and Cor549.16. Precedence549.17. Up and Down5510. Statements5710.2. Lipdate Statements5810.2.1. Element Update5810.3. Block Statement5810.4. Fork Statement5810.4. Fork Statement58		9.9. Drocedure Celle	50
9.11. Creator Calls519.12. Selection Operations519.12.1. Element Selection519.12.1. Element Selection519.12.2. Component Selection519.13.1. Sequence Constructors529.13.1. Sequence Constructors529.13.2. Array and Atomic Array Constructors529.13.3. Structure, Record, and Atomic Record Constructors529.14. Prefix and Infix Operators539.15. Cand and Cor549.16. Precedence549.17. Up and Down5510. Statements5710.1. Calls5710.2. Update Statements5810.2.1. Element Update5810.3. Block Statement5810.4. Fork Statement5810.4. Fork Statement5810.4. Fork Statement5810.4. Fork Statement58		9.10. Handler Calle	50
9.12. Selection Operations519.12.1. Element Selection519.12.2. Component Selection519.13.2. Constructors529.13.1. Sequence Constructors529.13.2. Array and Atomic Array Constructors529.13.3. Structure, Record, and Atomic Record Constructors529.14. Prefix and Infix Operators539.15. Cand and Cor549.16. Precedence549.17. Up and Down5510. Statements5710.1. Calls5710.2. Update Statements5810.2.1. Element Update5810.3. Block Statement5810.4. Fork Statement58		9.11. Creator Calls	51
9.12.1. Element Selection519.12.2. Component Selection519.13. Constructors529.13.1. Sequence Constructors529.13.2. Array and Atomic Array Constructors529.13.3. Structure, Record, and Atomic Record Constructors529.14. Prefix and Infix Operators539.15. Cand and Cor549.16. Precedence549.17. Up and Down5510. Statements5710.1. Calls5710.2. Update Statements5810.2.1. Element Update5810.3. Block Statement5810.4. Fork Statement5810.4. Fork Statement58		9.12. Selection Operations	51
9.12.2. Component Selection519.13. Constructors529.13.1. Sequence Constructors529.13.2. Array and Atomic Array Constructors529.13.3. Structure, Record, and Atomic Record Constructors529.14. Prefix and Inflx Operators539.15. Cand and Cor549.16. Precedence549.17. Up and Down5510. Statements5710.1. Calls5710.2. Update Statements5810.2.1. Element Update5810.3. Block Statement5810.4. Fork Statement58		9.12.1. Element Selection	51
9.13. Constructors529.13.1. Sequence Constructors529.13.2. Array and Atomic Array Constructors529.13.3. Structure, Record, and Atomic Record Constructors529.14. Prefix and infix Operators539.15. Cand and Cor549.16. Precedence549.17. Up and Down5510. Statements5710.1. Calls5710.2. Update Statements5810.2.1. Element Update5810.3. Block Statement5810.4. Fork Statement58		9.12.2. Component Selection	51
9.13.1. Sequence Constructors529.13.2. Array and Atomic Array Constructors529.13.3. Structure, Record, and Atomic Record Constructors529.14. Prefix and infix Operators539.15. Cand and Cor549.16. Precedence549.17. Up and Down5510. Statements5710.1. Calls5710.2. Update Statements5810.2.1. Element Update5810.3. Block Statement5810.4. Fork Statement58		9.13. Constructors	- 52
9.13.2. Array and Atomic Array Constructors529.13.3. Structure, Record, and Atomic Record Constructors529.14. Prefix and Infix Operators539.15. Cand and Cor549.16. Precedence549.17. Up and Down5510. Statements5710.1. Calls5710.2. Update Statements5810.2.1. Element Update5810.3. Block Statement5610.4. Fork Statement58		9.13.1. Sequence Constructors	52
9.13.3. Structure, Record, and Atomic Record Constructors529.14. Prefix and infix Operators539.15. Cand and Cor549.16. Precedence549.17. Up and Down5510. Statements5710.1. Calls5710.2. Update Statements5810.2.1. Element Update5810.3. Block Statement5810.4. Fork Statement58		9.13.2. Array and Atomic Array Constructors	52
9.14. Prenx and mix Operators539.15. Cand and Cor549.16. Precedence549.17. Up and Down5510. Statements5710.1. Calls5710.2. Update Statements5810.2.1. Element Update5810.2.2. Component Update5810.3. Block Statement5810.4. Fork Statement58		9.13.3. Structure, Record, and Atomic Record Constructors	52
9.16. Precedence549.17. Up and Down5510. Statements5710.1. Calls5710.2. Update Statements5810.2.1. Element Update5810.2.2. Component Update5810.3. Block Statement5810.4. Fork Statement58		9.14. Ment and mit Operators 9.15. Cond and Cor	
9.17. Up and Down 55 10. Statements 57 10.1. Calls 57 10.2. Update Statements 58 10.2.1. Element Update 58 10.2.2. Component Update 58 10.3. Block Statement 56 10.4. Fork Statement 56		9.13. Jenni Englovi 9.16. Dessedence	54 84
10. Statements5710.1. Calls5710.2. Update Statements5810.2.1. Element Update5810.2.2. Component Update5810.3. Block Statement5610.4. Fork Statement58		9.17 Lin and Down	54
10. Statements5710.1. Calls5710.2. Update Statements5810.2.1. Element Update5810.2.2. Component Update5810.3. Block Statement5810.4. Fork Statement58	41	Statemente	57
10.1. Cans5710.2. Update Statements5810.2.1. Element Update5810.2.2. Component Update5810.3. Block Statement5810.4. Fork Statement58	1(7. Juli 1119113	3/
10.2. Opciate Statements 58 10.2.1. Element Update 58 10.2.2. Component Update 58 10.3. Block Statement 58 10.4. Fork Statement 58		10.1. Calis 10.2. Undete Statemente	57
10.2.1. Extrate Optime3610.2.2. Component Update5810.3. Block Statement5810.4. Fork Statement58		IV.2. Update Statements 10.2.1 Element Lindets	00 20
10.3. Block Statement 58 10.4. Fork Statement 58		10.2.1. Extrement Update 10.2.2. Component Lindate	10 20
10.4. Fork Statement 58		10.3. Block Statement	
		10.4. Fork Statement	58

11

 $\label{eq:constraint} \left(\left(1 + \log g_{2} \right) \right) + \left(1 + \log g_{2} \right$

121

75

	10.5. En						
	10.6. Ct	enter	State	mont			
	10.7. Le			ent 🛛			
	10.8. 1	and a state					
	10.0. YE		Constant Services				
	10.10.6				NARK		
	10.11.1	Profession and Pro-					
	10.12. 6						
	10.18.8						
	10.14.1						
	10.16.1	No. Astron	- Alberter -			a dia Managina Ind	
	18		- Bankinski		-i - Corecterio		
	18		* Signade St	is all in the second	A THE AREA BE		
	10.		Andrea an Sainte I		Yaikind		
	10.16.		der Stat				
	10.17.8						
	10.18.3		series of				
-							
	le Létéra	(15 D T T S					B
	11.1.8						
	11.2. 8						
	11.8.6			hatter and a	b		
	11.4.6		1949 (1949) (1949) (1949)				
	11.5.	and a second	ST BLADER				
	11.6. 1						
1							
12.							
	12.1. 01						
		(de la color)					
		a an					
			3.57 1695				
			- ⁵⁴ 666943	an an an firm			
		and the second	- Section				
			5				
1	I. Gran	Contraction of the					
	12.1 10	-			а. Б. с. с.		
	13.3 6	લક્ષણી પંચ છ					
	19.9.8		E.				
		kan na selari	4. A. P.				
		in the	n generalis	ويتحديد الب	تحدد المغرية	-	
			a a a a a a a a a a a a a a a a a a a	er was not			
			-sér ing				
74	I. THIN	and a statistic		V			
	14.1.1		Sec. 1				
	والمستعد المراجع	1	A	NO. OF THE OWNER	CALCENCE .		

15. Adamste Type

S.

÷.

16.1. Automotive Symphone Income and Recovery 14.4 November 16.5 Constants

15.4. A Proceedation for Alconicity

15.5. Commuting Operations	102
Annendiy I. Suntey	107
Appendix I. Syntax	107
Appendix II. Built-in Types and Type Generators	119
II.1. Null	120
II.2. Nodes	120
	121
ii.4. Hitiogers II.5. Doolo	141
II.S. rodae II.S. Charactere	125
il.7. Strings	126
li.8. Sequences	128
II.9. Arrays	130
II.10. Atomic Arrays	133
II.11. Structs	138
II.12. Records	139
II.13. Atomic Records	141
li.14. Oneofs	143
II.15. Variants	144
II.16. Atomic variants	140
II.17. Procedures and Renators	140
II. 19. Nationals and Cranics II 19. Anve	150
II 20 Image	150
H.21. Matexes	151
Appendix III. Rules and Guidelines for Using Argus	153
III 1 Serielizebility and Actions	153
ill.2. Actions and Excentions	153
III.3. Stable Variables	154
III.4. Transmission and Transmissibility	154
III.5. Mutex	154
III.6. User-Defined Atomic Objects	156
III.7. Subordinate Where Clauses	157
Appendix IV. Changes from CLU	159
IV.1. Exception Handling	159
IV.2. Type Any	159
IV.3. Built-in Types	159
IV.4. Type inclusion	160
IV.5. Where Clauses	160
IV.6. Uninktialized Variables	160
IV.7. LOXICAL CHANGES	160
IV.5. Input/Output Changes	160
Index	161

an an Alexandre age

List of Figures

Figure 2-1:	Locking and Version Management Rules for a Subaction S, on Object X	10
Figure 13-1:	Spooler Guardian	91
Figure 14-1:	Partial implementation of table.	95

List of Tables

List of Tables

Table 5-1:	Pleesived Words		
Table 6-2:			
Table 6-1:			
Table 9-1:			i -
			- -
		ter per la companya de la companya d	

Vii

This empty page was substituted for a blank page in the original document.

Guide to the Manual

This document serves both as a reference manual and as an introduction to Argus. Sections 1 through 3 present an overview of the language. These sections highlight the essential features of Argus. Sections 4 through 15 and the appendices form the reference manual proper. These sections describe each aspect of Argus in detail, and discuss the proper use of various features. Appendices I and II provide summarizes of Argus's syntax and data types. Appendix III summarizes some of the pragmatic rules for using Argus.

Since Argus is based on the programming language CLU, the reader is expected to have some familiarity with CLU. Those readers needing an introduction to CLU might read Liskov, B. and Guttag, J., *Abstraction and Specification in Program Development* (MIT Press, Cambridge, 1986). A shorter overview of CLU appears in the article Liskov, B., *et al.*, "Abstraction Mechanisms in CLU" (*Comm. ACM*, volume 20, number 8 (Aug. 1977), pages 564-576). Appendix IV summarizes the changes made to Argus that are not upward compatible with CLU.

An overview and rationale for Argus is presented in Liskov, B. and Scheifler, R., "Guardians and Actions: Linguistic Support for Robust, Distributed Programs" (*ACM Transactions on Programming Languages and Systems*, volume 5, number 3 (July 1983), pages 381-404).

The Preliminary Argus Reference Manual appeared as Programming Methodology Group Memo 39 in October 1983. Since that time several new features have been added to the language; the most significant of these are closures (see Section 9.8), a fork statement (see Section 10.4), equate modules (see Section 12.4), and a more flexible instantiation mechanism (see Section 12.6). An earlier version of this document appeared as Programming Methodology Group Memo 54 in March 1987; this version is essentially identical, except that the locking policy for the built-in type generator atomic_array has been simplified.

We would greatly appreciate receiving comments on both the language and this manual. Comments should be sent to: Professor Barbara Liskov, Laboratory for Computer Science, Massachusetts Institute of Technology, 545 Technology Square, Cambridge, MA 02139.

The authors thank all the members of the Programming Methodology group at MIT for their help and suggestions regarding the language and this manual, with special thanks going to Elliot Kolodner, Deborah Hwang, Sharon Perl, and the authors of the *CLU Reference Manual*.

Though her unhappy rival was hers to keep Queen Juno also had a troubled mind: What would Jove turn to next? Better, she thought, To give the creature to Arestor's son, The frightful Argus whose unnatural head Shone with a hundred eyes, a perfect jailer For man or beast: the hundred eyes took turns At staring wide awake in pairs, and two At falling off to sleep; no matter how or Where he stood he gazed at lo; even when His back was turned, he held his prisoner In sight and in his care.

> --- Ovid, *The Metamorphoses*, Book 1 Translated by H. Gregory The Viking Press, Inc., New York, 1958

1. Overview

Argue is an experimental language/system designed to expend the construction and execution of distributed programs. Argue is intended to arguest out a construction distributed by a distributed program. The construction distributed base designed to a from being implemented by a distributed program. The construction distributed base designed to a make use of on-the data that could make a construct the construction distributed base distributed and they make use of on-the data that could make a construct the construction distributed base of another and they make use of on-the data that could make a construct the construction of the provide construction of the construction are they provide construct and the construction that can not be a set of a set of a could be a set of a construction of the could be a set of a could be a set of a

Argue is based on CLU. It is largely an extension of CLU, but there are number of differences (see Appendix IV). Like CLU, Argue provides processes for provided description, for some for control abstraction, and quarters for data statements. It addition, description that because the many substantian and control access to one or more many materians. These are discribed in the data in Burton 1. As in CLU, provides equate mathema as a convenient way to use to constant the firster 12.4). As in CLU, modules may be commutationed, as that a grade mathematication and of related statements.

1.1. Objects and Variables

The semantics of Argue destruits abjects and variables. Gainste are the data antible that are dreated and manipulated by applications. Manufacture are the names applications to active to detects.

Every object has a type that alternationality his bahaviar. A type dollars a set of primitive operations to create and manipulate objects of that type.

An object may refer to other objects or even to instit. It is also provide for an object to be referred to or shared by several objects. Objects outsi independently discussions and bester inconstitutes.

There are several categories of chippels in Agens. America ander is called a mutable object. A mutable dated has date that at sa Śrawski i chject's identity. A mulable citizet can then an state is incomparable from the adultities. Any bin are alomic I they provide gynationadaution and en h Chellen 2.2.2). Objects are transmissible if they can be a the ada (see Section 2.4). Since quardian, hundler, are these objects are said to be global oblicate. i a procedures, can only be shared within a single guar

Variables are names used in programs to denote particular delayer procession time. It is possible for two variables to denote the same object. Variables are anti-delayer permet he denoted by other variables or referred to by objects.

Variables in guardian modules can be declared to be apple. The objects densied by stable variables survive crashes (see Section 2) and are called apple apples.

Overview

1.2. Assignment and Calls

The basic events in Argus are assignments and calls. The assignment statement x := E, where x is a variable and E is an expression, causes x to denote the object resulting from the evaluation of E. The object is not copied.

A call involves passing argument objects from the caller to the called routine and returning result objects from the routine to the caller. For local calls, argument passing is defined in terms of assignment, or *call by sharing*; for remote calls, call by value is used. In a local call, the formal arguments of a routine are considered to be local variables of the routine and are initialized, by assignment, to the objects resulting from the evaluation of the argument expressions. In a remote call (see Section 2.3), a copy of the objects resulting from the evaluation of the argument expressions is made and transmitted to the called handler or creator (see Section 2.4). These copies are then used to initialize the formal arguments as before. Local objects are shared between the caller and a called procedure or iterator, but local objects are never shared between the caller and a called handler or creator.

1.3. Type Correctness

The declaration of a variable specifies the type of the objects which the variable may denote. In a legal assignment statement, x := E, the type of the expression E must be *included* in the type of the variable x. Type inclusion is essentially equality of types (see Section 12.6), except for routine types. (A routine type with fewer exceptions is included in an otherwise identical routine type with more exceptions. See Section 6.1 for details.)

Argus is a type-safe language, in that it is not possible to treat an object of type T as if it were an object of some other type S (the one exception is when T is a routine type and S includes T). The type safety of Argus, plus the restriction that only the code in a cluster may convert between the abstract type and the concrete representation (see Section 12.3), ensure that the behavior of an object can be characterized completely by the operations of its type.

1.4. Rules and Guidelines

Throughout this manual, and especially in the discussions of atomicity, there are pragmatic rules and guidelines for the use of the language. Certain properties that the language would like to guarantee, for example that atomic actions are really atomic, are difficult or impossible for the language to guarantee completely. As in any useful programming language, programmers have enough rope to hang themselves. The rules and guidelines noted throughout the manual (and collected in Appendix III) try to make the responsibilities of the language and the programmer clear.

1.5. Program Structure

An Argus distributed application consists of one or more guardians, defined by guardian modules. Guardian modules may in turn use all the other kinds of modules that Argus provides. Argus programmers may also write single-machine programs with no stable state, using Argus as essentially a "concurrent CLU." Such programs may be used to start up multi-guardian applications. Each module is a separate textual unit, and is compiled independently of other modules. Compilation is discussed in Section 3.

2. Concepts for Distributed Programs

In this chapter we present an overview of the new concepts in Argus that support distributed programs. In Section 2.1, we discuss *guardians*, the module used in Argus to distribute data. Next, in Section 2.2, we present *atomic actions*, which are used to cope with concurrency and failure. In Section 2.3 we describe *remote calls*, the inter-guardian communication mechanism. In Section 2.4 we discuss transmissible types: types whose objects can be sent as arguments or results of remote calls. Finally, in Section 2.4 we discuss orphans.

المجرة المنافي المراجع الماجعين والمراجع

2.1. Guardians

Distributed applications are implemented in Argus by one or more modules called *guardians*. A guardian abstraction is a kind of data abstraction, but it differs from the data abstractions supported by clusters (as found in CLU). In general, data abstractions consist of a set of operations and a set of objects. In a cluster the operations are considered to belong to the abstraction as a whole. However, guardian instances are objects and their handlers are their operations. Guardian abstraction is similar to the data abstractions in Simula and Smalltalk-80; guardians are like class instances.

A node is a single physical location, which may have multiple processors. A guardian instance resides at a single node, although a node may support several guardians. A guardian encapsulates and controls access to one or more resources, such as data or devices. Access to the protected resource is provided by a set of operations called *handlers*. Internally, a guardian consists of a collection of data objects and processes that can be used to manipulate those objects. In general, there will be many processes executing concurrently in a guardian: a new process is created to execute each handler call, processes may be explicitly created, and there may be other processes that carry out background activity of the guardian.

The data objects encapsulated by a guardian are *local*: they cannot be accessed directly by a process in another guardian. In contrast, guardians are *global* objects: a single guardian may be shared among processes at several different guardians. A process with a reference to a guardian can call the guardian's handlers, and these handlers can access the data objects inside the guardian. Handler calls allow access to a guardian's local data, but the guardian controls how that data can be manipulated.

When a node fails, it *crashes*. A crash is a "clean" failure, as opposed to a "Byzantine" failure. A guardian survives crashes of its node (with as high a probability as needed). A guardian's state consists of *stable* and *volatile* objects. When a guardian's node crashes, all processes running inside the guardian at the time of the crash are lost, along with the guardian's volatile objects, but the guardian's stable objects survive the crash. Upon recovery of the guardian's node, the guardian runs a special recovery process to reconstruct its volatile objects from its stable objects. Since the volatile objects are lost in a crash, they typically consist only of redundant data that is used to improve performance (for example, an index into a database). The persistent state of an application should be kept in stable objects.

Guardians are implemented by guardian definitions. These define:

7

- 1. The *creators*. These are operations that can be called to create new guardian instances that perform in accordance with the guardian definition.
- 2. The guardian's stable and volatile state.
- 3. The guardian's handlers.
- The background code. This is code that the guardian executes independent of any handler calls, for example, to perform some periodic activity.

5. The recover code. This is code that is executed after a crash to restore the volatile objects. Guardians and guardian definitions are discussed in Section 13.

2.2. Actions

The distributed data in an Argus application can be shared by concurrent processes. A process may attempt to examine and transform some objects from their current states to new states, with any number of intermediate state changes. Interactions among concurrent processes can leave data in an inconsistent state. Failures (for example, node crashes) can occur during the execution of a process, raising the additional possibility that data will be left in an inconsistent intermediate state. To support applications that need consistent data, Argus permits the programmer to make processes atomic.

We call an atomic process an action. Actions are atomic in that they are both serializable and recoverable. By serializable, we mean that the overall effect of executing multiple concurrent actions is as if they had been executed in some sequential order, even though they actually execute concurrently. By recoverable, we mean that the overall effect of an action is "all-or-nothing:" either all changes made to the data by the action happen, or none of these changes happen. An action that completes all its changes successfully commits; otherwise it aborts, and objects that it modified are restored to their previous states.

Before an action can commit, new states of all modified, stable objects must be written to stable storage¹: storage that survives media crashes with high probability. Argus uses a two-phase commit protocol² to ensure that either all of the changes made by an action occur or none of them do. If a crash occurs after an action modifies a stable object, but before the new state has been written to stable storage, the action will be aborted.

2.2.1. Nested Actions

Actions in Argus can be nested: an action may be composed of several *subactions*. Subactions can be used to limit the scope of failures and to introduce concurrency within an action.

An action may contain any number of subactions, some of which may be performed sequentially, some

¹Lampson, B. W., "Atomic Transactions", in *Distributed Systems-Architecture and Implementation*, Lecture Notes in Computer Science, volume 105, pages 246-265. Springer-Verlag, New York, 1981.

²Gray, J. N., "Notes on data base operating systems", in *Operating Systems, An Advanced Course*, Bayer, R., Graham, R. M., and Seegmüller, G. (editors), Lecture Notes in Computer Science, volume 60, pages 393-481. Springer-Verlag, New York, 1978.

concurrently. This structure cannot be observed from outside the action; the overall action is still atomic. Subactions appear as atomic actions with respect to other subactions of the same parent. Thus, subactions can be executed concurrently.

Subactions can commit and abort independently, and a subaction can abort without forcing its parent action to abort. However, the commit of a subaction is conditional: even if a subaction commits, aborting its parent action will abort it.

The root of a tree of nested actions is called a *topaction*. Topactions have no parent; they cannot be aborted once they have committed. Since the effects of a subaction can always be undone by aborting its parent, the two-phase commit protocol is used only when topactions attempt to commit.

In Argus, an action (e.g., a handler call) may return objects through either a normal return or an exception and then abort. The following rule should be followed to avoid violating serializability: a subaction that aborts should not return any information obtained from data shared with other concurrent actions.

2.2.2. Atomic Objects and Atomic Types

Atomicity of actions is achieved via the data objects shared among those actions. Shared objects must be implemented so that actions using them appear to be atomic. Objects that support atomicity are referred to as *atomic objects*. Atomic objects provide the synchronization and recovery needed to ensure that actions are atomic. An *atomic type* is a type whose objects are all atomic. Some objects do not need to be atomic: for example, objects that are local to a single process. Since the synchronization and recovery needed to ensure atomicity may be expensive, we do not require that all types be atomic. (For example, Argus provides all the built-in mutable types of CLU; these types are not atomic.) However, it is important to remember that atomic actions must share only atomic objects.

Argus provides a number of built-in atomic types and type generators. The built-in scalar types (null, node, bool, char, int, real, and string) are atomic. Parameterized types can also be atomic. Typically, an instance of a type generator will be atomic only if any actual type parameters are also atomic. The built-in immutable type generators (sequence, struct, and oneof) are atomic if their parameter types are atomic. In addition, Argus provides three mutable atomic type generators: atomic___array, atomic__record, and atomic_variant. The operations on these types are nearly identical to the normal array, record, and variant types of CLU. Users may also define their own atomic types (see Section 15).

The implementation of the built-in mutable atomic type generators is based on a simple locking model. There are two kinds of locks: read locks and write locks. When an action calls an operation on an atomic object, the implementation acquires a lock on that object in the appropriate mode: it acquires a write lock if it mutates the object, or a read lock if it only examines the object. The built-in types allow multiple concurrent readers, but only a single writer. If necessary, an action is forced to wait until it can obtain the appropriate lock. When a write lock on an object is first obtained by an action, the system makes a copy of the object's state in a new version, and the operations called by the action work on this version³. If, ultimately, the action commits, this version will be retained, and the old version discarded. A subaction's locks are given to its parent action when it commits. When a topaction commits, its locks are discarded and its effects become visible to other actions. If the action aborts, the action's locks and the new version will be discarded, and the old version retained (see Figure 2-1).

Figure 2-1: Locking and Version Management Rules for a Subaction S, on Object X

Acquiring a read lock:	
All holders of write locks on X must be ancestors of S.	
Acquiring a write lock:	
All holders of read and write locks on X must be ancestors of S.	
If this is the first time S has acquired a write lock on X,	
push a copy of X on the top of its version stack.	
Commit:	
S's parent acquires S's lock on X.	
If Sholds a write lock on X, then S's version becomes S's parent's vers	sion.
Abort:	
S's lock and version (if any) are discarded.	

More precisely, an action can obtain a read lock on an object if every action holding a write lock on that object is an ancestor of the requesting action. An action can obtain a write lock on an object if every action holding a (read or write) lock on that object is an ancestor. When a subaction commits, its locks are inherited by its parent and its new versions replace those of its parent; when a subaction aborts, its locks and versions are discarded (see Figure 2-1). Because Argus guarantees that parent actions never run concurrently with their children, these rules ensure that concurrent actions never hold write locks on the same object simultaneously.

The ancestors of a subaction are itself, its parent, its parent's parent, and so on; a subaction is a descendant of its ancestors. A subaction commits to the top if it and all its ancestors, including the topaction, commit. A subaction is a committed descendant of an ancestor action if the subaction and all intervening ancestors have committed. When a topaction attempts to commit, the two-phase commit protocol is used to ensure that the new versions of all objects modified by the action and all its committed descendants are copied to stable storage. After the new versions have been recorded stably, the old versions are thrown away.

User-defined atomic types can provide greater concurrency than built-in atomic types⁴. An

³This operational description (and others in this manual) is not meant to constrain implementors. However, this particular description does reflect our current implementation.

⁴An example can be found in Weihl, W. and Liskov, B., "Implementation of Peeilient, Atomic Data Types," ACM Transactions on Programming Languages and Systems, volume 7, number 2 (April 1965), pages 244-269.

implementation of a user-defined atomic type must address several issues. First, it must provide proper synchronization so that concurrent calls of its operations do not interfere with each other, and so that the actions that call its operations are serialized. Second, it must provide recovery for actions using its objects so that aborted actions have no effect. Finally, it must ensure that changes made to its objects by actions that commit to the top are recorded properly on stable storage. The built-in atomic types and the mutex type generator are useful in coping with these issues. User-defined atomic types are discussed further in Section 15.

2.2.3. Nested Topactions

In addition to nesting subactions inside other actions, it is sometimes useful to start a new topaction inside another action. Such a *nested topaction*, unlike a subaction, has no special privileges relative to its "parent"; for example, it is not able to read an atomic object modified by its "parent". Furthermore, the commit of a nested topaction is not relative to its "parent"; its versions are written to stable storage, and its locks are released, just as for normal topactions.

Nested topactions are useful for benevolent side effects that change the representation of an object without affecting its abstract state. For example, in a naming system a name look-up may cause information to be copied from one location to another, to speed up subsequent look-ups of that name. Copying the data within a nested topaction that commits ensures that the changes remain in effect even if the "parent" action aborts.

A nested topaction is used correctly if it is serializable before its "parent". This is true if either the nested topaction performs a benevolent side effect, or if all communication between the nested topaction and its parent is through atomic objects.

2.3. Remote Calls

An action running in one guardian can cause work to be performed at another guardian by calling a handler provided by the latter guardian. An action can cause a new guardian to be created by calling a creator. Handler and creator calls are *remote calls*. Remote calls are similar to local procedure calls; for example, the calling process waits for the call to return. Remote calls differ from local procedure calls in several ways, however.

First, the arguments and results of a remote call are passed by value (see below and also Section 14) rather than by sharing. This ensures that the local objects of one guardian remain local to that guardian, even if their values are used as arguments or results of remote calls to other guardians. The only objects that are passed by sharing in remote calls are the global objects: guardians, handlers, creators, and nodes.

Second, any remote call can raise the exceptions *failure* and *unavailable*. (Unlike CLU, not all local calls can raise *failure*, see Appendix IV.) The occurrence of *failure* means that the call is unlikely to ever succeed, so there is no point in retrying the call in the future. *Unevailable*, on the other hand, means that

常い間違いです。こうとうです。

the call should succeed if retried in the future, but is unlikely to succeed if retried immediately. For example, *failure* can arise because it is impossible to transmit the arguments or results of the call (see Section 14); *unavailable* can arise if the guardian being called has crashed, or if the network is partitioned.

Third, a handler or creator can be called only from inside an action, and the call runs as a subaction of the calling action. This ensures that a remote call succeeds *at most once*: either a remote call completes successfully and commits, or it aborts and all of its modifications are undone (provided, of course, that the actions involved are truly atomic). Although the effect of a remote call occurs at most once, the system may need to attempt it several times; this is why remote calls are made within actions.

2.4. Transmissible Types

Arguments and results of remote calls are passed by value. This means that the argument and result objects must be copied to produce distinct objects. Not all objects can be copied like this; those that can are called *transmissible objects*, and their types are called *transmissible types*. Only transmissible objects may be used as arguments and results of a remote call. In addition, image objects (see Section 6.6) can contain only transmissible objects. Parameterized types may be transmissible in some instances and not in others; for example, instantiations of the built-in type generators are transmissible only if their parameter types are transmissible. While guardians, creators, and handlers are always transmissible, procedures and iterators are never transmissible.

Users can define new transmissible types. For each transmissible type T the external representation type of T must be defined; this describes the format in which objects of type T are transmitted. Each cluster that implements a transmissible type T must contain two procedures, encode and decode, to translate objects of type T to and from their external representation. More information about defining transmissible types can be found in Section 14.

2.5. Orphans

An *orphan* is an action that has had some ancestor "perish" or has had the pertinent results of some relative action lost in a crash. Orphans can arise in Argus due to crashes and explicit aborts. For example, when a parent action is aborted, the active descendents it leaves behind become orphans. Crashes also cause orphans: when a guardian crashes, all active actions with an ancestor at the crashed guardian and all active actions with committed descendants that ran at the crashed guardian become orphans⁵. However, having a descendent that is an orphan does not necessarily imply that the parent is an orphan; as previously described, actions may commit or abort independently of their subactions.

Argus programmers can largely ignore orphans. Argus guarantees that orphans are aborted before

⁵Walker, E. F., "Orphan Detection in the Argus System", Massachusetts Institute of Technology, Laboratory for Computer Science, Technical Report MIT/LCS/TR-326, June 1984.

they can view inconsistent data (provided actions are written as that they only communicate through atomic data). Remote calls that fail for any reason wing the statistic lay the system, including some cases where the call action becomes an eighten due to emphasize (see Booten 6.5).

Orphane always abort. They may abort valuationly or they may be funced to abort by the run-time system; however, an orphan that is in a critical continuity descenting is statement, see Section 19.16) may not be forcibly aborted by the run-time system, assumpting sections from generation. On the other hand, the system may encourage explains (expectedly topological and provide to abort descentions by having their remote calls signal encourage.

2.6. Deedlocks

Actions in Argan programs may become developing. For example, if action A is waiting for a took that B holds and B is waiting for a took that A holds, then A and B and developing and anything implementations may provide some form of developing statements. They are not examined to do so. This is because detenting developing is difficult in a bagange with anti-fitteet starmic types, since it is not always clear when actions are "waiting" for each other.

If an implementation of Argue chooses to de detailed detailed growthatly for the built-in atomic types), it may only break deathates by deathing actions arity adapting growthats.

3 Environment

3. Environment

The Argus environment ensures complete static type checking of programs. It also supports separate compilation and the independence of guardians.

3.1. The Library

Argus modules are compiled in the context of a library that gives meaning to external identifiers and allows inter-module type checking. The Argus library contains type information about abstractions; for each abstraction, the library contains a *description unit*, or DU, describing that abstraction and its implementations. Each DU has a unique name and these names form the basis of type checking.

3.2. Independence of Guardian Images

The code run by a guardian comes from some guardian image. A guardian image contains all the code needed to carry out any local activity of the guardian; any procedure, iterator or cluster used by that guardian will be in its guardian image. Any handler calls made by the guardian, however, are carried out at the called guardian, which contains the code that performs the call. Thus a guardian is independent of the implementations of the guardians it calls and the implementation of a guardian can be changed without affecting the implementations of its clients.

3.3. Guardian Creation

When a guardian is created, it is necessary to select the guardian image that will supply the code run by the new guardian. To this end, each guardian has an associated *creation environment* that specifies the guardian images for other guardians it may create. The creation environment is a mapping from guardian types to information that can be used to select a guardian image appropriate for each kind of node. For greater flexibility, this information can be associated with particular creator objects.

3.4. The Catalog

Somehow, guardians must be able to find other guardians to call for services. A guardian usually has a reference to any guardian it creates. Also, if a guardian can call some other server guardian, it can learn about the guardians that the server "knows", because guardians can be passed in remote calls. In addition, Argus provides a built-in subsystem known by all guardians. This subsystem is called the *catalog*. The catalog provides an atomic mapping from names to transmissible objects. For example, when a new guardian is created, it can be catalogued under some well-known name, so that other guardians can find it in the future. Since we are currently experimenting with various interfaces to the catalog, we do not include an interface specification here.

{a}

4. Notation

We use an extended BNF grammar to define the syntax of Argus. The general form of a production is:

guilt of the second second

nonterminal ::= alternative

alternative ... alternative

The following extensions are used:

a ,	a list of one or more	a's separated by commas:	"a" or "	a, a " or	"a, a, a"	etc.

a sequence of zero or more a's: " " or "a" or "a a" etc.

[a] an optional a: " " or "a".

Nonterminal symbols appear in normal face. Reserved words appear in **bold** face. All other terminal symbols are non-alphabetic, and appear in normal face.

Full productions are not always shown in the body of this manual; often alternatives are presented and explained individually. Appendix I contains the complete syntax.

5. Lexical Considerations

A module is written as a sequence of tokens and separators. A *token* is a sequence of "printing" ASCII characters (values 40 octal through 176 octal) representing a reserved word, an identifier, a literal, an operator, or a punctuation symbol. A *separator* is a "blank" character (space, vertical tab, horizontal tab, carriage return, newline, form feed) or a comment. Any number of separators may appear between tokens.

5.1. Reserved Words

The following character sequences are reserved word tokens:

abort	else	leave	signals
action	elseif	mutex	stable
any	end	nil	string
array	enter	node	struct
atomic_array	equates	null	tag
atomic record	except	oneof	tagcase
atomic_variant	exit	others	tagtest
background	false	own	tagwait
begin	for	pause	terminate
bind	foreach	proc	then
bool	fork	process	topaction
break	guardian	proctype	transmit
cand	handler	real	true
char	handlertype	record	type
cluster	handles	recover	up
coenter	has	rep	variant
continue	if.	resignal	when
cor	image	return	where
creator	in	returns	while
creatortype	int	seize	with
cvt	is	self	wtag
do	iter	sequence	yield
down	itertype	signal	vields

Upper and lower case letters are not distinguished in reserved words. For example, 'end', 'END', and 'eNd' are all the same reserved word. Reserved words appear in **bold** face in this document.

5.2. Identifiers

An *identifier* is a sequence of letters, digits, and underscores (_) that begins with a letter or underscore, and that is not a reserved word. Upper and lower case letters are not distinguished in identifiers.

In the syntax there are two different nonterminals for identifiers. The nonterminal idn is used when the identifier has scope (see Section 7.1); idns are used for variables, parameters, module names, and as abbreviations for constants. The nonterminal *name* is used when the identifier is not subject to scope rules; names are used for record and structure selectors, one of and variant tags, operation names, and exceptional condition names.

5.3. Literals

There are literals for naming objects of the built-in types **null, bool, Int, real, char, and string**. Their forms are described in Appendix I.

5.4. Operators and Punctuation Tokens

The following character sequences are used as operators and punctuation tokens.

Table 5-2: Operator and Punctuation Tokens

([•	~	*	<	~<	E	
)]	\$	**	II	<=	~<=	~=	
{	:	:=	//	+	>=	~>=	&	
)	,	0	1	_	>	~>	1	
-								

5.5. Comments and Other Separators

A comment is a sequence of characters that begins with a percent sign (%), ends with a newline character, and contains only printing ASCII characters (including blanks) and horizontal tabs in between. For example:

z := a[i] + % a comment in an expression b[i]

A separator is a blank character (space, vertical tab, horizontal tab, carriage return, newline, form feed) or a comment. Zero or more separators may appear between any two tokens, except that at least one separator is required between any two adjacent non-self-terminating tokens: reserved words, identifiers, integer literals, and real literals. This rule is necessary to avoid lexical ambiguities.

6. Types, Type Generators, and Type Specifications

A type consists of a set of objects together with a set of operations used to manipulate the objects. Types can be classified according to whether their objects are mutable or immutable, and atomic or non-atomic. An *immutable* object (e.g., an integer) has a value that never varies, while the value (state) of a *mutable* object can vary over time. Objects of *atomic* types provide serializability and recovery for accessing actions. *Non-atomic* types may provide synchronization by specifying that particular operations are executed *indivisibly* on objects of the type. An operation is indivisible if no other process may affect or observe intermediate states of the operation's execution. Indivisibility properties will be described for all the built-in non-atomic types of Argus.

A type generator is a parameterized type definition, representing a (usually infinite) set of related types. A particular type is obtained from a type generator by writing the generator name along with specific values for the parameters; for every distinct set of legal values, a distinct type is obtained (see Section 12.6). For example, the array type generator has a single parameter that determines the element type; array[Int], array[real], and array[array[Int]] are three distinct types defined by the array type generator. Types obtained from type generators are called *parameterized* types or *instantiations* of the type generator; others are called *simple* types.

In Argus code, a type is specified by a syntactic construct called a *type_spec*. The type specification for a simple type is just the identifier (or reserved word) naming the type. For parameterized types, the type specification consists of the identifier (or reserved word) naming the type generator, together with the actual parameter values.

To be used as arguments or results of handler and creator calls, or as **image** objects (see Section 6.6), objects must be *transmissible*. Most of the built-in Argus types are transmissible, that is, they have transmissible objects. However, procedures and iterators are never transmissible. For type generators, transmissibility of a particular instantiation of the generator may depend upon transmissibility of any type parameters. A transmissible type provides the pseudo-operation transmit and two internal operations *encode* and *decode*. Generally, *encode* and *decode* are hidden from clients of the type. They are called implicitly during message transmission (see Section 14) and in creating and decomposing Image objects (see Section 6.6). Transmissibility is discussed further in Section 14.

Argus provides all the built-in types of CLU as well as some new types and type generators. This section gives an informal introduction to the built-in types and type generators provided by Argus. Many details are not discussed here, but a complete definition of each type and type generator is given in Appendix II.

6.1. Type inclusion

The notion of type inclusion in Argue is different from that in CLU. The type any is a type like every other type, and there is no implicit constants to type, and there is no implicit constants to type the same type inclusion rate. Type inclusion in Argue is the same spicing equality (see Beatlen 12.6), except for procedure, humber and types of arguments, and the number and types of results. Note that if a consumption are equal to the same results are the types of results. Note that if a generation more analytics the type of arguments and results. The type of the type, if we have the tolowing declarations in effect:

p : produced and allowing algorith (realised, and allowing)

Children and a start and a

then the type of p is individed in the type of p but not vice verse. Thus the automatem p := q is legal.

6.2. The Sequential Built-In Types and Types generations

In this section, we introduce the expected will be preside to the section of the section of the section concentrates on their new descriptions.

Recovery from aborted estions is trivial for immutable adjusts, along the shorted estions cannot have modified these objects. In particular the built-is and space and, have the shorted when, and adding are immutable, atomic, and transmissible. The built-is mutable type generations interface term CLU are not stomic.

6.2.1. Nuli

The type null has exactly one inventable object, represented by the iteral all, which is atomic and transmissible. See Section it. I for details.

6.2.2. Bool

The two immutable objects of type basis, with illurate true and false, supment logical truth values. The binary operations equal (-), and (ii), and or (j), are provident, as well as unary not (-). Objects of type boot are stormic and tratemicable. See Section 1.3 for details.

6.2.3. int

The type int models (a range of) the mathematical integers. The analt range is not part of the improve definition⁴. Magaze are immuchile, adminis, transmission, and the desired are written as a sequence of one or more destination (ight). (There are also additional backhold fibrate, are Appendix I.)

[&]quot;However, implementations are encouraged to provide this and other interimation about the limits of the individe types in an experience module.

The binary operations add (+), sub (-), mul (*), div (/), mod (//), power (**), max, and min are provided, as well as unary minus (-) and abs. There are binary comparison operations lt (<), le (<=), equal (=), ge (>=), and gt (>). There are two operations, from_to and from_to_by, for iterating over a range of integers. See Section II.4 for details.

6.2.4. Real

The type real models (a subset of) the mathematical real numbers. The exact subset is not part of the language definition. Reals are immutable, atomic, and transmissible, although transmission of real objects between heterogeneous machine architectures may not be exact. Real literals are written as a *mantissa* with an optional *exponent*. A mantissa is either a sequence of one or more decimal digits, or two sequences (one of which may be empty) joined by a period. The mantissa must contain at least one digit. An exponent is 'E' or 'e', optionally followed by '+' or '-', followed by one or more decimal digits. An exponent is required if the mantissa does not contain a period. As is usual, $mEx = m^2 10^x$. Examples of real literals are:

3.14 3.14E0 314e-2 .0314E+2 3. .14

As with integers, the operations add(+), sub(-), mul(*), div(l), mod(l), power(**), max, min, minus(-), abs, lt(<), le(<=), equal(=), ge(>=), and gt(>), are provided. It is important to note that there is no form of *implicit* conversion between types. The *l2r* operation converts an integer to a real, *r2i* rounds a real to an integer, and *trunc* truncates a real to an integer. See Section 11.5 for details.

6.2.5. Char

The type char provides the alphabet for text manipulation. Characters are immutable, atomic, transmissible, and form an ordered set. Every implementation must provide at least 128, but no more than 512, characters; the first 128 characters are the ASCII characters in their standard order.

Literals for the printing ASCII characters (octal 40 through octal 176), other than single quote (') or backslash (\), can be written as that character enclosed in single quotes. Any character can be written by enclosing one of the escape sequences listed in Table 6-1 in single quotes. The escape sequences may be written using upper case letters, but note that escape sequences of the form \&* are case sensitive. A table of literals is given at the end of Appendix I. Examples of character literals are:

\7' 'a' '™' \"' \B' \177'

There are two operations, *i2c* and *c2i*, for converting between integers and characters: the smallest character corresponds to zero, and the characters are numbered sequentially. Binary comparison operations exist for characters based on this numerical ordering: *It* (<), *Ie* (<=), *equal* (=), *ge* (>=), and *gt* (>). For details, see Section II.6.

((etause eigenie)	
Vn NL (Continue)	
V HT Guidenmint Aubt	
V CR	
n an	
Va*	

Table 6-1: Character Escape Surpose Forms

6.2.6. String

The type string is used for representing text. A string is an annualitie, stomic, and transmissible sequence of zero or store characters. Although an background define and background of the antiperiod string text or character antipe sequences (a string text), endowed in double groups (").

The characters of a string an induced sequentially starting from one. The lateb operation is used to obtain a character by index. The subar operation is send to character by index. The subar operation is send to character by index. The subar operation is send to character by index. The subar operation is send to character by index.

Two strings can be concatenated together with concer(i), and a single discussion can be opponded to the end of a string with append. One converts a descent to a single descent of the . The else of a string can be determined with size. Chara tensors ever the discussion of early from the flat to the test character. There are also the usual leatengraphic completions appendices: #(a), its (---), appen(-), ge (>--), and gt (>). For details, use Begins #.7:

6.2.7. Any

Objects of type any may contain objects of any type, and thus provide an eacape from comple-time type checking. Linike CLU, which tracks any differentiation all informations, and is a normal type in Argue. To this and there is an another areas ground to ground the line areas and is a normal type or operation generator of type any.

An object of type any can be thought of an containing an almost and in type. Since there are no operations provided by type and that change bits state, and alticular solutions and the term the homestable. However, the state of the contained anjug may change if that exploit terminants are trees this point of view, the mutability and atomicity of an any object depend on the mutability and atomicity of the contained object. Objects of type any are not transmissible.

The create operation is parameterized by a type; create takes a single argument of that type and returns an any object containing the argument. The force operation is also parameterized by a type; it takes an any and extracts an object of that type, signalling wrong_type if the contained object's type is not included in the parameter type. The *is_type* operation is parameterized by a type and checks whether its argument contains an object whose type is included in the parameter type. The detailed specification is found in Section II.19.

6.2.8. Sequence Types

Sequences are immutable and they are atomic or transmissible when instantiated with atomic or transmissible type parameters. Although an individual sequence can have any length, the length and members of a sequence are fixed when the sequence is created. The elements of a sequence are indexed sequentially, starting from one. A sequence type specification has the form:

sequence [type_actual]

where a type_actual is a type_spec, possibly augmented with operation bindings (see Section 12.6).

The new operation returns an empty sequence. A sequence constructor has the form:

```
type spec $ [ expression .... ] ]
```

and can be used to create a sequence with the given elements.

Although a sequence, once created, cannot be changed, new sequences can be constructed from existing ones by means of the addh, addl, remh, and reml operations. Other operations include fetch, replace, top, bottom, size, the elements and indexes iterators, and subseq. Invocations of the fetch operation can be written using a special form:

q[i] % fetch the element at index i of q .

Two sequences with equal elements are equal. The equal (=) operation tests if two sequences have equal elements, using the equal operation of the element type. Similar tests if two sequences have similar elements, using the similar operation of the element type.

All operations are indivisible except for *fill_copy*, *equal*, *similar*, *copy*, *encode*, and *decode*, which are divisible at calls to the operations of the type parameter.

For the detailed specification, see Section II.8.

6.2.9. Array Types

Arrays are one-dimensional, and mutable but not atomic. They are transmissible only if their type parameter is transmissible. The number of elements in an array can vary dynamically. There is no notion of an "uninitialized" element.

The state of an array achieves at an integer called the four bound, and a pergraphe of objects called the elements. The elements of an array are induced acquired by the tank the ten bound. All of the elements must be of the same type; this type is specified in the array type specification, which has the form:

arrey [type_actual]

There are a number of ways to create a new array, of which only two are mentioned here. The create operation takes an experiment epochylag the law bound, and another a law array with that law bound and no elements. Alternatify, an array constitution can be used to create an antipy offic an addition number of initial elements. For example,

array[int] \$ [5: 1, 2, 3, 4]

creates an integer army with low bound 5, and four elements, while

army formal \$ (map, total)

creates a boolean array with low bound 1 (the default), and ano elements.

An array type specification states stating about the travels of an array. This is because arrays can grow and shrink dynamically, using the solid, and, and, and and and are stated with and the state of the state o

all 1. So for the channel at index 1 of a all > 3. States 3 at index 1 of a for online atom.

Every newly created every ten an identity that is claimed than all allow annual: two arrays can have the same elements without being the same annual signal. The site that with the equal (--) operation. The site last spectrum tensor if two arrays tensor the same, using the equal (--) operation of the element type. Allowershape if two arrays include the state to generate the tensor interview tensor interview and the element type.

All operations are indivisible, encept ill_copy, eithlier, similari, expy, encepte, and decode, which are divisible at calls to operations of the type parameter.

For the detailed specification, see Bection 11.9.

8.2.10. Structure Types

A structure is an instructable collection of one or more sufficients. An instruction is alomic or transmissible only if the type generations are all depine at all the structure. An instruction of sufficient and the structure of the structure of

struct [Reid apac. ...]

where

field_spec IIII name , ... : type_actual Selectors must be unknow within a specification, but the ordering and grouping of selectors is unimportant.

26
6.2.10 Structure Types

A structure is created using a structure constructor. For example, assuming that "info" has been equated to a structure type:

info = struct[last, first, middle: string, age: int] the following is a legal structure constructor:

info \$ {last: "Scheifler", first: "Robert", age: 32, middle: "W."}

An expression must be given for each selector, but the order and grouping of selectors need not resemble the corresponding type specification.

For each selector "sel", there is an operation <u>get_sel</u> to extract the named component, and an operation <u>replace_sel</u> to create a new structure with the named component replaced with some other object. Invocations of the <u>get</u> operations can be written using a special form:

st.age % get the 'age' component of st

As with sequences, two structures with equal components are in fact the same object. The equal (=) operation tests if two structures have equal components, using the equal operations of the component types. Similar tests if two structures have similar components, using the similar operations of the component types.

All operations are indivisible except for equal, similar, copy, encode, and decode, which are divisible at calls to the operations of the type parameter.

For the detailed specification, see Section II.11.

6.2.11. Record Types

A record is a mutable collection of one or more named objects. Records are never atomic, and are transmissible only if the parameter types are all transmissible. A record type specification has the form:

```
record [field_spec, ....]
```

where (as for structures)

field_spec ::= name , ... : type_actual

Selectors must be unique within a specification, but the ordering and grouping of selectors is unimportant.

A record is created using a record constructor. For example: professor \$ {last: "Herlihy", first: "Maurice", age: 32, middle: "P."}

For each selector "sel", there is an operation *get_sel* to extract the named component, and an operation *set_sel* to replace the named component with some other object. Invocations of these operations can be written using a special form:

r.middle % get the 'middle' component of r r.age := 33 % set the 'age' component of r to 33 (by calling set_age)

As with arrays, every newly created record has an identity that is distinct from all other records; two records can have the same components without being the same record object. The identity of records

can be distinguished with the equal (=) operation. The similar1 operation tests if two records have equal components, using the equal operations of the component types. Similar tests if two records have similar components, using the similar operations of the component types.

All operations are indivisible, except *similar*, *similar*, *copy*, *encode*, and *decode*, which are divisible at calls to operations of the type parameters.

For the detailed specification, see Section II.12.

6.2.12. Oneof Types

A oneof type is a *tagged*, *discriminated union*. A oneof is an immutable labeled object, to be thought of as "one of" a set of alternatives. The label is called the *tag*, and the object is called the *value*. A oneof type specification has the form:

oneof [field_spec , ...]
where (as for structures)

field spec ::= name, ... : type actual

Tags must be unique within a specification, but the ordering and grouping of tags is unimportant. An instantiation is atomic or transmissible if and only if all the type parameters are atomic or transmissible.

For each tag "t" of a oneof type, there is a make_t operation which takes an object of the type associated with the tag, and returns the object (as a oneof) labeled with tag "t".

To determine the tag and value of a oneof object, one normally uses the tagcase statement (see Section 10.14).

The equal (=) operation tests if two oneofs have the same tag, and if so, tests if the two value components are equal, using the equal operation of the value type. Similar tests if two oneofs have the same tag, and if so, tests if the two value components are similar, using the similar operation of the value type.

All operations are indivisible, except equal, similar, similar1, copy, encode, and decode, which are divisible at calls to operations of the type parameters.

For the detailed specification, see Section II.14.

6.2.13. Variant Types

A variant is a mutable oneof. Variants are never atomic and are transmissible if and only if their type parameters are all transmissible. A variant type specification has the form:

```
variant [ field_spec , ... ]
```

where (as for oneofs)

field_spec ::= name , ... : type_actual

6.2.13 Variant Types

The state of a variant is a pair consisting of a label called the *tag* and an object called the *value*. For each tag "t" of a variant type, there is a *make_t* operation which takes an object of the type associated with the tag, and returns the object (as a variant) labeled with tag "t". In addition, there is a *change_t* operation, which takes an existing variant and an object of the type associated with "t", and changes the state of the variant to be the pair consisting of the tag "t" and the given object. To determine the tag and value of a variant object, one normally uses the tagcase statement (see Section 10.14).

Every newly created variant has an identity that is distinct from all other variants; two variants can have the same state without being the same variant object. The identity of variants can be distinguished using the equal (=) operation. The similar1 operation tests if two variants have the same tag, and if so, tests if the two value components are equal, using the equal operation of the value type. Similar tests if two variants have the same tag, and if so, tests if the two value components are similar, using the similar operation of the value type.

All operations are indivisible, except *similar*, *similar*, *copy*, *encode*, and *decode*, which are divisible at calls to operations of the type parameters.

For the detailed specification, see Section II.15.

6.2.14. Procedure and Iterator Types

Procedures and iterators are created by the Argus system or by the bind expression (see Section 9.8). They are not transmissible. As the identity of a procedure or iterator is immutable, they can be considered to be atomic. However, their atomicity can be violated if a procedure or iterator has own data and thus a mutable state. The immutability and atomicity of a procedure or iterator with own data depends on that operation's specified semantics.

The type specification for a procedure or iterator contains most of the information stated in a procedure or iterator heading; a procedure type specification has the form:

proctype ([type_spec, ...]) [returns][signals] and an iterator type specification has the form:

```
itertype ( [ type_spec , ... ] ) [ yields ] [ signals ]
```

where

```
returns::= returns ( type_spec , ... )yields::= yields ( type_spec , ... )signals::= signals ( exception , ... )exception::= name [ ( type_spec , ... ) ]
```

The first list of type specifications describes the number, types, and order of arguments. The returns or yields clause gives the number, types, and order of the objects to be returned or yielded. The signals clause lists the exceptions raised by the procedure or iterator; for each exception name, the number, types, and order of the objects to be returned is also given. All names used in a signals clause must be unique. The ordering of exceptions is not important.

Procedure and iterator types have an equal (=) operation. Invocation is not an operation, but a primitive in Argus. For the detailed specification of proctype and itertype, see Section II.17.

医心脏 计编辑系统 化合理学 经济利益分析

6.3. Atomic_Array, Atomic_Record, and Atomic_Variant

Having described the types that Argus inherited from CLU, we now describe the new types in Argus. The mutable atomic type generators of Argus are **atomic_array**, **atomic_record**, and **atomic_variant**. Types obtained from these generators provide the same operations as the analogous types obtained from array, record, and variant, but they differ in their synchronization and recovery properties. Conversion operations are provided between each atomic type generator and its non-atomic partner (for example, atomic_array[t]\$aa2a converts from an atomic array to a (non-atomic) array).

An operation of an atomic type generator can be classified as a *reader* or *writer* depending on whether it examines or modifies its *principal* argument, that is, the argument or result object of the operation's type. (For binary operations, such as ar__gets__ar, the operation is classified with respect to each argument.) Intuitively, a *reader* only examines (reads) the state of its principal argument, while a *writer* modifies (writes) its principal argument. Operations that create objects of an atomic type are classified as readers. Reader/writer exclusion is achieved by locking: readers acquire a read lock while writers acquire a write lock. The locking rules are discussed in Section 2.2.2.

If one or more of the type parameters is non-atomic, then the resulting type is not atomic because modifications to component objects are not controlled. However, read/write locking still occurs, as described above. Thus, an atomic type generator instantiated with a non-atomic parameter incurs the expense of atomic types without gaining any benefit; such an instantiation is unlikely to be a correct solution to a problem. Atomic type generators yield transmissible types only if the type parameters are all transmissible.

Special operations are provided for each atomic type generator to test and manipulate the locks associated with reader/writer exclusion. These operations are useful for implementing user-defined atomic types (see Section 15). The tagtest and tagwalt statements (see Section 10.15) provide additional structured support for atomic_variants. The operations can_read, can_write, Test_and_read, and test_and_write provide relatively unstructured access to lock information. For complete definitions of these operations, see Sections II.10, II.13, and II.16.

Assuming normal termination, the following operations acquire read locks on their principal arguments or the objects that they create.

atomic_array :	create, new, predict, fill, fill_copy, size, low, high, empty, top, bottom, fetch, similar, similar1, copy, copy1, elements, indexes, test_and_read, a2aa, aa2a, encode, decode
atomic_record:	create, get_, similar, similar1, copy, copy1, test_and_read, ar_gets_ar (second argument), r2ar, ar2r, encode, decode
atomic_variant:	make_, is_, value_, av_gets_av (second argument), similar, similar1, copy, copy1, test_and_read, v2av, av2v, encode, decode

6.3 Atomic_Array, Atomic_Record, and Atomic_Variant

The operations *similar* and *similar1* acquire read locks on both arguments. The operations *copy* and *copy1* acquire a read lock on the value returned as well as their principal argument. *Test_and_read* is a reader only if it returns true; otherwise it is neither a reader nor a writer.

Assuming normal termination, the following operations acquire write locks on their principal arguments.

 atomic_array:
 set_low, trim, store, addh, addl, remh, reml, test_and_write

 atomic_record:
 set_, ar_gets_ar (first argument), test_and_write

 atomic_variant:
 change_, av_gets_av (first argument), test_and_write

 Test_and_write is a writer only if it returns true; otherwise it is neither a reader nor a writer.

The equal, can_read, and can_write operations are neither readers nor writers.

When an operation of **atomic_array** terminates with an exception, its principal argument is never modified; however, the **atomic_array** operations listed above as writers always obtain a write lock before the principal argument is examined, hence there are cases in which they will obtain a write lock and only read, but not modify their principal argument. For example, **atomic_array[t]\$***trim* is a writer when it signals *bounds*. On the other hand, when an **atomic_array** operation raises a signal because of an invalid argument, no locks are obtained. For example, when **atomic_array[t]\$***trim* signals *negative_size*, it is neither a reader nor a writer since the array's state is neither examined nor modified (only the integer argument is examined).

For the detailed specification of atomic arrays, see Section II.10; for atomic records, see Section II.13; and for atomic variants, see Section II.16.

6.4. Guardian Types

Guardian types are user-defined types that are implemented by guardian definitions (see Section 13). A guardian definition has a header of the form:

idn = guardian [parms] is idn , ... [handles idn , ...] [where]

The creators are the operations named in the identifier list following is; a creator is a special kind of operation that can be called to create new guardians that behave in accordance with the guardian definition. Each guardian optionally provides *handlers* that can be called to interact with it; the names of these handlers are listed in the identifier list following **handles**. (See Section 13 for more details.)

A guardian definition named g defines a guardian interface type g. An object of the guardian interface type provides an interface to a guardian that behaves in accordance with the guardian definition. An interface object is created whenever a new guardian is created, and then the interface object can be used to access the guardian's handlers. Interface objects are transmissible, and after transmission they still give access to the same guardian. In this manual a "guardian interface object" is often called simply a "guardian object".

The guardian type g for the guardian definition named g has the following operations.

- 1. The creators listed in the is list of the guardian definition.
- 2. For each handler name h listed in the handles list, an operation get_h with type: proctype (g) returns (ht), where ht is the type of h.
- 3. Equal and similar, both of type: proctype (g, g) returns (bool), which return true only if both arguments are the same guardian object.
- 4. Copy, of type: proctype (g) returns (g), which simply returns its argument.
- 5. transmit.

A creator may not be named equal, similar, copy, print, or get h where h is the name of a handler.

Thus if x is a variable denoting a guardian interface object of type g, and h is a handler of g, then g\$get_h(x) will return this handler. As usual with get_ operations, this call can be abbreviated to x.h. Note that the handlers themselves are not operations of the guardian interface type; thus g\$h would be illegal.

A guardian interface type is somewhat like a structure type. Its objects are constructed by the creators, and decomposed by the *get_* operations. Guardian interface objects are immutable and atomic.

6.5. Handler and Creator Types

Creators are operations of guardian types. Handler objects are created as a side-effect of guardian creation. Unlike procedures and iterators, handlers and creators are transmissible.

The types of handlers and creators resemble the types of procedures:

handlertype ([type_spec,]		[returns]		signals	
creatortype (type_spec,])[returns]	ĺ	signals]	

The argument, normal result, and exception result types must all be transmissible. The signals list for a handlertype or creatortype cannot include either failure or unavailable, as these signals are implicit in the interface of all creators and handlers.

Handler and creator types provide equal and similar operations which return true if and only if both arguments are the same object, and copy operations which simply return their argument. For the detailed specification of handlertype and creatortype, see Section 11.18.

6.6. Image

The **Image** type provides an escape from compile-time type checking. The main difference between **Image** and **any** is that **Image** objects are transmissible. An **image** object can be thought of as a portion of an undecoded message or as the information needed to recreate an object of some type. **Image** objects are immutable and atomic.

The create operation is parameterized by a transmissible type; it takes a single argument of that type and encodes it (using the encode operation of that type) into an image object. The force operation is also

6.6 Image

parameterized by a transmissible type; it takes an **image** object and decodes it (using the *decode* operation of that type) to an object of that type, signalling *wrong_type* if the encoded object's type is not included in the parameter type. The *is_type* operation is parameterized by a type and checks whether its argument is an encoded object of a type included in the parameter type. See Section II.20 for the detailed specification.

6.7. Mutex

Mutex objects are mutable containers for information. They are not atomic, but they provide synchronization and control of writing to stable storage for their contained object. Mutex itself does not provide operations for synchronizing the use of mutex objects. Instead, mutual exclusion is achieved using the selze statement (see Section 10.16), which allows a sequence of statements to be executed while a process is in exclusive possession of the mutex object. Mutex objects are transmissible if the contained object is transmissible.

The type generator mutex has a single parameter that is the type of the contained object. A mutex type specification has the form:

mutex [type_actual]

Mutex types provide operations to create and decompose mutex objects, and to notify the system of modifications to the mutex object or its contained object.

The create operation takes a single argument of the parameter type and creates a new mutex object containing the argument object. The *get_value* operation obtains the contained object from its mutex argument, while *set_value* modifies a mutex object by replacing its contained object. As with records, these operations can be called using special forms, for example:

m: mutex[int] := mutex[int]\$create (0)x: int := m.value% extract the contained objectm.value := 33% change the contained object

Set_value and get_value are indivisible.

Mutexes can be distinguished with the equal (=) operation. There are no operations that could cause or detect sharing of the contained object by two mutexes. Such sharing is dangerous, since two processes would not be synchronized with each other in their use of the contained object if each possessed a different mutex. In general, if an object is contained in a mutex object, it should not be contained in any other object, nor should it be referred to by a variable except when in a seize statement that has possession of the containing mutex.

There are some mutex operations that seize the mutex object automatically. Copy seizes its single argument object. Similar seizes its two argument objects; the first argument object is seized first and then the second. In both cases possession is retained until the operations return. Also, when a mutex object is encoded (for a message or when making an image), the object is seized automatically. See Section II.21 for the detailed specification of mutex.

Mutexes are used primarily to provide process synchronization and mutual exclusion on shared data, especially to implement user-defined atomic types. In such implementations, it is important to control writing to stable storage. The mutex operation *changed* provides the necessary control. *Changed* informs the system that the calling action requires that the argument object be copied to stable storage before the commit of the action's top-level parent (topaction). Any mutex is asynchronous: its contained object is written to stable storage independently of objects that contain that mutex. See Section 15 for further discussion of user-defined atomic objects.

6.8. Node

Objects of type node stand for physical nodes. The operation here takes no arguments and returns the node object that denotes its caller's node. Equal, similar, and copy operations are also provided.

The main use of **node** objects is in guardian creation (see Section 13), where they are used to cause a newly created guardian to reside at a particular node. Objects of type **node** are immutable, atomic, and transmissible. For the detailed specification, see Section 11.2.

6.9. Other Type Specifications

A type specification for a user-defined type has the form of a reference:

reference ::= idn | idn [actual_parm , ...] | reference \$ name

where each actual_parm must be a compile-time computable constant (see Section 7.2) or a type_actual (see Section 12.6). A reference must denote a data abstraction to be used as a type specification; this syntax is provided for referring to a data abstraction that is named in an equate module (see Section 12.4). For type generators, actual parameters of the appropriate types and number must be supplied. The order of parameters is always significant for user-defined types (see Section 12.5).

There are two special type specifications that are used when implementing new abstractions: rep, and cvt. These forms may only be used within a cluster; they are discussed further in Section 12.3.

Within an implementation of an abstraction, formal parameters declared with type can be used as type specifications. Finally, identifiers that have been equated to type specifications can also be used as type specifications.

7. Scopes, Declarations, and Equates

This section describes how to introduce and use constants and variables, and the scope of constant and variable names. Scoping units are described first, followed by a discussion of variables, and finally constants.

7.1. Scoping Units

Scoping units follow the nesting structure of statements. Generally, a scoping unit is a body and an associated "heading". The scoping units are as follows (see Appendix I for details of the syntax).

- 1. From the start of a module to its end.
- From a cluster, proc, iter, equates, guardian, handler, or creator to the matching end.
- 3. From a for, do, begin, background, recover, enter, coenter, or seize to the matching end.
- 4. From a then or else in an if statement to the end of the corresponding body.
- 5. From a tag, wtag, or others in a tagcase, tagwait, or tagtest statement to the end of the corresponding body.
- 6. From a when or others in an except statement to the end of the corresponding body.
- 7. From the start of a type_set to its end.
- 8. From an action or topaction to the end of the corresponding body.

The structure of scoping units is such that if one scoping unit overlaps another scoping unit (textually), then one is fully contained in the other. The contained scope is called a *nested* scope, and the containing scope is called a *surrounding* scope.

New constant and variable names may be introduced in a scoping unit. Names for constants are introduced by equates, which are syntactically restricted to appear grouped together at or near the beginning of scoping units (except in type sets). For example, equates may appear at the beginning of a body, but not after any statements in the body.

In contrast, declarations, which introduce new variables, are allowed wherever statements are allowed, and hence may appear throughout a scoping unit. Equates and declarations are discussed in more detail in the following two sections.

In the syntax there are two distinct nonterminals for identifiers: *idn* and *name*. Any identifier introduced by an equate or declaration is an *idn*, as is the name of the module being defined, and any operations it has. An *idn* names a specific type or object. The other kind of identifier is a *name*. A *name* is generally used to refer to a piece of something, and is always used in context; for example, *names* are used as record selectors. The scope rules apply only to *idns*.

The scope rules are simple:

- 1. An *idn* may not be redefined in its scope.
- 2. Any *idn* that is used as an external reference in a module may not be used for any other purpose in that module.

Unlike other "block-structured" languages, Argus prohibits the redefinition of an identifier in a nested scope. An identifier used as an external reference names a module or constant; the reference is resolved using the compilation environment.

7.1.1. Variables

Objects are the fundamental "things" in the Argus universe; variables are a mechanism for denoting (i.e., naming) objects. A variable has three properties: its type, whether it is stable or not, and the object that it currently denotes (if any). A variable is said to be *uninitialized* if it does not denote any object. Attempts to use uninitialized variables are programming errors and (if not detected at compile-time) cause the guardian to crash.

There are only three things that can be done with variables:

- 1. New variables can be introduced. Declarations perform this function, and are described below.
- 2. An object may be assigned to a variable. After an assignment the variable denotes the object assigned.
- 3. A variable may be used as an expression. The value of a variable is the object that the variable denotes at the time the expression is evaluated.

7.1.2. Declarations

Declarations introduce new variables. The scope of a variable is from its declaration to the end of the smallest scoping unit containing its declaration; hence, variables must be declared before they are used.

There are two sorts of declarations: those with initialization, and those without. Simple declarations (those without initialization) take the form

decl ::= idn , ... : type_spec

A simple declaration introduces a list of variables, all having the type given by the type_spec. This type determines the types of objects that can be assigned to the variable. The variables introduced in a simple declaration initially denote no objects, i.e., they are uninitialized.

A declaration with initialization combines declarations and assignments into a single statement. A declaration with initialization is entirely equivalent to one or more simple declarations followed by an assignment statement. The two forms of declaration with initialization are:

idn : type_spec := expression

and

```
decl<sub>1</sub>, ..., decl<sub>n</sub> := call [ @ primary ]
These are equivalent to (respectively):
```

```
idn : type_spec
idn := expression
```

and

decl₁ ... decl_n % declaring idn₁ ... idn_m

idn₁, ..., idn_m := call [@ primary]

In the second form, the order of the idns in the assignment statement is the same as in the original declaration with initialization. (The call must return *m* objects.)

7.2. Equates and Constants

An equate allows an identifier to be used as an abbreviation for a constant, type set, or equate module name that may have a lengthy textual representation. An equate also permits a mnemonic identifier to be used in place of a frequently used constant, such as a numerical value. We use the term constant in a very narrow sense here: constants, in addition to being immutable, must be computable at compile-time. Constants are either types (built-in or user-defined), or objects that are the results of evaluating constant expressions. (Constant expressions are defined below.)

The syntax of equates is:

```
equate ::= idn = constant
| idn = type_set
| idn = reference
```

```
constant ::= type_spec
expression
```

type_set ::= { idn | idn has oper_decl , ... { equate } }

```
reference ::= idn
idn [ actual_parm , ... ]
reference $ name
```

References can be used to name equate modules.

An equated identifier may not be used on the left-hand side of an assignment statement.

The scope of an equated identifier is the smallest scoping unit surrounding the equate defining it; here we mean the entire scoping unit, not just the portion after the equate. All the equates in a scoping unit must appear grouped near the beginning of the scoping unit. The exact placement of equates depends on the containing syntactic construct; usually equates appear at the beginnings of bodies.

Equates may be in any order within the a scoping unit. Forward references among equates in the same scoping unit are allowed, but cyclic dependencies are illegal. For example,

x = y y = z z = 3

is a legal sequence of equates, but

X = Y Y = Z Z = X

is not. Since equates introduce idns, the scoping restrictions on idns apply (i.e., the idns may not be defined more than once).

7.2.1. Abbreviations for Types

Identifiers may be equated to type epoclications, giving ablanulations for type names.

7.2.2. Constant Expressions

We define the subset of objects that equated identifiers may decide by stating which expressions are constant expressions. (Expressions are decused in detail in Section 6.) A constant expression is an expression that can be evaluated at compile-time to produce an immutable adjust of a built-in type. This includes:

1. Literais.

- 2. Identifiers equated to constants.
- 3. Formal parameters.
- 4. Procedure, iterator, and creator names.
- 5. Bind expressions (see Section 9.8), where the routine bound and the explicit arguments are all constants.
- 6. Invocations of procedure operations of the built-is immutable types, provided that all operands are constant expressions that are not formed presidents.

The built-in immutable types are: and, int, weat, band, allow, allow, express types, oneof types, structure types, procedure types, fierator types, and creative types.

We explicitly forbid the use of formal parameters as operands to calls in constant expressions, since the values of formal parameters are not income at compile thes. If the evaluation of a constant expression would signal an exception, the constant defined by that expression is fligged.

38

8. Assignment and Calls

The two fundamental activities of Argus programs are calls and assignment of computed objects to variables.

Argus programs should use mutual exclusion or atomic data to synchronize access to all shared variables, because Argus supports concurrency and thus processes can interfere with each other during assignments. For example,

i := 1 j := 2

is not equivalent to

i, j := 1, 2

in the presence of concurrent assignments to the same variables, because any interleaving of indivisible events is possible in the presence of concurrency.

Argus is designed to allow complete compile-time type-checking. The type of each variable is known by the compiler. Furthermore, the type of objects that could result from the evaluation of any expression is known at compile time. Hence, every assignment can be checked at compile time to ensure that the variable is only assigned objects of its declared type. An assignment v := E is legal only if the type of E is included the type of v. The definition of type inclusion is given in Section 6.1.

8.1. Assignment

Assignment causes a variable to denote an object. Some assignments are implicitly performed as part of the execution of various mechanisms of the language (in exception handling, and the tagcase, tagtest, and tagwalt statements). All assignments, whether implicit or explicit, are subject to the type inclusion rule.

8.1.1. Simple Assignment

The simplest form of assignment statement is:

idn := expression

In this case the *expression* is evaluated, and then the resulting object is assigned to the variable named by the *idn* in an indivisible event. Thus no other process may observe a "half-assigned" state of the variable, but another process may observe various states during the expression evaluation and between the evaluation of the expression and the assignment. The expression must return a single object (whose type must be included in that of the variable).

8.1.2. Multiple Assignment

There are two forms of assignment statement that assign to more than one variable at once:

```
idn , .... := expression , ....
```

and

idn , := call [@ primary]

The first form of multiple assignment is a generalization of simple assignment. The first variable is assigned the first expression, the second variable the second expression, and so on. The expressions are all evaluated (from left to right) before any assignments are performed. The assignment of multiple objects to multiple variables is an indivisible event, but evaluation of the expressions is divisible from the actual assignment. The number of variables in the list must equal the number of expressions, no variable may occur more than once, and the type of each variable must include the type of the corresponding expression.

The second form of multiple assignment allows one to retain the objects resulting from a call returning two or more objects. The first variable is assigned the first object, the second variable the second object, and so on, but all the assignments are carried out indivisibly. The order of the objects is the same as in the return statement executed in the called routine. The number of variables must equal the number of objects returned, no variable may occur more than once, and the type of each variable must include the corresponding return type of the called procedure.

8.2. Local Calls

In this section we discuss procedure calls; iterator calls are discussed in Section 10.12. However, argument passing is the same for both procedures and iterators.

Local calls take the form:

```
primary ([expression, ...])
```

The sequence of activities in performing a local call are as follows:

- 1. The primary is evaluated.
- 2. The expressions are evaluated, from left to right.
- New variables are introduced corresponding to the formal arguments of the routine being called (i.e., a new environment is created for the called routine to execute in).
- 4. The objects resulting from evaluating the expressions (the actual arguments) are assigned to the corresponding new variables (the formal arguments). The first formal is assigned the first actual, the second formal the second actual, and so on. The type of each expression must be included in the type of the corresponding formal argument.
- 5. Control is transferred to the routine at the start of its body.

A call is considered legal in exactly those situations where all the (implicit) assignments are legal.

A routine may assign an object to a formal argument variable; the effect is just as if that object were assigned to any other variable. From the point of view of the called routine, the only difference between its formal argument variables and its other local variables is that the formals are initialized by its caller.

Procedures can terminate in two ways: they can terminate *normally*, returning zero or more objects, or they can terminate *exceptionally*, signalling an exceptional condition. When a procedure terminates

8.2 Local Calls

normally, any result objects become available to the caller, and can be assigned to variables or passed as arguments to other routines. When a procedure terminates exceptionally, the flow of control will not go to the point of return of the call, but rather will go to an exception handler (see Section 11).

8.3. Handler Calls

As explained in Section 2 and in Section 13, a handler is an operation that belongs to some guardian. A handler call causes an activation of the called handler to run at the handler's guardian; the activation is performed at the called handler's guardian by a new subaction created solely for this purpose. Usually the handler's guardian is not the same as the one in which the call occurs, and the called handler's guardian is likely to reside at a different node in the network than the calling guardian. However, it is legal to call a handler that belongs to a guardian residing at the caller's node, or even to call a handler belonging to the caller's guardian.

Although the form of a handler call looks like a procedure call:

primary ([expression, ...])

its meaning is very different. Among other things, a handler is called remotely, with the arguments and results being transmitted by value in messages, and the call is run as a subaction of its calling action. Below we present an overview of what happens when executing a handler call and then a detailed description.

A handler call runs as a subaction of the calling action. We will refer to this subaction as the *call action*. The first thing done by the call action is the transmission of the arguments of the call. Transmission is accomplished by encoding each argument object, using the *encode* operation of its type. The arguments are decoded at the called guardian by a subaction of the call action called the *activation action*. Each argument is decoded by using the *decode* operation of its type. The effect of transmission is that the arguments are passed by value from the caller to the handler activation: new objects come into existence at the handler's guardian that are copies of the argument objects. Object values are transmitted in such a way as to preserve the internal sharing structure of each argument object is preserved⁶, as well as any sharing structure between the argument objects in a single call. See Section 14 for further discussion of transmission.

After the arguments have been transmitted, the activation action performs the handler body. When the handler body terminates, by executing a return, abort return, signal, or abort signal statement, the result objects are transmitted to the caller by encoding them at the handler's guardian, and committing or aborting the activation action (as it specified). The call action then decodes the results at the caller's guardian. Once the results have been transmitted to the caller's code. (Note that the call action will commit even if the activation action aborts.)

⁸This is only strictly true for the built-in types. A user-defined type might not preserve internal sharing structure.

The above discussion has ignored the possibility of several problems that may arise in executing a handler call. These problems either cause the call added of the authorities a separation to short or result in the orash of the calling guardian. A handler call attempted hass solution a separation or subsection is a programming error, and so if the baggeous the calling guardian is because the call action or the activation action to be aboved, and the baggeous the calling guardian is a called a separation or an an analytic several data is an attempted back to the call action or the activation action to be aboved, and the baggeous the calling guardian because the call action or the activation action to be aboved, and the baggions are appeared by the Argue system. Two such executions are the problem that two extended attempts.

The meaning of a failure exception generated by the Argue system is that this particular call did not succeed, and turbannore it is unlikely to succeed it repeated. These are two success why failure is raised: an error occurred in transmitting an argument or taxill, anthe familier's gravitien no longer ories.

The Argue system raises the unexaliable enception when it is unable to approximate with the bandler's in and a second of the collect guardian. Reasons why communication may left instants addresses guardian or its node. The Argus system relate the units impossible at that time; it may by many finant in call nn, when a coll terminates with the unavailable acception, there is this pain in dialaty. Hannard, unite a cell terminated by the failure comption, a cell tail complete successfully if retried later. Note that the argument 1 as the system tries to astabilish communication.

For example, suppose we have a handler call:

meand mail(user, my meanage)

where m is a malter guardian, and the source multihandler has the baselor

and_mail - handler be car it may be made all the second

Then your and my_minings and exactled using the analytic gravities of types your is and meneage, respectively, and the special values as demonstrative addition of the second second specializes of these types. If user is actually registered to reacte mail. The second second second remaily; otherwise it signals no_such_user. In other case no encoding or describing of the range is meeted along these there is no result.

Possible exceptions from this call are no_euch_user, failure, and unevaliable. So the call might be performed in an except statement:

m.cont Malifusor, my macaaco) exidept vitadi ne (anth antr: ... vitadi amandalite (a: daing); ... vitadi takev (a: daing); ...

8.3.1. Semantics of Handler Calls

In this section we describe the semantics of a handler call in detail. A handler call causes activity at both the calling guardian and at the called guardian. At the calling guardian, the sequence of activities in performing a handler call is as follows:

www.cest

ことにより、このものないという「こうないない」というとうない。 あままで 教育学校 教育のないから かくうう

- 1. The primary is evaluated.
- 2. The argument expressions are evaluated from left to right.
- 3. A subaction, which we will refer to as the call action, is created for the remote call. All subsequent activity on behalf of the call will be performed by the call action or one of its descendants. For it to be possible to create the call action, the caller must already be running as an action. Remote calls by non-actions are programming errors and cause the calling guardian to crash.
- 4. A call message is constructed. As part of constructing this message, encode operations are performed on the argument objects. If any of the encode operations terminates with a failure exception, then the remote call will terminate with the same exception, and the call action will be aborted.
- The call message is sent to the guardian of the called handler, and the call action waits for the completion of the call.
- 6. If the call message arrives at the node of the target guardian, and the target guardian does not exist, then the call action is aborted with the *failure* exception having the string "guardian does not exist" as its exception result.
- 7. If the system determines that it cannot communicate with the called guardian, it aborts the call action. The call action may be retried several times (beginning at step 3) in attempts to communicate. If repeated communication failures are ensauntered, the system aborts the call action and causes the call to terminate with the unavailable exception. The system will cause this kind of termination only when it is extremely unlikely that retrying the call immediately will succeed.
- 8. Ordinarily, a call completes when a reply message containing the results is received. When the reply message arrives at the caller, it is decoded using the decode operation for each result object. If any decode terminates with a *failure* exception, the call action is aborted, and the call terminates with the same exception. Otherwise, the call action commits.
- 9. The call will terminate normally if the result message indicates that the handler activation returned (instead of signalled); otherwise it terminates with whatever exception was signalled.

At the called guardian, the following activities take place.

- A subaction of the call action is created at the target guardian to run the call. We will refer to this subaction as the activation action. All activity at the target guardian occurs on behalf of the activation action or one of its descendants.
- 2. The call message is decomposed into its constituent objects. As part of this process decode operations are performed on each argument. If any decode terminates with a failure exception, then the activation action is aborted, and the call terminates with the same exception.
- The called handler is called within the activation action. This call is like a regular procedure call. The objects obtained from decoding the message are the actual arguments, and they are bound to the formals via implicit assignments.
- 4. If the handler terminates by executing an abort return or an abort signal statement (see Section 11.1), then all committed descendents of the activation action are aborted. Then the reply message is constructed by encoding the result objects, the activation action is

aborted, and the reply message is sent to the caller. Otherwise, when the handler terminates, the reply message is constructed by encoding the result objects, the activation action commits, and the reply message is sent to the caller. If one of the calls of encode terminates with a *failure* exception, then the activation action is aborted, and the call terminates with the same exception.

ميندين أوان

When the Argus system terminates a call with the *unavailable* exception, it is possible that the activation action and/or some of its descendants are actually running. This could happen, for example, if the network partitions. These running processes are called "orphans". The Argus system makes sure that orphans will be aborted before they can view inconsistent data (see Section 2.5).

8.4. Creator Calls

Creators are called to cause new guardians to come into existence. As part of the call, the node at which the newly created guardian will be located may be specified. If the node is not specified, then the new guardian is created at the same node as the caller of the creator. The form of a creator call is:

primary ([expression, ...]) [@ primary]

The primary following the at-sign (@) must be of type node.

A creator call causes two activities to take place. First, a new guardian is created at the indicated node. Second, the creator is called as a handler at the newly created guardian. This handler call has basically the same semantics as the regular handler call described above.

The Argus system may also cause a creator call to abort with the *failure* or *unavailable* exceptions. The reasons for such terminations are the same as those for handler calls, and the meanings are the same: the *failure* exception means that the call should not be retried, while the unavailable exception means that the call should not be retried, while the unavailable exception means that the call should not be retried.

8.4.1. Semantics of Creator Calls

The activities carried out in executing a creator call are as follows.

- 1. The (first) primary is evaluated.
- 2. The argument expressions are evaluated from left to right.
- 3. The optional *primary* following the at-sign is evaluated to obtain a node object. If this *primary* is missing, the node at which the call is taking place is used.
- 4. A subaction, which we will refer to as the *call action*, is created. All subsequent activity takes place within this subaction. As was the case for handler calls, creators can be called only from within actions. A creator call by a non-action is a programming error and causes the calling guardian to crash.
- A new guardian is created at the indicated node. The creator obtained in step 1 will indicate the type of this guardian. The selection of a particular load image for this type will occur as discussed in Section 3.3.
- 6. As was the case for handler calls, if the system cannot communicate with the indicated node, the creator call will terminate with the unavailable exception. If the system is unable

8.4.1 Semantics of Creator Calls

to determine what implementation to load, or if there is no implementation of the type that can run on the indicated node, or if the manager of the node refuses to allow the new guardian to be created, the creator call will terminate with the failure exception. In either case the call action will be aborted.

这次的现在分词 化乙酰氨酸乙酰氨酸乙酯 化分子子 化丁丁

7. A remote call is now performed to the creator. This call has the same semantics as described for handler calls above in steps 4 through 9 of the activities at the calling node and also steps 1 through 4 of activities at the called node. However, if either the call action or the activation action aborts, the newly created guardian will be destroyed.

For example, suppose we execute the creator call

x: G := G\$create(3) @ n

where G is a guardian type, n denotes an object of type node, and create has header

create = creator (n: Int) returns (G) signals (not_possible(string))

The system will select an implementation of G that is suitable for use at node n, and will then create a guardian at node n running that implementation. Next create (3) is performed as a handler call at that new guardian. If create returns, then the assignment to x will occur, causing x to refer to the new guardian that create returned; now we can call the handlers provided by G. The exceptions that can be signalled by this call are not_possible, failure, and unavailable. An example of a call that handles all these exceptions is:

x: G := G\$create (3) @ n except when not_possible (s: string): ... when failure (s: string): ... when unavailable (s: string): ... end

Creators are described in more detail in Section 13.

9 Expressions

9. Expressions

An expression evaluates to an object in the Argus universe. This object is said to be the *result* or *value* of the expression. Expressions are used to name the object to which they evaluate. The simplest forms of expressions are literals, variables, parameters, equated identifiers, equate module references, procedure, iterator, and creator names, and self. These forms directly name their result object. More complex expressions are built up out of nested procedure calls. The result of such an expression is the value returned by the outermost call.

9.1. Literals

Integer, real, character, string, boolean and null literals are expressions. The type of a literal expression is the type of the object named by the literal. For example, true is of type bool, "abc" is of type string, etc. (see the end of Appendix I for details).

9.2. Variables

Variables are identifiers that denote objects of a given type. The type of a variable is the type given in the declaration of that variable. An attempt to use an uninitialized variable as an expression is a programming error and causes the guardian to crash.

9.3. Parameters

Parameters are identifiers that denote constants supplied when a parameterized module is instantiated (see Section 12.5). The type of a parameter is the type given in the declaration of that parameter. Type parameters cannot be used as expressions.

9.4. Equated Identifiers

Equated identifiers denote constants. The type of an equated identifier is the type of the constant which it denotes. Identifiers equated to types, type__sets, and equate modules cannot be used as expressions.

9.5. Equate Module References

Equate modules provide a named set of equates (see Section 12.4). To use a name defined in an equate module as an expression, one writes:

reference \$ name

where

```
reference ::= idn
| idn [ actual_parm , ... ]
| reference $ name
```

The type of a *reference* is the type of the constant which it denotes. Identifiers equated to types, type sets, and equate modules cannot be used as expressions.

9.6. Self

The approacion and evaluates to the object (of guardies band commenced as its its merdian instance within which the experiments is predominal. A l to sheet with minister of a guardian. See Suction 10 (pr tothard

9.7. Procedure, Nacator, and Classics Manage

-----Providence state andy the databast willing a g includes are numbed by expressions of the lowe:

ldn [[actual gamm....]]

The assessing and the providence of a providence stands or type aduate (see Section 12.6).

in the new second second second second When a present on Jacobs, or preside is defined as an a of the souther. The factor has entropy an open

The type of this expression is just the type of the manual re-

9.8. Blad

Chineses may be coulded by the black expression: hind only ([time, mg.,])

```
bind_arg :: m*
```

```
expression
```

```
entity 230 reduces
```

```
1000 . --- H
```

An entity is a simple i we will be a set to get

The sumber of blad es a vincle is a routine bound.

The evaluation of a bind expression proceeds by first evaluating the entity and then evaluating, from left to right, any bind_args that are expressions. The entity may evaluate to a procedure, iterator, handler, or creator object. Suppose that the entity is a procedure or iterator object. (Creator and handler bindings are discussed below.) Then the result is formed by binding the argument objects to the corresponding formals of the entity to form a closure; note that the procedure or iterator is not called when the bind expression is evaluated. When the closure is called, the object denoted by the entity is passed all the bound objects and any actual arguments supplied in the call, all in the corresponding argument positions.

For example, suppose we have:

p = proc(x: T, y: int, w: S) returns(R) signals(too_big)

Then

q := bind p(*, 3 + 4, *)

produces a procedure whose type is proctype(T, S) returns(R) signals(too_big) and assigns it to q. A call of q(a, b) is then equivalent to the call p(a, 7, b).

Bound routines will be stored in stable storage if they are accessible from a stable variable (see Section 13.1). In this case the entity and the *bind args* should denote atomic objects.

There is only one instance of a routine's own data for each parameterization; thus all the bindings of a routine share its own data, if any (see Section 12.7). Each binding is generally a new object; thus the relevant equal operation may treat syntactically identical bindings as distinct.

The semantics of binding a creator or handler are similar to binding a procedure or iterator; the differences arise from argument transmission. Encoding of bound argument objects happens when the bind expression is evaluated and sharing is only preserved among objects bound at the same time (see Section 14). In more detail, the evaluation of a bind expression proceeds by first evaluating the *entity* and then evaluating, from left to right, any *bind_args* that are expressions. Then the argument objects are encoded, from left to right, preserving sharing among these objects. The result is formed by binding the encoded argument objects to the corresponding formals of the entity to form a closure. Note that the entity is not called when the bind expression is evaluated.

When the closure is called, first any other arguments are evaluated and encoded (not sharing with the bound objects) and then the call to the entity is initiated. Decoding of the arguments at the called guardian is done in reverse of the order of encoding; that is, other arguments are decoded before bound arguments and the most recently bound arguments are decoded first. Sharing is preserved on decoding only among groups of bound arguments and among the other arguments, not between groups. Thereafter the call proceeds as normally.

For example, if we execute h1 := bind h(x, y, *) h1(z) then sharing of objects between x and y will be preserved by transmission, but sharing will not be preserved between x and z or y and z.

Closures can be used in equates, provided all the expressions are constants (see Section 7.2.2). However, a handler cannot appear in an equate, since it is not a constant.

9.9. Procedure Calls

Procedure calls have the form:

```
primary ([expression, ....])
```

The primary is evaluated to obtain a procedure object, and then the expressions are evaluated left to right to obtain the argument objects. The procedure is called with these arguments, and the object returned is the result of the entire expression. For more discussion see Section 8.

Any procedure call $p(E_1, ..., E_n)$ must satisfy two constraints to be used as an expression: the type of p must be of the form:

```
proctype (T<sub>1</sub>, ..., T<sub>n</sub>) returns (R) signals (...)
```

and the type of each expression E_i must be included in the corresponding type T_i . The type of the entire call expression is given by R.

9.10. Handler Calls

Handler calls have the form:

```
primary ([expression, ...])
```

The *primary* is evaluated to obtain a handler object, and then the expressions are evaluated left to right to obtain the argument objects. The handler is then called with these arguments as discussed in Section 8.3. The following expressions are examples of handler calls:

```
h(x)
info_guard.who_is_user("john", "doe")
dow_jones.info("XYZ Corporation")
```

Any handler call $h(E_1, ..., E_n)$ must satisfy the following constraints when used as an expression. The type of h must be of the form:

```
handlertype (T<sub>1</sub>, ... T<sub>n</sub>) returns (R) signals (...)
```

and the type of each expression E_i must be included in the corresponding type T_i . The type of the entire call expression is given by R.

As explained in Section 8.3, the execution of a handler call starts by creating a subaction. Therefore an attempt to call a handler from a process that is not running an action is a programming error and will cause the calling guardian to crash. This crash occurs after all of the component expressions have been evaluated.

9.11. Creator Calls

Creator calls have the form:

primary ([expression, ...]) [@ primary]

The first *primary* is evaluated to obtain a creator object, the argument expressions are evaluated left to right to obtain the argument objects, and then the *primary* following the at-sign (@), if present, is evaluated to obtain a node object. If the *primary* following the at-sign is omitted, then node\$/here() is used. The guardian is then created at that node, and the creator called, as discussed in Section 8.4. The following are examples of creator calls:

mailer\$create() @ n spooler[devtype]\$create()

A creator call $c(E_1,...,E_n)$ @n must satisfy the following constraints when used as an expression. The type of c must be of the form:

creatortype (T₁,...,T_n) returns (R) signals (...)

where each T_i includes the type of the corresponding expression E_i . N must be of type node. The type of the entire call expression is given by R.

As with handler calls, an attempt to call a creator from a process that is not running an action will cause the calling guardian to crash after all component expressions have been evaluated.

9.12. Selection Operations

Selection operations provide access to the individual elements or components of a collection. Simple notations are provided for calling the *fetch* operations of array-like types, and the *get* operations of record-like types. In addition, these "syntactic sugarings" for selection operations may be used for user-defined types with the appropriate properties.

9.12.1. Element Selection

An element selection expression has the form:

primary [expression]

This form is just syntactic sugar for a call of a fetch operation, and is computationally equivalent to:

T\$fetch(primary, expression)

where T is the type of the primary. T must provide a procedure operation named fetch, which takes two arguments whose types include the types of primary and expression, and which returns a single result.

9.12.2. Component Selection

The component selection expression has the form:

primary . name

This form is just syntactic sugar for a call of a get_name operation, and is computationally equivalent to:

T\$get name(primary)

where T is the type of primary. T must provide a procedure operation named get name, that takes one

argument and returns a single result. Of course, the type of the procedure's argument must include the type of the *primary*.

9.13. Constructors

Constructors are expressions that enable users to create and initialize sequences, arrays, atomic arrays, structures, records, and atomic records. There are no constructors for user-defined types.

9.13.1. Sequence Constructors

A sequence constructor has the form:

type_spec \$ [[expression , ...]]

The type_spec must name a sequence type: **sequence**[7]. This is the type of the constructed sequence. The expressions are evaluated to obtain the elements of the sequence. They correspond (left to right) to the indexes 1, 2, 3, etc. For a sequence of type **sequence**[7], the type of each element expression in the constructor must be included in *T*.

A sequence constructor is computationally equivalent to a sequence *new* operation, followed by a number of sequence *addh* operations.

9.13.2. Array and Atomic Array Constructors

An array or atomic array constructor has the form:

type spec \$ [[expression :] [expression , ...]]

The type_spec must name an array or atomic array type: array[7] or atomic_array[7]. This is the type of the constructed array. The optional expression preceding the colon (:) must evaluate to an integer, and becomes the low bound of the constructed array or atomic array. If this expression is omitted, the low bound is 1. The optional list of expressions is evaluated to obtain the elements of the array. These expressions correspond (left to right) to the indexes *low_bound*, *low_bound*+1, *low_bound*+2, etc. For an array or atomic array of type array[7] or atomic_array[7], the type of each element expression in the constructor must be included in 7. A constructor of the form array[7]\$[] has a low bound of 1 and no elements.

An array constructor is computationally equivalent to a *create* operation, followed by a number of *addh* operations.

9.13.3. Structure, Record, and Atomic Record Constructors

A structure, record, or atomic record constructor has the form:

```
type_spec $ { field , .... }
```

where

```
field ::= name , ... : expression
```

Whenever a field has more than one name, it is equivalent to a sequence of fields, one for each name. Thus, if R = record[a: Int, b: Int, c: Int], then the following two constructors are equivalent: R\${a, b: p{), c: 9} R\${a: p(), b: p(), c: 9}

In the following we discuss only record constructors; structure and atomic record constructors are similar. In a record constructor, the type specification must name a record type: record[S_1 : T_1 , ..., S_n : T_n]. This is the type of the constructed record. The component names in the field list must be exactly the names S_1 , ..., S_n , although these names may appear in any order. The expressions are evaluated left to right, and there is one evaluation per component name even if several component names are grouped with the same expression. The type of the expression for component S_i must be included in T_i . The results of these evaluations form the components of a newly constructed record. This record is the value of the entire constructor expression.

9.14. Prefix and Infix Operators

Argus allows prefix and infix notation to be used as a shorthand for the operations listed in Table 9-1. The table shows the shorthand form and the computationally equivalent expanded form for each operation. For each operation, the type *T* is the type of the first operand.

Shorthand form	Expansion
expr ₁ ** expr ₂	T \$power(expr 1, expr2)
expr, // expr2	T\$mod(expr ₁ , expr ₂)
expr ₁ / expr ₂	T\$div(expr ₁ , expr ₂)
expr ₁ * expr ₂	T\$mul(expr1, expr2)
expr, expr ₂	T\$concat(expr, expr ₂)
expr ₁ + expr ₂	T\$add(expr., expr.)
expr1 - expr2	T\$sub(expr., expr.)
expr ₁ < expr ₂	T\$k(expr, expra)
expr ₁ <= expr ₂	T\$le(expr ₁ , expr ₂)
$expr_1 = expr_2$	TSequal(expr., expr.)
$expr_1 >= expr_2$	T\$ge(expr., expr.)
$expr_1 > expr_2$	T\$gt(expr,, expr ₂)
expr ₁ ~< expr ₂	$\sim (\exp_1 < \exp_2)$
expr1 ~<= expr2	$\sim (\exp r_1 \le \exp r_2)$
expr ₁ ~= expr ₂	$\sim (\Theta X D r_1 = \Theta X D r_2)$
expr ₁ ~>= expr ₂	$\sim (\exp r_1 >= \exp r_2)$
expr1 ~> expr2	$\sim (expr_1 > expr_2)$
expr ₁ & expr ₂	TSand(expr., expr.)
expr ₁ expr ₂	T\$or(expr ₁ , expr ₂)
- expr	T\$minus(expr)
~ expr	T\$not(expr)

Table 9-1: Prefix and Infix Operators: shorthands and expansions

Operator notation is used most heavily for the built-in types, but may be used for user-defined types as well. When these operations are provided for user-defined types, they should be free of side-effects, and

Expressions

they should mean roughly the same thing as they do for the built-in types. For example, the comparison operations should only be used for types that have a natural partial or total order. Usually, the comparison operations (*It*, *Ie*, *equal*, *ge*, *gt*) will be of type

proctype (T, T) returns (bool)

the other binary operations (e.g., add, sub) will be of type

```
proctype (T, T) returns (T) signals (...)
```

and the unary operations will be of type

proctype (T) returns (T) signals (...)

9.15. Cand and Cor

Two additional binary operators are provided. These are the *conditional and* operator, cand, and the *conditional or* operator, cor. The result of evaluating:

expression, cand expression,

is the boolean and of expression₁ and expression₂. However, if expression₁ is false, expression₂ is never evaluated. The result of evaluating:

expression, cor expression₂

is the boolean or of expression, and expression, but expression, is not evaluated unless expression, is false. For both cand and cor, expression, and expression, must have type bool.

Because of the conditional expression evaluation involved, uses of cand and cor are not equivalent to any procedure call.

9.16. Precedence

When an expression is not fully parenthesized, the proper nesting of subexpressions might be ambiguous. The following precedence rules are used to resolve such ambiguity. The precedence of each infix operator is given in the table below. Higher precedence operations are performed first. Prefix operators always have precedence over infix operators.

Table 9-2: Precedence for Infix Oper	ators
--------------------------------------	-------

5 ** 4 * / // 3 + -
4 * / // 3 + -
3 + - 2
0
1 & cand
0 cor

9.16 Precedence

The order of evaluation for operators of the same precedence is left to right, except for **, which is right to left.

- Land Barry and Carl State Barry and

化建筑调制

9.17. Up and Down

There are no implicit type conversions in Argus. Two forms of expression exist for explicit conversions. These are:

up (expression) down (expression)

Up and down may be used only within the body of a cluster operation (see Section 12.3). Up changes the type of the expression from the representation type of the cluster to the abstract type. Down converts the type of the expression from the abstract type to the representation type.

10. Statements

In this section, we discuss must of the eleterments of Argus, simplicating the interaction of actions and the various kinds of control flow eleterments. We participate discussion of Pro signal, and, and except statements, which are used for signaling and hending encoditoring and flowing the first of the complete syntax of statements.

Atomic actions allow sequences of statements to appear to be included to other actions. Sequences of statements that are not within an action are encoded debuilds that is, other processes may observe intermediate states between statements. Statements are encoded for that side-effects and do not return any values. Most statements are control elemented in an action of the programmer to dreate processes and to dictate how control flows in a process. The real are adapted becaused: analyzements and calls (ase Section 8).

A control statement can control a group of equation, declarations, and statements rather than just a single statement. Such a group is called a foody, and has the hour:

body ::= { equate }

{ statement }

Note that statements include declarations (see Sections 7.1.2 and Appendix I). No special terminator is needed to signify the and of a lange; constant control and a sector to delimit the bodies. The statements in a body are encoded a sector statement in the sector.

10.1. Calls

A call statement may be used to call a precedure, handler, or creator. For procedures and handlers its form is the same as a call expression:

primary ([supression ,])

The primary must be a procession, or handler abject. The type of each actual expression must be included in the type of the concentrating formal argument. The procession or handler may or may not return results; if it does return repute, they are disconded.

For creator calls the syntax is similar, but one can optionally specify the node at which the guardian is to be created:

primary ([expression ,]) [@ primary] The primary following the st-sign (@) must be of type made.

The details of procedure, handler, and creater calls are decoding in Bestions 8.2, 8.3, and 8.4.

Statements

10.2. Update Statements

Two special statements are provided for updating components of record and array-like objects. In addition they may be used with user-defined types with the appropriate properties. These statements resemble assignments syntactically, but are actually call statements.

10.2.1. Element Update

The element update statement has the form:

primary [expression,] := expression,

This form is merely syntactic sugar for a call of a store operation; it is equivalent to the call statement:

T\$store(primary, expression, expression₂)

where T is the type of the primary. T must provide a procedure named store that takes three arguments whose types include those of primary, expression, and expression, respectively.

10.2.2. Component Update

The component update statement has the form:

primary . name := expression

This form is syntactic sugar for a call of a *set_* operation whose name is formed by attaching *set_* to the name given. For example, if the name is *f*, then the statement above is equivalent to the call statement:

T\$set_f(primary, expression)

where *T* is the type of the *primary*. *T* must provide a procedure operation named set_*f*, where *f* is the name given in the component update statement. This procedure must take two arguments whose types include the types of *primary* and *expression*, respectively.

10.3. Block Statement

The block statement permits a sequence of statements to be grouped together into a single statement. Its form is:

begin body end

Since the syntax already permits bodies inside control statements, the main use of the block statement is to group statements together for use with the **except** statement (see Section 11).

10.4. Fork Statement

A fork statement creates an autonomous process. The fork statement has the form:

fork primary ([expression,])

where the *primary* is a procedure object whose type has no results or signals (see Section 12.1). The type of each actual *expression* must be included in the type of the corresponding formal.

Execution of the fork statement starts by evaluating the primary and actual argument expressions from left to right. Any exceptions raised by the evaluation of the primary or the expressions are raised by the fork statement. If no exceptions are raised, then a new process is created and execution resumes after

58

the fork statement in the old process. The new process starts by calling the given procedure with the argument objects. This new process terminates if and when the procedure call does. However, if the guardian crashes the process goes away (like any other process).

Note that the new process does not run in an action, although the procedure called can start a topaction if desired. There is no mechanism for waiting for the termination of the new process. The procedure called in a fork statement cannot return any results or signal any exceptions.

10.5. Enter Statement

Sequential actions are created by means of the enter statement, which has two forms:

enter topaction body end

and

enter action body end

The topaction qualifier causes the body to execute as a new top-level action. The action qualifier causes the body to execute as a subaction of the current action; an attempt to execute an enter action statement in a process that is not executing an action is a programming error and causes the guardian to crash. When the body terminates, it does so either by committing or aborting. Normal completion of the body results in the action committing. Statements that transfer control out of the enter statment (exit, leave, break, continue, return, signal, and resignal) normally commit the action unless are prefixed with abort (e.g., abort exit). Two-phase commit of a topaction may fail, in which case the enter topaction statement raises an *unavailable* exception.

10.6. Coenter Statement

Concurrent actions and processes are created by means of the coenter statement:

```
coenter coarm { coarm } end
```

where

```
coarm ::= armtag [ foreach decl , ... in call ]
body
armtag ::= action
```

topaction process

Execution of the coenter starts by creating all of the coarm processes, sequentially, in textual order. A foreach clause indicates that multiple instances of the coarm will be created. The call in a foreach clause must be an iterator call. At each yield of the iterator, a new coarm process is created and the objects yielded are assigned to newly declared variables in that process. (This implicit assignment must be legal, see Section 6.1.) Each coarm process has separate, local instances of the variables declared in the foreach clause.

Statements

The process executing the **coenter** is suspended until after the **coenter** is finished. Once all coarm processes are created, they are started simultaneously as concurrent siblings. Each coarm instance runs in a separate process, and each coarm with an *armtag* of topaction or action executes within a new top-level action or subaction, respectively. An attempt to execute a **coenter** with a **process** coarm when in an action, or to execute a **coenter** with an **action** coarm when not in an action is an error and will cause the guardian to crash (see Table 10-1).

Table 10-1: Legality of coenter statements.

	process executing the coenter is:	
armtag	not in an action	running an action
action	not legal	legal
topaction	legal	legal
process	legal	not legal

A simple example making use of foreach is:

coenter action foreach i: Int in Int\$from to (1, 5)

p (i) end

which creates five processes, each with a local variable *i*, having the value 1 in the first process, 2 in the second process, and so on. Each process runs in a newly created subaction. This statement is legal only if the process executing it is running an action.

A coarm may terminate without terminating the entire **coenter** (and sibling coarms) either by normal completion of its body, or by executing a **leave** statement (see Section 10.7). The commit of a coarm declared as a topaction may terminate in an *unavailable* exception if two-phase commit fails. Such an exception can only be handled outside the **coenter** statement, and thus will force termination of the entire **coenter** (as explained below).

A coarm may also terminate by transferring control outside the coenter statement. When such a transfer of control occurs, the following steps take place.

- 1. Any containing statements are terminated divisibly, to the outermost level of the coarm, at which point the coarm becomes the *controlling* coarm.
- Once there is a controlling coarm, every other active coarm will be terminated (and abort if declared as an action) as soon as it leaves all selze statements; the controlling coarm is suspended until all other coarms terminate.
- The controlling coarm then commits or aborts if declared as an action; if declared as a topaction and the two-phase commit fails, an *unavailable* exception is raised by the coenter statement.
- 4. Finally, the entire coenter terminates, and control flow continues outside the coenter statement.

Divisible termination implies, for instance, that a nested topaction may commit while its parent action aborts.

60

A simple example of early termination is reading from a replicated database, where any copy can supply the necessary information:

a last the start is a \$454

```
coenter action foreach db: database in ait_replicas (...)
return( database$read (db))
end
```

When one of these coarms completes first, it tries to commit itself and abort the others. The aborts take place immediately (since there are no seize statements); it is not necessary for the handler calls to finish. It is possible that some descendants of an aborted coarm may be running at remote sites when the coarm aborts; the Argus system ensures that such orphans will be aborted before they can make their presence known or detect that they are in fact orphans (see Section 2.5).

10.7. Leave Statement

The leave statement has the form:

[abort] ieave

Executing a leave statement terminates the innermost enter statement or coenter coarm in which it appears. If the process terminated is an action, then it commits unless the abort qualifier is present, in which case the action aborts. The abort qualifier can only be used textually within an enter statement or within an action or topaction coarm of a coenter statement.

Note that unlike the other control flow statements, leave does not affect concurrent siblings in a coenter (see Section 10.6).

10.8. Return Statement

The form of the return statement is:

[abort] return [(expression,)]

The return statement terminates execution of the containing routine. If the return statement occurs in an iterator no results can be returned. If the return statement is in a procedure, handler, or creator the type of each *expression* must be included in the corresponding return type of the routine. The *expressions* (if any) are evaluated from left to right, and the objects obtained become the results of the routine.

If no abort qualifier is present, then all containing actions (If any) terminated by this statement are committed. If the abort qualifier is present, then all terminated actions are aborted. Note that unlike the leave statement, return will abort concurrent siblings if executed within a coarm of a coenter statement (see Section 10.6). The abort qualifier can only be used textually within an enter statement, an action or topaction coarm of a coenter statement, or the body of a handler or creator.

Within a handler or creator, the result objects are encoded just before the activation action terminates, but after all control flow and nested action termination. If encoding of any result object terminates in a *failure* exception, then the activation action aborts and the handler or creator terminates with the same exception.

10.9. Yield Statement

The form of a yield statement is:

yield [(expression , ...)]

The yield statement may occur only in the body of an iterator. The effect of a yield statement is to suspend execution of the iterator invocation, and return control to the calling for statement or foreach clause. The values obtained by evaluating the *expressions* (left to right) are passed back to the caller. The type of each *expression* must be included in the corresponding yield type of the iterator. Upon resumption, execution of the iterator continues at the statement following the yield statement.

A yield statement cannot appear textually inside an enter, coenter, or seize statement.

10.10. Conditional Statement

The form of the conditional statement is:

```
If expression then body
```

```
{ elself expression then body }
[ else body ]
```

end

The *expressions* must be of type **bool**. They are evaluated successively until one is found to be true. The *body* corresponding to the first true expression is executed, and the execution of the H statement then terminates. If there is an **else** clause and if none of the *expressions* is true, then the *body* in the else clause is executed.

10.11. While Statement

The while statement has the form:

while expression do body end

Its effect is to repeatedly execute the *body* as long as the *expression* remains true. The *expression* must be of type **bool**. If the value of the expression is true, the body is executed, and then the entire while statement is executed again. When the expression evaluates to false, execution of the while statement terminates.

10.12. For Statement

An iterator (see Section 12.2) can be called by a for statement. The iterator produces a sequence of *items* (where an item is a group of zero or more objects) one item at a time; the *body* of the for statement is executed for each item in the sequence.

The for statement has the form:

```
for [deci, ....] in call do body end
```

or

for [idn , ...] in call do body end
10.12 For Statement

The call must be an iterator call. The second form (with an *idn* list) uses distinct, previously declared variables to serve as the loop variables, while the first form (with a *decl* list) form introduces new variables, local to the for statement, for this purpose. In either case, the type of each variable must include the corresponding yield type of the called iterator (see Section 12.2) and the number of variables must also match the yield type.

Execution of the for statement begins by calling the iterator, which either yields an item or terminates. If it yields an item (by executing a yield statement), its execution is temporarily suspended, the objects in the item are assigned to the loop variables, and the body of the for statement is executed. The next cycle of the loop is begun by resuming execution of the iterator after the yield statement which suspended it. Whenever the iterator terminates, the entire for statement terminates.

10.13. Break and Continue Statements

The break statement has the form:

[abort] break

Its effect is to terminate execution of the smallest for or while loop statement in which it appears. Execution continues with the statement following that loop.

The continue statement has the form:

[abort] continue

Its effect is to start the next cycle (if any) of the smallest for or while loop statement in which it appears.

Terminating a cycle of a loop may also terminate one or more containing actions. If no abort qualifier is present, then all these terminated actions (if any) are committed. If the abort qualifier is present, then all of the terminated actions are aborted. Unlike leave, break and continue will abort concurrent sibling actions when control flow leaves a containing coenter (see Section 10.6).

The abort qualifier can only be used textually within an enter statement or an action or topaction coarm of a coenter statement.

10.14. Tagcase Statement

The tagcase statement can be used to decompose one of and variant objects; atomic_variant objects can be decomposed with the tagtest or tagwait statements. The decomposition is indivisible for variant objects; thus, use of the tagcase statement for variants is not equivalent to using a conditional statement in combination with *is* and *value* operations (see Section II.15).

The form of the tagcase statement is:

```
tagcase expression
tag_arm { tag_arm }
[ others : body ]
end
```

```
where
```

tag_arm ::= tag name , ... [(idn: type_spec)] : body

The *expression* must evaluate to a **oneof** or **variant** object. The tag of this object is then matched against the names on the *tag_arms*. When a match is found, if a declaration (*idn: type_spec*) exists, the value component of the object is assigned to the new local variable *idn*. The matching *body* is then executed; *idn* is defined only in that body. If no match is found, the *body* in the others arm is executed.

In a syntactically correct tagcase statement, the following three constraints are satisfied.

- 1. The type of the expression must be some oneof or variant type, T.
- 2. The tags named in the tag_arms must be a subset of the tags of T, and no tag may occur more than once.
- 3. If all tags of T are present, there is no others arm; otherwise an others arm must be present.

On any tag_arm containing a declaration (*idn: type_spec*), *type_spec* must include the type(s) of T corresponding to the tag or tags named in that *tag arm*.

10.15. Tagtest and Tagwait Statements

The tagtest and tagwait statements are provided for decomposing atomic_variant objects, permitting the selection of a body based on the tag of the object to be made indivisibly with the testing or acquisition of specified locks.

10.15.1. Tagtest Statement

The form of the tagtest statement is:

```
tagtest expression
  atag_arm { atag_arm }
  [ others : body ]
  end
```

where

```
atag_arm ::= tag_kind name , ... [ ( idn: type_spec ) ] : body
tag_kind ::= tag
```

wtag

The *expression* must evaluate to an **atomic_variant** object. If a read lock could be obtained on the **atomic_variant** object by the current action, then the tag of the object is matched against the names on the *atag_arms*; otherwise the others arm, if present, is executed. If a matching name is found, then the *tag kind* is considered.

- If the tag_kind is tag, a read lock is obtained on the object and the match is complete.
- If the tag_kind is wtag and the current action can obtain a write lock on the object, then a write lock is obtained and the match is complete.

When a complete match is found, if a declaration (*idn: type_spec*) exists, the value component of the object is assigned to the new local variable *idn*. The matching *body* is then executed; *idn* is defined only in that body. The entire matching process, including testing and acquisition of locks, is indivisible.

10.15.1 Tagtest Statement

If a complete match is not found, or the object was not readable by the action, then the others arm (if any) is executed; if there is no others arm, the tagtest statement terminates. If no complete match is found, then no locks are acquired.

8400

The tagtest statement will only obtain a lock if it is possible to do so without "waiting". For example, suppose that the internal state of the **atomic__variant** indicates that some previous action acquired a conflicting lock. This action may have since aborted, or may have committed up to an ancestor of the action executing the tagtest, but determining such facts may require system-level communication to other guardians. In this case the tagtest statement may give misleading information, because it may not indicate a match. Apparent anomalies in testing locks may occur even if the action executing the tagtest "knows" that the lock can be acquired, so that the use of tagtest to avoid deadlocks or long delays may result in excessive aborts.

10.15.2. Tagwait Statement

The form of the tagwalt statement is:

```
tagwait expression
atag_arm { atag_arm }
```

end

Execution of the tagwait statement proceeds as for the tagtest statement, but if no complete match is found, or if the object is not readable by the current action, then the entire matching process is repeated (after a system-controlled delay), until a complete match is found. Although there is no others arm in a tagwait statement, all tag names do not have to be listed.

10.15.3. Common Constraints

Tagtest and tagwalt statements may be executed only within an action. An attempt to execute a tagtest or tagwalt statement in a process that is not executing an action is an error and will cause the guardian to crash after evaluating the *expression*.

- In a syntactically correct tagtest or tagwalt statement, the following three constraints are satisfied.
 - 1. The type of the expression must be some atomic_variant type, T.
 - 2. The tags named in the *atag_arms* must be a subset of the tags of *T*, and no tag may occur more than once.
 - 3. Finally, on any *atag_arm* containing a declaration (*idn: type_spec*), *type_spec* must include the type(s) specified as corresponding in T to the tag or tags named in the *atag_arm*.

A simple example of a tagtest statement is garbage collecting the elements of an array that are in the dequeued state:

```
item - stemic_variantjanqueued: int, dequeued: sullj
for i: item in anagitumplatumenta() de
tempet i
```

```
tag darpmand: annajitarnjirarni(i
othare: brask
and
```

10.16. Seize Statement

The online statement has the form:

tales aproved as the party and

The expression must evaluate to a matter object. The example, process then attempts to gain procession of that matter object, and value to do to if examples. Only one process, whether user or system defined, may process a grant states object at each time. Deal discussions whether user or body of the eaters statement is exampled. When the body to this body and a procession of the texture is released. This includus termination of the body by electronic time the body out of the texture.

The body of a cales statement is considered to be a callest smaller: a process executing in the body of a cales distanced as only be faultly transmissible analysis for gradient to process is running. See Section 15 for the supreme for this state the state distance fille upon a mutan.

Multiple, needed acknes of the same display aligned are aligned, and shall and south for the A process solubly a mater that it has already ackned will not deadlock with family and processing its net worth released with the outermost solue terminates.

10.17. Pause Statement

The passes statement has the lerm:

The process statement must occur within an enclosing axists datament. Its affect is to release the matex object associated with the annulast controlog axists datament. Associated associates of the process for a system-controlied period of time, and then regular processing and antitality and another.

If multiple, nested seizes on the mater object have been parlement, prove will not actually release possession. For example, possession is not released in the following:

estas na de seize na do

une % does not really release possession

and

In general, needed solves should be availed when pause must be word, and pause should be avoided when needed values must be used.

10.18. Terminate Statement

NUMBER OF CONTRACTOR OF STREET

The terminate statement may occur only within a guardian definition (see Sect 13). The form of a terminate statement is:

terminate

and the second and the second and the second s

When executed within an action, its effect is to cause the eventual destruction of the guardian after the enclosing action commits to the top. If a process attempts to execute terminate while not running an action, a topaction is created to execute the terminate and immediately commit.

Let A be the action that is executing the terminate. The effect of this statement is the following:

- Action A must wait until the action that created the guardian is committed relative to A. In the case of a permanent guardian whose creation has accommitted to the top there will be no wait, but for a recently created guardian there may be a delay.
- 2. If multiple processes are attempting to execute terminate statements, at most one at at time may proceed to the next step.
- 3. If A commits to the top, the guardian will be destroyed at some time after topaction commit. If some ancestor of A aborts, however, the guardian will be unaffected. The guardian is also unaffected during the time between A executing terminate and A committing to the top.

In order to avoid serialization problems, creation or destruction of a guardian must be synchronized with use of that guardian via atomic objects such as the catalog (see Section 3.4).

11. Exception Handling and Exits

A routine is designed to perform a certain task. However, in some cases that task may be impossible to perform. In such a case, instead of returning normally (which would imply successful performance of the intended task), the routine should notify its caller by signalling an *exception*, consisting of a descriptive name and zero or more result objects.

The exception handling mechanism consists of two parts: signalling exceptions and handling exceptions. Signalling is the way a routine notifies its caller of an exceptional condition; handling is the way the caller responds to such notification. A signalled exception always goes to the immediate caller, and the exception must be handled in that caller. When a routine signals an exception, the current activation of that routine terminates and the corresponding call (in the caller) is said to *raise* the exception. When a call raises an exception, control immediately transfers to the closest applicable exception handler. Exception handlers are attached to statements; when execution of the exception handler is attached. For brevity, exception handlers will be called "handlers" in this chapter; these should not be confused with the remote call handlers of guardians (see Section 13).

11.1. Signal Statement

An exception is signalled with a signal statement, which has the form:

[abort] signal name [(expression , ...)]

A signal statement may appear anywhere in the body of a routine. The execution of a signal statement begins with evaluation of the expressions (if any), from left to right, to produce a list of exception results. The activation of the routine is then terminated. Execution continues in the caller as described in Section 11.2 below.

The exception name must be one of the exception names listed in the routine heading. If the corresponding exception specification in the heading has the form:

name(T_1, \ldots, T_n)

then there must be exactly *n* expressions in the signal statement, and the type of the *i*th expression must be included in T_i .

If no abort qualifier is present, then all containing actions (if any) terminated by this statement are committed. If the abort qualifier is present, then all terminated actions are aborted. Unlike the leave statement, signal will terminate (abort) concurrent siblings if executed within a coenter statement (see Section 10.6). The abort qualifier can only be used textually within an enter statement, an action or topaction coarm of a coenter statement, or the body of a handler or creator.

Within a handler or creator, the result objects are encoded just before the activation action terminates, but after termination of all control flow and nested actions. If encoding of any result object terminates in a *failure* exception, then the activation action aborts and the handler or creator terminates with the *failure* exception.

11.2. Except Statement

When a routine activation terminates by signalling an exception, the called routine is said to raise that exception. By attaching exception handlers to statements, the caller can specify the action to be taken when an exception is raised by a call within a statement or by the statement itself.

A statement with handlers attached is called an except statement, and has the form:

statement except { when_handler } [others_handler] end

where

```
when_handler ::= when name , ... [ ( decl , ... ) ] : body
when name , ... ( * ) : body
```

others_handler ::= others [(idn : string)] : body

Let S be the statement to which the handlers are attached, and let X be the entire except statement. Each when handler specifies one or more exception names and a body. The body is executed if an exception with one of those names is raised by a call in S. Each of the names listed in the when handlers must be distinct. The optional others handler is used to handle all exceptions not explicitly named in the when handlers. The statement S can be any form of statement, and can even be another except statement. As an example, consider the following except statement:

```
m.send_mail(user, my_message)
except when no_such_user: ... % body 1
when unavailable, failure (s: string): ... % body 2
when others (ename: string): ... % body 3
end
```

This statement handles exceptions arising from a remote call. If the call raises a *no_such_user* exception, then "body 1" will be executed. If the call raises a *failure* or *unavailable* exception, then "body 2" will be executed. Any other exception will be handled by "body 3."

If, during the execution of S, some call in S raises an exception E, control transfers to the textually closest handler for E that is attached to a statement containing the call. When execution of the handler completes, control passes to the statement following the one to which the handler is attached. Thus if the closest handler is attached to S, the statement following X is executed next. If execution of S completes without raising an exception, the attached handlers are not executed.

An exception raised inside a handler is treated the same as any other exception: control passes to the closest handler for that exception. Note that an exception raised in some handler attached to S cannot be handled by any handler attached to S; the exception can be handled within the handler, or it can be handled by some handler attached to a statement containing X. For example, in the following except statement:

times3_plus1(a) except when limits: a := a + a when overflow: ... % body 2 end

any overflow signal raised by the expression a + a will not be handled in "body 2," because this overflow handler is not in an except statement attached to the assignment statement a := a + a.

We now consider the forms of exception handlers in more detail. The form:

when name , [(deci ,)] : body

is used to handle exceptions with the given names when the exception results are of interest. The optional declared variables, which are local to the handler, are assigned the exception results before the body is executed. Every exception potentially handled by this form must have the same number of results as there are declared variables, and the types of the variables must include the types of the results. The form:

when name, (*): body

handles all exceptions with the given names, regardless of whether or not there are exception results; any actual results are discarded. Using this form, exceptions with differing numbers and types of results can be handled together.

The form:

others [(idn:string)]:body

is optional, and must appear last in a handler list. This form handles any exception not handled by other handlers in the list. If a variable is declared, it must be of type string. The variable, which is local to the handler, is assigned a lower case string representing the actual exception name; any results are discarded.

Note that number and type of exception results are ignored when matching exceptions to handlers; only the names of exceptions are used. Thus the following is illegal, in that int\$div signals zero_divide without any results (see Section II.4), but the closest handler has a declared variable:

begin

```
y: Int := 0

x: Int := 3 / y

except when zero_divide (z: Int): return end

end

except when zero_divide: return end
```

A call need not be surrounded by except statements that handle all potential exceptions. In many cases the programmer can prove that a particular exception will not arise; for example, the call Int div(x, 7) will never signal zero_divide. However, if some call raises an exception for which there is no handler, then the guardian crashes due to this error⁹.

⁹The implementation of the Argus should log unhandled exceptions in some fashion, to aid later debugging. During debugging, an unhandled exception would be trapped by the debugger before the crash.

11.3. Resignal Statement

A resignal statement is a syntactically abbreviated form of exception handling:

statement [abort] resignal name , ...

Each name listed must be distinct, and each must be one of the condition names listed in the routine heading. The resignal statement acts like an except statement containing a handler for each condition named, where each handler simply signals that exception with exactly the same results. Thus, if the resignal clause names an exception with a specification in the routine heading of the form:

 $name(T_1, \ldots, T_n)$

then effectively there is a handler of the form:

when name $(x_1: T_1, ..., x_n: T_n)$: [abort] signal name $(x_1, ..., x_n)$ which has an abort qualifier if and only if the resignal statement did. As for an explicit handler of this form, every exception potentially handled by this implicit handler must have the same number of results as declared in the exception specification, and the types of the results must be included in the types listed in the exception specification.

If no abort qualifier is present, then all containing actions (if any) terminated by this statement are committed. If the abort qualifier is present, then all terminated actions are aborted. Unlike the leave statement, resignal will abort concurrent siblings if executed within a coenter statement (see Section 10.6). The abort qualifier can only be used textually within an enter statement, an action or topaction coarm of a coenter statement, or the body of a handler or creator.

11.4. Exit Statement

An exit statement has the form:

abort exit name (expression)

An exit statement is similar to a signal statement except that where the signal statement signals an exception to the *calling* routine, the exit statement raises the exception directly in the *current* routine. Thus an exit causes a transfer of control within a routine but does not terminate the routine. An exception raised by an exit statement must be handled explicitly by a containing except statement with a handler of the form:

when name , [(decl ,)] : body

As usual, the types of the expressions in the **exit** statement must be included in the types of the variables declared in the handler. The handler must be an explicit one, i.e., exits to the implicit handlers of resignal statements are illegal.

If no abort qualifier is present, then all containing actions (if any) terminated by the exit statement are committed. If the abort qualifier is present, then all terminated actions are aborted. Unlike the leave statement, exit will abort concurrent siblings when control flow leaves a containing coenter statement (see Section 10.6). The abort qualifier can only be used textually within an enter statement or an action or topaction coarm of a coenter statement.

11.4 Exit Statement

The exit statement and the signal statement mesh nicely to form a uniform mechanism. The signal statement can be viewed simply as terminating a routine activation; an exit is then performed at the point of invocation in the caller. (Because this exit is implicit, it is not subject to the restrictions on exits listed above.)

11.5. Exceptions and Actions

A new action is created by a handler call, creator call, enter statement, or action or topaction arm of a coenter statement. In addition, the recover code of a guardian runs as an action. When control flows out of an action, that action is committed unless action is taken to prevent its committing. To abort an action, it is necessary to qualify control flow statements such as exit, signal, resignal, and leave with the keyword abort (see Section 10).

However, there is an additional complication. Not only will explicit termination of actions by exit, signal, and resignal statements commit actions, but also *implicit* termination by flow of control out of an action body when an exception raised within that body is handled outside the action's body. Thus, if an exception which is raised by a call within an action is not to commit the action, then it is necessary to catch the exception within the action. This is particularly important when dealing with topactions. A common desire is to catch all "unexpected" exceptions, but still have the topaction abort. In this case, the catch-all exception handler must be placed inside the topaction. However, an *unavailable* handler must still be placed outside the topaction, since the two-phase commit may fail.

An action or topaction coarm of a coenter statement will not abort its concurrent siblings when it ends in either normal completion of its body or by a leave statement. However, if control flows otherwise out of the coenter statement from within one of the coarms, the entire coenter is terminated as described in Section 10.6. Thus, a coenter statement should must be used carefully to ensure the proper behavior in case of exceptions. There may be circumstances where a separate exception handler will have to be used for each coarm to ensure the proper behavior, even when the exception handling is identical for each coarm.

11.6. Failure Exceptions

Argus responds to unhandled exceptions differently than CLU. In CLU, an unhandled exception in some routine causes that routine to terminate with the *failure* exception. In Argus, however, an unhandled exception causes the guardian that is running the routine to crash. Our motivation for this change is that an unhandled exception is typically a symptom of a programming error that cannot be handled by the calling routine. Furthermore, crashing the guardian limits the damage that the programming error causes.

Procedures and iterators in Argus no longer have an implicit failure exception associated with them. Instead, such a routine may list failure explicitly in its signals clause and failure may have any number (and type) of exception results. Failure should be used to indicate an unexpected (and possibly catastrophic) failure of a lower-level abstraction, for example, when there is a failure in a type parameter's routines (for instance in *similar* or *copy* operations). Another example is when there is an unwanted side effect, such as a bounds exception in array[t]Selements caused by a mutation of the array argument. Various operations of the built-in types signal failure under such circumstances.

والمعاد والمراجع وال

For handlers and creators, *failure* is used to indicate that a remote call has failed; thus the exception *failure*(string) is implicit in the type of every handler and creator (see Section 13.5). When a remote call terminates with the *failure* exception, this means that not only has this call failed, but that the call is unlikely to succeed if repeated.

74

12. Modules

Besides guardian modules, Argus has procedure, iterator, cluster, and equate modules.

module ::= { equate } guardian

{ equate } procedure { equate } iterator { equate } cluster

{ equate } equates

Guardians are discussed in Section 13, the rest are described below.

12.1. Procedures

A procedure performs an action on zero or more *arguments*, and when it terminates it returns zero or more *results*. A procedure implements a *procedural abstraction*: a mapping from a set of argument objects to a set of result objects, with possible modification of some of the argument objects. A procedure may terminate in one of a number of *conditions*; one of these is the *normal condition*, while others are *exceptional conditions*. Differing numbers and types of results may be returned in the different conditions.

The form of a procedure is:

```
idn = proc [ parms ] args [ returns ] [ signals ] [ where ]
routine_body
end idn
```

where

```
      args
      ::= ( [ decl , ... ] )

      returns
      ::= returns ( type_spec , ... )

      signals
      ::= signals ( exception , ... )

      exception
      ::= name [ ( type_spec , ... ) ]

      routine_body
      ::= { equate }

      { own_var }
      { statement }
```

In this section we discuss non-parameterized procedures, in which the *parms* and *where* clauses are missing. Parameterized modules are discussed in Section 12.5. Own variables are discussed in Section 12.7.

The heading of a procedure describes the way in which the procedure communicates with its caller. The args clause specifies the number, order, and types of arguments required to call the procedure, while the *returns* clause specifies the number, order, and types of results returned when the procedure terminates normally (by executing a return statement or reaching the end of its body). A missing returns clause indicates that no results are returned.

The signals clause names the exceptional conditions in which the procedure can terminate, and specifies the number, order, and types of result objects returned in each condition. All names of

exceptions in the *signals* clause must be distinct. The *idn* following the end of the procedure must be the same as the *idn* naming the procedure.

A procedure is an object of some procedure type. For a non-parameterized procedure, this type is derived from the procedure heading by removing the procedure name, rewriting the formal argument declarations with one *idn* per *decl*, deleting the *idns* of all formal arguments, and finally, replacing proc by proctype.

The call of a procedure causes the introduction of the formal variables, and the actual arguments are assigned to these variables. Then the procedure body is executed. Execution terminates when a return statement or a signal statement is executed, or when the textual end of the body is reached. If a procedure that should return results reaches the textual end of the body, the guardian crashes due to this error. At termination the result objects, if any, are passed back to the caller of the procedure.

12.2. Iterators

An iterator computes a sequence of *items*, one item at a time, where an item is a group of zero or more objects. In the generation of such a sequence, the computation of each item of the sequence is usually controlled by information about what previous items have been produced. Such information and the way it controls the production of items is local to the iterator. The user of the iterator is not concerned with how the items are produced, but simply uses them (through a for statement) as they are produced. Thus the iterator abstracts from the details of how the production of the items is controlled; for this reason, we consider an iterator to implement a control abstraction. Iterators are particularly useful as operations of data abstractions that are collections of objects (e.g., sets), since they may produce the objects in a collection without revealing how the collection is represented.

An iterator has the form:

```
idn = Iter [ parms ] args [ yields ] [ signais ] [ where ]
routine_body
end idn
```

where

```
yields ::= yields (type_spec, ...)
```

In this section we discuss non-parameterized iterators, in which the *parms* and *where* clauses are missing. Parameterized modules are discussed in Section 12.5. Own variables are discussed in Section 12.7.

The form of an iterator is similar to the form of a procedure. There are only two differences:

- 1. An iterator has a yields clause in its heading in place of the returns clause of a procedure. The yields clause specifies the number, order, and types of objects yielded each time the iterator produces the next item in the sequence. If zero objects are yielded, then the yields clause is omitted. The *idn* following the end of the iterator must be the same as the *idn* naming the iterator.
- 2. Within the iterator body, the yield statement is used to present the caller with the next item

in the sequence. An iterator terminates in the same manner as a procedure, but it may not return any results.

An iterator is an object of some iterator type. For a non-parameterized iterator, this type is derived from the iterator heading by removing the iterator name, rewriting the formal argument declarations with one idn per decl, deleting the idns of all formal arguments, and finally, replacing iter by itertype.

An iterator can be called only by a for statement or by a foreach clause in a coenter statement.

12.3. Clusters

A cluster is used to implement a new data type, distinct from any other built-in or user-defined data type. A data type (or data abstraction) consists of a set of objects and a set of primitive operations. The primitive operations provide the most basic ways of manipulating the objects; ultimately every computation that can be performed on the objects must be expressed in terms of the primitive operations. Thus the primitive operations define the lowest level of observable object behavior¹⁰.

The form of a cluster is:

```
idn - cluster [ parms ] is opidn , ... [ where ]
cluster_body
end idn
```

where

```
opidn ::= idn

| transmit

cluster_body ::= { equate } rep = type_spec { equate }

{ own_var }

routine { routine }
```

```
routine :::= procedure
```

In this section we discuss non-parameterized clusters, in which the *parms* and *where* clauses are missing. Parameterized modules are discussed in Section 12.5. Own variables are discussed in Section 12.7.

The primitive operations are named by the list of *opidns* following the reserved word **Is**. All of the *opidns* in this list must be distinct. The *idn* following the end of the cluster must be the same as the *idn* naming the cluster.

To define a new data type, it is necessary to choose a *concrete representation* for the objects of the type. The special equate:

¹⁰Readers not familiar with the concept of data abstraction might read Liskov, B. and Guttag, J., Abstraction and Specification in Program Development, MIT Press, Cambridge, 1986.

Modules

rep = type_spec

within the cluster body identifies the *type_spec* as the concrete representation. Within the cluster, rep may be used as an abbreviation for this *type_spec*.

The identifier naming the cluster is available for use in the cluster body. Use of this identifier within the cluster body permits the definition of recursive types.

In addition to giving the representation of objects, the cluster must implement the primitive operations of the type. One exception to this, however, is the transmit operation. The transmit operation is not directly implemented by a cluster; instead, the cluster must implement two operations: *encode* and *decode* (see Section 14 for details). The primitive operations may be either procedural or control abstractions; they are implemented by procedures and iterators, respectively. Any additional routines implemented within the cluster are *hidden*: they are private to the cluster and may not be named directly by users of the abstract type. All the routines must be named by distinct identifiers; the scope of these identifiers is the entire cluster.

Outside the cluster, the type's objects may only be treated abstractly (i.e., manipulated by using the primitive operations). To implement the operations, however, it is usually necessary to manipulate the objects in terms of their concrete representation. It is also convenient sometimes to manipulate the objects abstractly. Therefore, inside the cluster it is possible to view the type's objects either abstractly or in terms of their representation. The syntax is defined to specify unambiguously, for each variable that refers to one of the type's objects, which view is being taken. Thus, inside a cluster named T, a declaration:

v: T

indicates that the object referred to by v is to be treated abstractly, while a declaration:

w: rep

indicates that the object referred to by w is to be treated concretely. Two primitives, up and down, are available for converting between these two points of view. The use of up permits a type rep object to be viewed abstractly, while down permits an abstract object to be viewed concretely. For example, given the declarations above, the following two assignments are legal:

v := up(w) w := down(v)

Only routines inside a cluster may use up and down. Note that up and down are used merely to inform the compiler that the object is going to be viewed abstractly or concretely, respectively.

A common place where the view of an object changes is at the interface to one of the type's operations: the user, of course, views the object abstractly, while inside the operation, the object is viewed concretely. To facilitate this usage, a special type specification, evt, is provided. The use of evt is restricted to the *args*, *returns*, *yields* and *signals* clauses of routines inside a cluster, and may be used at the top level only (e.g., *array[evt]* is illegal). When used inside the *args* clause, it means that the view of the argument object changes from abstract to concrete when it is assigned to the formal argument variable. When evt is used in the *returns*, *yields*, or *signals* clause, it means the view of the result object

changes from concrete to abstract as it is returned (or yielded) to the caller. Thus cvt means abstract outside, concrete inside: when constructing the type of a routine, cvt is equivalent to the abstract type, but when type-checking the body of a routine, cvt is equivalent to the representation type. The type of each routine is derived from its heading in the usual manner, except that each occurrence of cvt is replaced by the abstract type. The cvt form does not introduce any new ability over what is provided by up and down. It is merely a shorthand for a common case.

Inside the cluster, it is not necessary to use the compound form (type_spec\$op_name) for naming locally defined routines. Furthermore, the compound form cannot be used for calling hidden routines.

12.4. Equate Modules

An equate module provides a convenient way to define a a set of equates for later use by other modules.

The form of an equate module is:

```
idn = equates [ parms [ where ] ]
equate { equate }
end idn
```

The usual scope rules apply. The *idn* following the end of the equate module must be the same as the *idn* naming the equate module.

In this section we discuss non-parameterized equate modules. Parameterized modules are discussed in Section 12.5.

An equate module defines a set of equates, that is, it defines a set of named constants. The set of equates is also a constant, although it is not an object. Thus the name of an equate module can be used in an equate, but an equate module cannot be assigned to a variable. The equates defined by an equate module *E* may be referenced using the same syntax as for naming the operations of a cluster. For example, an object or type named *n* in equate module *E* can be referred to as *E*\$*n*. If equate modules contain equates that give names to other equate modules, compound names can be used. For example:

A[Int]\$B\$C\$name

where A, B, and C are equate modules is legal.

As always, equates to type specifications do not define new types but merely abbreviations for types. For example, in the following:

```
my_types = equates
ai = array[Int]
float = real
end my_types
```

the types my_types\$ai and array[int] are equivalent.

Modules

12.5. Parameterized Modules

Procedures, iterators, clusters, guardians (see Section 13), and equate modules may all be parameterized. Parameterization permits a set of related abstractions to be defined by a single module. In each module heading there is an optional *parms* clause and an optional *where* clause (see Appendix I). The presence of the *parms* clause indicates that the module is parameterized; the *where* clause declares the types of any operation parameters that are expected to accompany the formal type parameters.

The form of the parms clause is:

[parm ,]

where

parm ::= idn , ... : type_spec

idn , : type

Each parm declares some number of formal parameters. Only the following types of parameters can be declared in a *parms* clause: **int, real, bool, char, string, null, and type.** The declaration of operation parameters associated with type parameters is done in the *where* clause, as discussed below. The actual values for parameters are required to be constants that can be computed at compile-time. This requirement ensures that all types are known at compile-time, and permits complete compile-time type-checking.

In a parameterized module, the scope rules permit the parameters to be used throughout the module. Type parameters can be used freely as type specifications, and all other parameters (including the operations parameters specified in the *where* clause) can be used freely as expressions.

A parameterized module implements a set of related abstractions. A program must instantiate a parameterized module before it can be used; that is, it must provide actual, constant values for the parameters (see Section 12.6). The result of an instantiation is a procedure, iterator, type, guardian, or equate module that may be used just like a non-parameterized module of the same kind. Each distinct list of actual parameters produces a distinct procedure, iterator, type, guardian, or equate module (see Section 12.6 for details).

The meaning of a parameterized module is given by binding the actual parameters to the formal parameter names and deleting the *parms* clause and the *where* clause. That is, in an an instantiation of a parameterized module, each formal parameter name denotes the corresponding actual parameter. The resulting module is a regular (non-parameterized) module. In the case of a cluster some of the operations may have additional parameters; further bindings take place when these operations are instantiated.

In the case of a type parameter, one can also declare what operation parameters must accompany the type by using a *where* clause. The *where* clause also specifies the type of each required operation parameter. The *where* clause constrains the parameterized module as well: the only operations of the type parameter that can be used are those listed in the *where* clause.

80

The form of the where clause is:

There are two forms of restrictions. In both forms, the initial idn must be a type parameter. The has form lists the set of required operation parameters directly, by means of oper_decls. The type_spec in each oper_decl must be a proctype, itertype, or creatortype (see Appendix I). The in form requires that the actual type be a member of a type_set, a set of types with the required operations. The two identifiers in the type_set must match, and the notation is read like set notation; for example,

{t | t has f: ... }

means "the set of all types t such that t has f ... ". The scope of the identifier is the type_set.

The in form is useful because an abbreviation can be given for a type_set via an equate. If it is helpful to introduce some abbreviations in defining the type_set, these are given in the optional equates within the type_set. The scope of these equates is the entire type_set.

A routine in a parameterized cluster may have a *where* clause in its heading, and can place further constraints on the cluster parameters. For example, any type is permissible for the array element type, but the array *similar* operation requires that the element type have a *similar* operation. This means that **array**[7] exists for any type 7, but that **array**[7]\$*similar* exists only when an actual operation parameter is provided for *T*\$*similar* (see Section 12.6). Note that a routine need not include in its where clause any of the restrictions included in the cluster where clause.

12.6. Instantiations

To instantiate a parameterized module, constants or type specifications are provided as actual parameters:

actual_parm ::= constant type_actual

type_actual ::= type_spec [with { opbinding , ... }]

```
opbinding ::= name , ... : primary
```

If the parameter is a type, the module's *where* clause may require that some routines be passed as parameters. These routines can be passed implicitly by omitting the *with* clause; the routine selected as a default will be the operation of the type that has the same name as that used in the *where* clause.

Routines may also be passed explicitly by using the *with* clause, overriding the default. In this case, the actual routine parameter need not have the same name as is required in the *where* clause, and need not even be one of the type's primitive operations.

The syntactic sugar that allows default routines to be selected implicitly works as follows. If a generator requires an operation named op from a type parameter, and if the corresponding type_actual, TS with { ... }, has no explicit binding for op, then Argus adds an opbinding of op to TS\$op. (It will be an error if TS\$op is not defined.) Thus one only has to provide an explicit opbinding if the default is unsatisfactory.

For example, suppose a procedure generator named sort has the following heading:

sort = proc[t: type](a: array[t]) where t has gt: proctype(t,t) returne(bool)

and consider the three instantiations:

sont[int with {gt: int\$gt}]
sont[int]
sont[int with {it: int\$it}]

The first two instantiations are equivalent; in the first the routine **int\$***gt* is passed explicitly, while in the second it is passed implicitly as the default. In the third instantiation, however, **int\$***it* is passed in place of the default. All three instantiations result in a routine of type:

proctype (array[int])

and so each could be called by passing it an array[Int] as an argument. However a call of the third instantiation will sort its array argument in the opposite order from a call of either the first or second instantiation.

Within an instantiation of a parameterized module, an operation of a type parameter named \$\$op denotes the actual routine parameter bound to op in the instantiation of that module. For example, suppose we make the call:

```
sort[Int with {gt: int$it}] (my_ints)
```

where my_ints is an array of integers. If, in the body of sort, there is a recursive call:

```
sort[t with {gt: t$gt} ] (a, i, j)
```

then t denotes the type int, and isgt denotes the routine intsit, so that the recursive sort happens in the correct order.

A cluster generator may include routines with *where* clauses that place additional requirements on the cluster's type parameters. A common example is to require a *copy* operation only within the cluster's *copy* implementation.

```
set = cluster[t: type] is ..., copy
where t has equal: proctype(t,t) returns(bool)
rep = array[t]
...
copy = proc(s: cvt) returns(cvt) where t has copy: proctype(t) returns(t)
return(rep$copy(s))
end copy
```

The intent of these subordinate where clauses is to allow more operations to be defined if the actual type parameter has the additional required operations, but not to make the additional operations an absolute

82

12.6 instantiations

requirement for obtaining an instance of the type generator. For example, with the above definition of *set*, *set*[any] would be defined, but *set*[any]\$copy would not be defined because any does not have a *copy* operation. We shall call the routine parameters required by subordinate where clauses optional parameters.

Like regular required parameters, optional parameters can be provided when the cluster as a whole is instantiated and can be provided explicitly or by default. For any optional parameter op that is not provided explicitly by the *type_actual*, *TS* with { ... }, we add an opbinding of op to *TS*\$op if *TS*\$op exists; otherwise the opbinding is not added. The resulting cluster contains just those operations for which opbindings exist for all the required routine parameters. For example, as mentioned above, set[any] would not have a *copy* operation because any\$copy does not exist and therefore the needed opbinding is not present. On the other hand, set[int] does have a *copy* operation because int\$copy does exist. Finally, *set[any with {copy: foo}]*, where foo is a procedure that takes an any as an argument and returns an any as a result, would have a *copy* operation.

For an instantiation to be legal it must type check. Type checking is done after the syntactic sugars are applied. The types of constant parameters must be included in the declared type, type actuals must be types, and the types of the actual routine parameters must be included in the proctypes, itertypes, or creatortypes declared in the appropriate where clauses. Of course, the number of parameters declared must match the number of actuals passed and with each type actual parameter there must be an opbinding for each required routine parameter. If the generator is a cluster, then opbindings must be provided for all operations required in the cluster's where clause; opbindings can (but need not) be provided for optional parameters. Extra actual routine parameters are illegal.

Because the meaning of an instantiation may depend on the actual routine parameters, type equality makes instances with different actual routine parameters distinct types. For example, consider the set type generator again; the instance

set[array[int] with {equal: array[int]\$equal}]

is not equal to

set[array[Int] with {equal: array[Int]\$similar}]

Intuitively these instances should be unequal because the two equal procedures define different equivalence classes and therefore the abstract behaviors of the two instances are different. However, optional parameters do not effect type equality. For example,

set[array[int] with {copy: int\$copy}]

and

set[array[int] with {copy: my copy}]

are equal types. This is intuitively justified because in each case set objects behave the same way even though different sets are produced when sets are copied in the two cases.

Thus we have the following type equality rule, which defines when two type_specs denote equal types (after syntactic sugars are applied). A similar notion is also needed for routine equality. A formal type

Modules

identifier is equal only to itself for type checking purposes. Otherwise, two type names denote equal types if they denote the same Description Unit (DU).¹¹ Similarly, Argus compares the names of routine formals or the DUs of routines, or checks that they are the same operation in equal types. To decide the equality of two type generator instantiations:

 $T[t_1 \text{ with } \{op_1: act_1, ... op_m: act_m\}, ..., t_n \text{ with } \{...\}\}$ and

 $T'[t_1' \text{ with } \{op_1: act_1', ... op_m: act_m'\}, ..., t_n' \text{ with } \{...\}]$

Argus first checks whether:

1. T and T denote the same DU, and whether

2. they have the same number of type_actuals, and t_1 is equal to t_1 , etc.

Second, any optional parameter opbindings in either instantiation are deleted. After this step, Argus checks that for each corresponding type_actual there is the same number of opbindings and that each corresponding opbinding is the same. (That is, the corresponding actual routines are equal.) The order of the actual routine parameters does not matter, since Argus matches opbindings by operation names. (The definition of routine equality for instantiations of routine generators is similar.) This definition, for example, tells us that

set[array[int] with {equal: array[int]\$equal}]

is different from

set[array[int] with {equal: array[int]\$similar}] ,

(assuming set requires an equal operation from its type parameter). It also tells us that:

```
set[ int with {equal: foo, copy: bar} ]
```

and

```
set[ int with {equal: foo, copy: xerox} ]
```

are equal (assuming copy is required only by the set[Int]\$copy operation).

This type equality rule allows programmers to control what requirements affect type equality by choosing whether to put them on a cluster or on each operation. A requirement on the cluster should be used whenever the actuals make some difference in the abstraction. For example, in the set cluster, the type parameter's equal operation should be required by the cluster as a whole, since using different equality tests for a set's objects causes the set's behavior to change.

One can require that a type parameter, say t, be transmissible by stating the requirement:

t has transmit

This requirement is regarded as a formal parameter declaration for a special "transmit actual", but Argus does not provide syntax for passing it explicitly. The "transmit actual" is passed implicitly just when the actual type parameter is transmissible and the generator requires it.

¹¹This is name equality unless the type environment has synonyms for types.

12.7. Own Variables

Occasionally it is desirable to have a module that retains information internally between calls. Without such an ability, the information would either have to be reconstructed at every call, which can be expensive (and may even be impossible if the information depends on previous calls), or the information would have to be passed in through arguments, which is undesirable because the information is then subject to uncontrolled modification in other modules (but see also the binding mechanism described in Section 9.8).

Procedures, iterators, handlers, creators, and clusters may all retain information through the use of own variables. An own variable is similar to a normal variable, except that it exists for the life of the program or guardian, rather than being bound to the life of any particular routine activation. Syntactically, own variable declarations must appear immediately after the equates in a routine or cluster body; they cannot appear in bodies nested within statements. Declarations of own variables have the form:

own var ::= own decl

own idn : type_spec := expression own decl , ... := call [@ primary]

Note that initialization is optional.

The own variables of a module are created when a guardian begins execution or recovers from a crash, and they always start out uninitialized. The own variables of a routine (including cluster operations) are initialized in textual order as part of the first call of an operation of that routine (or the first such call after a crash), before any statements in the body of the routine are executed. Cluster own variables are initialized in textual order as part of the first call of the first cluster operation to be called (even if the operation does not use the own variables). Cluster own variables are initialized before any operation own variables are initialized. Argus insures that only one process can execute a cluster's or a routine's own variable initializations.

Aside from the placement of their declarations, the time of their initialization, and their lifetime, own variables act just like normal variables and can be used in all the same places. As with normal variables, an attempt to use an uninitialized own variable (if not detected at complic-time) will cause the guardian to crash.

Declarations of own variables in different modules always refer to distinct own variables, and distinct guardians never share own variables. Furthermore, own variable declarations within a parameterized module produce distinct own variables for each distinct instantiation of the module. For a given instantiation of a parameterized cluster, all instantiations of the type's operations share the same set of cluster own variables, but distinct instantiations of parameterized operations have distinct routine own variables.

Declarations of own variables cannot be enclosed by an except statement, so care must be exercised when writing initialization expressions. If an exception is raised by an initialization expression, it will be

treated as an exception raised, but not handled, in the body of the routine whose call caused the initialization to be attempted. Thus, the guardian will crash due to this error.

13. Guardians

This section is concerned with the form and meaning of the machine used to define guardiens. Such a module, called a guardient definition, declares the adjuste mating as the guardients static state and volatile state, and provides implementatives for the guardiants building. It also defines one or more creators: operations that provides may gravitly building the building of the guardient with the guardient definition, in addition, a guardien definition may provide building the guardient building as the guardient architect, and recovery code to restore the volatile state when the guardient to state when the guardient building of the guardient of

The syntactic form of a guardian datinition is as follows:

ich = guardian [parms] to ide , [handles ide ,] [where]

```
{ equale }
{ state_deci }
[ reserver backy and ]
[ background backy and ]
{ operation } creator { execution }
and ich
```

where

operation IIII creator handler

The initial ide names the guardian type or type generator (as applicant in Section 4.4) and must agree with the final ide. The guardian boular deviates two ide into. You does the device the sectors of the creators, which can be called to create and initialize two ide into guardiant (the device and initialize two ide into the device the dev

The remaining portions of the guardian definition are discussed in the submatilens below.

13.1. The Guardian State

The state decision the guardian dativities declare a number of variables (with optional initialization):

```
state_deci :: 24 [ atustate ] deal
```

annha in the speet is expression

I and the I doubly not to said

The scope of these destandance is the entry guardian definition. The signals reachable from variables declared to be stable survive gradues of the guardian, while share definitions do not.

For example, If the state deals were:

stable buffer: atomic_array[int] := atomic_array[int]\$new() cache: array[int] := array[int]\$new()

then the atomic_array object denoted by *buffer* would survive a guardian crash, but the array object denoted by *cache* would not. See Section 13.3 for more details of crash recovery. Volatile variables can be assigned wherever an assignment statement is legal. However, stable variables may only be assigned by an initialization when declared or in the body of a creator. The initializations of both stable and volatile variables are executed within an action, as described below. However, the stable variables are not reinitialized upon crash recovery, whereas volatile variables are reinitialized upon crash recovery.

Stable variables should denote resilient objects (see Section 15.2), because only resilient data objects (reachable from the stable variables) are written to stable storage when a topaction commits. (This can be ensured by having stable variables only denote objects of an atomic type or objects protected by mutex.) Non-resilient objects stored in stable variables are only written to stable storage once, when the guardian is created. Furthermore, the stable variables should usually denote atomic objects, because the stable variables are potentially shared by all the actions in a guardian.

13.2. Creators

A guardian definition must provide one or more creators. The names of these creators must be listed in the guardian header (internal creators are not allowed); each such name must correspond to a single creator definition appearing in the body of the guardian definition.

A creator definition has the same form as a procedure definition, except that creators cannot be parameterized, and the reserved word creator is used in place of proc:

```
idn = creator ([ args ]) [ returns ] [ signals ]
routine_body
and idn
```

The initial *idn* names the creator and must agree with the final *idn*. The types of all arguments and all results (normal and exceptional) must be transmissible.

A creator is an object of some creator type. This type is derived from the creator heading by removing the creator name, rewriting the formal argument declarations with one *idn* per *decl*, deleting the idns of all formal arguments, deleting any *failure* or *unavailable* signals, and finally, replacing creator by creatortype. The signals *failure*(string) and *unavailable*(string) are implicit in every creator type (since they can arise from any creator call). However, if these signals are raised explicitly by a creator, they must be listed in the *signals* clause with string result types.

The semantics of a creator call are explained in Section 8.4. Typically, the body of a creator will initialize some stable and volatile variables. It can also return the name of the guardian being created using the expression self. Since the creator (and the state initialization) runs as an action, the creator terminates by committing or aborting. If it aborts, the guardian is destroyed. If it commits, the guardian begins to accept handler calls, and runs the background code, if any (see below). If an ancestor of the creator aborts, the guardian is destroyed. If the creator and all its ancestors commit, the guardian becomes permanent, and will survive subsequent crashes.

13.2 Creators

13.3. Crash Recovery

Once a guardian becomes permanent, it will be recreated automatically after a crash with its stable variables initialized to the same state they were in at the last topaction commit before the crash. The volatile variables are then initialized (in declaration order) by a topaction. To aid in this reinitialization, the guardian definition can provide a *recover section*:

recover body end

to be run, as part of this topaction, after the initializations attached to the volatile variable declarations are performed. The recover section commits when control reaches the end of the body, or when a return statement is executed. The recover section may abort by executing an abort return statement or as a result of an unhandled exception. The guardian crashes if the recover section aborts.

13.4. Background Tasks

Tasks that must be performed periodically, independent of handler calls, can be defined by a *background section*:

background body end

The system creates a process to run this body as soon as creation or recovery commits successfully. The body of the background section does *not* run as an action; typically it will perform a sequence of topactions.

If the background process finishes executing its *body* (either by reaching the end of the block or by returning), the process terminates, but the guardian continues to execute incoming handler calls.

13.5. Handlers and Other Routines

Typically, the principal purpose of a guardian is to execute incoming handler calls. A guardian accepts handler calls as soon as creation or recovery commits.

The guardian header lists the names of the externally available handlers. Each handler listed must be defined by a handler definition. Additional handler definitions may also be given, but these handlers can be named only within the guardian to which they belong.

A handler definition has the same form as a procedure definition, except that handlers cannot be parameterized, and the reserved word handler is used in place of proc:

```
idn = handler ([ args ]) [ returns ] [ signals ]
routine_body
end idn
```

The initial *idn* names the handler and must agree with the final *idn*. The types of all arguments and all results (normal and exceptional) must be transmissible.

A handler is an object of some handler type. This type is derived from the handler heading by removing the handler name, rewriting the formal argument declarations with one *idn* per *decl*, deleting the

iches of all formal anguments, detailing any follow or scandilities stands, and finally, replacing franklin' by bandlertype. The algorith following and another bound and the following and the following franklin' by However, if these algorith are related anything by a bandle for scandar the standard of the algorith should with atring as their result type.

As explained in Section 8.3, a handler cell note as a submittee of the only. and expenses and results are passed by value. A new property is the first first

A guardian definition may also cartain passations and families definitions. These procedures and herefore may be called only without the guardian families in the families.

13.6. Guardian Lifetime and Destruction

A generalizer dasse entre source and an entre of the source of the sourc

Create a guardian becomes parameteri, il vill surdire cost product bits and but and the may bet the series of the series ments during contain the series of the series of

A short-lived guardian can be implemented by using basic provident where them:

The backgrout disease plants for any descence on the scarpe plants of the scarpe back of

The following is an example of a hundler for destructing a province grantite finish = hundler (...) tokense (...) alignetic (not subletime):

```
terneinete
retern(...)
end finist
```

Here, finish might charts whether its online is authorized in such this second, and signal and authorized if not. Otherwise it wanted is a statistical state interval and an and a statistical state interval and a state i

13.7. An Example

To Restrict how must of the components of a guardian deficition are used, as example of a simple guardian is given in Figure 13-1. An active care use a guardian guardian deficition deficition with the action has committed to the top. The specter than general field of the stars guardians for

13.7 An Example

consumption. The spooler provides an operation for adding (object, consumer) pairs, and for destroying the quardian.

结合的 医内静脉 化分子 化化合金 网络美国大学教师

Figure 13-1: Spooler Guardian

```
spooler = guardian [t: type] is create handles eng, finish
            where t has transmit
utype = handiertype (t)
entry = struct(object: t, consumer: utype)
queue = semiqueue[entry]
stable state: queue := queue$create()
background
   while true do
       enter topaction
           e: entry := queue$deq(state)
           e.consumer(e.object)
             except when unavailable (*): abort leave end
           ond except when failure, unavailable (*): ond
       end
   end
create = creator () returns (spooler[t])
   return(self)
   end create
eng = handler (item: t, user: utype)
   queue$enq(state, entry${object: item, consumer: user})
   end enq
finish = handler ()
   terminate
   end finish
end spooler
```

The spooler guardian is parameterized by the type of object to be stored. The *enq* handler takes an object of this type, and a handler for sending the object to the consumer, and adds this information to the stable state of the spooler. This state is an object of the *semiqueue* abstract data type¹². Each entry in the semiqueue is a structure containing a stored object and its corresponding consumer handler. The background code of the guardian runs an infinite loop that starts a topaction, removes an entry from the queue, and sends the object using the associated handler.

Note that an *unavailable* exception arising from this handler call is caught inside the topaction, so that an explicit abort can be performed. If the exception were caught outside the topaction, it would cause the

¹²See W. Weihl and B. Liskov, "Implementation of Resilient, Atomic Data Types", in ACM Transactions on Programming Languages and Systems, volume 7, number 2, (April 1985), pages 244-269.

Guardians

and the present of th

topaction to commit, and the entry would be removed without being consumed. Note also that *failure* is caught outside the topaction, since if an *encode* were to fail, or if the guardian did not exist, the background process might aimlessly loop forever, because it would not be able to remove that entry.

and the second second

A more extended example of a distributed system appears in the paper Liskov, B. and Scheifler, R., "Guardians and Actions: Linguistic Support for Robust, Distributed Programs," ACM Transactions on Programming Languages and Systems, volume 5, number 3, (July 1983), pages 381-404.

14. Transmissibility

A type is said to be surpresentable if it defines a transmit appreation that allows the values of its objects to be sent in measures or stared in image objects. Cate about all measurementable type may be used as arguments to handler calls or creater cate. This another departure inter-immendeaton is defined for the Argus built-in types and for user-defined types.

14.1. The Transmit Operation

Transmissibility is a property of a data abstraction and must be dated in the specification of that abstraction. A transmissible data type 7 and to the throught a building an additional operation,

transmit - prestype (1) address (1) adjusted (1)

which is called implicitly during minimum terministics. Clears an engine, monomit produces a different object, which may even reache at a different growthing terministics and approximation for the address to different by the second set for the second to be second t

14.2. Transmission for Bullt-in Types

The unstructured built-in-types girl, along, Seat. ...) are encoded by the encodion of procepto, Hertype, and any. The toppont operations of the surface based process webs assetty, with the exception of the real type, which, because of preside the surface of anyon, granumbus only that the two values diller by very little.

The structured types (instances of entry, etnest, etenitic section, ...) are transmissible if and only if all their type parameters are transmissible. The transmitting elements (generations of the entryment types. For elements (generation, r is an entry established elements of type T, then the transmitting elements of the entryment types. All elements along y with the same bounds as the original, and with elements:

yl - Themaniette

Thus transmission of the built-in almostered types will preserve value equality only if transmission of the component types does.

The transmit specific for material acquires and balls the last during the transmission (actually, during the encoding, see below) of the contained object.

19Lishov, B. et al., CLU Reference Manual, Lastere Notes in Computer Subscie, valiance 114, (Spitzger-Verleg, New York, 1981).

Transmissibility

14.3. Transmit for Abstract Types

The type implemented by a cluster is transmissible if the reserved word transmit appears in the la-list at the head of the cluster. Unlike the other operations provided by a type, the transmit operation cannot be called directly by users, and in fact is not implemented directly in the cluster. Instead, transmit is implemented indirectly in the following way. Each transmissible type is given a canonical representation, called its *external representation type*. The external representation type of an abstract type T is any convenient transmissible type XT. This type can be another abstract type if desired; there is no requirement that XT be a built-in type. Intuitively, the meaning of the external representation is that values of type XT will be used in messages to represent values of type T. The choice of external representation type is made for the abstract type as a whole and must be used in every implementation of that type. (There are currently no provisions for changing the external representation of a type once it has been established in the library.)

Each implementation of the abstract type 7 must provide two operations to map between values of the abstract type and values of the external representation type. There is an operation

encode = proc (a: T) returns (XT) [signals (failure(string))]

to map from T values to XT values (for sending messages) and an operation

decode = proc (x: XT) returns (T) [signals (failure(string))]

to map from XT values to T values (for receiving messages). The transmit operation for T is defined by the following identity:

T\$transmit (x) = T\$decode (XT\$transmit (T\$encode(x)))

Intuitively, the correctness requirement for *encode* and *decode* is that they preserve the abstract T values: encode maps a value of type T into the XT value that represents it, while *decode* performs the reverse mapping¹⁴.

Encode and decode are called implicitly by the Argus system during handler and creator calls. If encode and decode do not appear in the cluster's is-list, then they will be accessible to the Argus system, but may not be named directly by users of the type. A *failure* exception raised by one of these operations will be caught by the Argus system and resignalled to the caller (see Section 8.3).

An abstract type's encode and decode operations should not cause any side effects. This is because the number of calls to encode or decode is unpredictable, since arguments or results may be encoded and decoded several times as the system tries to establish communication. In addition, verifying the correctness of transmission is easier if encode and decode are simply transformations to and from the external representation.

When defining a parameterized module (see Section 12.5), it may be necessary to require a type parameter to be transmissible. A special type restriction:

¹⁴Herlihy, M. and Liskov, B., "A Value Transmission Method for Abstract Data Types", ACM Transactions on Programming Languages and Systems, volume 4, number 4, (Oct. 1962), pages 527-551.

14.3 Transmit for Abstract Types

has transmit

is provided for this purpose. To permit instantiation only with transmissible type parameters, this restriction should appear in the where clause of the cluster. Alternatively, by placing identical where clauses in the headings of *encode* and *decode* procedures, one can ensure that an instantiation of the cluster is transmissible only if the type parameters are transmissible (see Section 12.5).

As an example, Figure 14-1 shows part of a cluster defining a key-item table that stores pairs of values, where one value (the key) is used to retrieve the other (the *item*). The key-item table type has operations for creating empty tables, inserting pairs, retrieving the item paired with a given key, deleting pairs, and iterating through all key-item pairs. The table is represented by a sorted binary tree, and its external representation is an array of key-item pairs. The table type is transmissible only if both type parameters are transmissible.

Figure 14-1: Partial implementation of table.

```
table = cluster [key, item: type] is create, insert, lookup, alipairs, delete, transmit, ....
     where key has it: proctype (key, key) returns (bool),
                     equal: proctype (key, key) returns (bool)
     pair = record[k: key, i: item]
     nod = record[k: key, i: item, left, right: table[key, item]]
     rep = variant[empty: null, some: nod]
    xrep = array[pair]
                           % the external representation type
     % The internal representation is a sorted binary tree. All pairs in the table
     % to the left (right) of a node have keys less than (greater than) the key in
    % that node.
    % ... other operations omitted
    encode = proc (t: table[key, item]) returns (xrep)
                     where key has transmit, item has transmit
            xr: xrep := xrep$new() % create an empty array
            % use allpairs to extract the pairs from the tree
            for p: pair in alipairs(t) do
                % Add the pair to the high end of the array.
                xrep$addh(xr, p)
                end
            return(xr)
            end encode
    decode = proc (xtbl: xrep) returns (table[key, item])
                where key has transmit, item has transmit
            t: table[key, item] := create() % create emoty table
            for p: pair in xrep$elements(xr) do
                % xrep$elements yields all elements of array xr
                insert(t, p.key, p.item) % enter pair in table
                end
            return(t)
            end decode
     end table
```

14.4. Sharing

When an object of structured built-in type is encoded and decoded, sharing among the object's components is preserved. For example, let a be an **array**[7] object such that a[i] and a[j] refer to a single object of type T. If a^2 is an **array**[7] object created by transmitting a, then $a^2[i]$ and a2[i] also name a single object of type T.

and march sugar W

All sharing is preserved among all components of multiple objects of built-in type when those objects are encoded together. Thus, sharing is preserved for objects that are arguments of the same remote call or are results of the same remote call, unless the arguments are encoded at different times (see the discussion of the bind expression in Section 9.8). For example, let *a* and *b* be array[7] objects such that a[i] and b[j] refer to a single object of type 7. If *a*2 and *b*2 are arrays created by sending *a* and *b* as arguments in a single handler call, then a2[i] and b2[j] also refer to a single object.

Whether an abstract type's transmit operation preserves sharing is part of that type's specification, but sharing should usually be preserved for abstract types. In the key-item table implementation of Figure 14-1, there are two types of sharing that should be preserved: sharing of keys and items among multiple tables sent in a single message, and sharing of items bound to the same key in a single table. The key-item table example shows how to implement an abstract type whose transmission preserves sharing by choosing an external representation type whose transmit operation preserves sharing.

Care must be taken when the references among objects to be transmitted are cyclic, as in a circular list. Decoding such objects can result in a *failure* exception unless *encode* and *decode* are implemented in one of two ways:

1. the internal and external representation types are identical and encode and decode return their argument object without modifying it or accessing its components, or

2. the external representation object must be free of cycles.

15. Atomic Types

In Argus, atomicity is enforced by the objects shared among actions, rather than by the individual actions themselves. Types whose objects ensure atomicity of the actions sharing them are called *atomic types*; objects of atomic types are called *atomic objects*. In this chapter we define what it means for a type to be atomic and describe the mechanisms provided by Argus to support the implementation of atomic types.

Atomicity consists of two properties: serializability and recoverability. An atomic type's objects must synchronize actions to ensure that the actions are serializable. An atomic type's objects must also recover from actions that abort to ensure that actions appear to execute either completely or not at all.

In addition, an atomic type must be *resilient*: the type must be implemented so that its objects can be saved on stable storage. This ensures that the effects of an action that commits to the top (that is, an action that commits, as do all of its ancestors) will survive crashes.

This chapter provides definitions of the mechanisms used for user-defined types in Argus. For example implementations, see Weihl, W. and Liskov, B., "Implementation of Resilient, Atomic Data Types," ACM Transactions on Programming Languages and Systems, volume 7, number 2 (April 1985), pages 244-269.

The remainder of this chapter is organized as follows. In Section 15.1 and Section 15.2, we present the details of the mechanisms. Section 15.1 focuses on synchronization and recovery of actions, while Section 15.2 deals primarily with resilience. In Section 15.3, we discuss some guidelines to keep in mind when using the mechanisms described in Section 15.1 and Section 15.2. In Sections 15.4 and 15.5, we define more precisely what it means for a type to be atomic. Finally, in 15.6, we discuss some details that are important for user-defined atomic types that are implemented using multiple mutexes.

15.1. Action Synchronization and Recovery

In this section we describe the mechanisms provided by Argus to support synchronization and recovery of actions. These mechanisms are designed specifically to support implementations of atomic types that allow highly concurrent access to objects.

Like a non-atomic type, an atomic type is implemented by a cluster that defines a representation for the objects of the type, and an implementation for each operation of the type in terms of that representation. However, the implementation of an atomic type must solve some problems that do not occur for ordinary types, namely: synchronizing concurrent actions, making visible to other actions the effects of committed actions, hiding the effects of aborted actions, and providing resilience against crashes.

An implementation of a user-defined atomic type must be able to find out about the commits and aborts of actions. In Argus, implementations use objects of built-in atomic types for this purpose. The representation of a user-defined atomic type is typically a combination of atomic and non-atomic objects;

the non-atomic objects are used to bail information that can be assumed by consument actions, while the atomic objects can be used to anonor the televing quanties. All this water flat, cannot a particular change to the representiation:

· commit (as the new information is now available to allow authority,

- · abort (so the change should be impation), or
- · is it still active (ap the information convex in related of ?

The operations available on build in stands adjusts been block to adjust in apport the type of use; in particular, the care respirant data and an available and the deviation of the product interval and the test of the standard of the deviation of the product of the deviation of the standard of the standard of the standard of the deviation of the product of the deviation of the standard of the st

The use of atomic objects in the representative parallelistic department of the department of the parallelistic department of the representative constraints defined atomic depart of the response of the department of the response defined atomic department of the department of the response defined atomic department of the department of

Synchronization for two-densis data is provided by Section 2000 sectors and the Andrewski in Section C.7, a material field of the two-density is control or two density of sectors and the Andrewski in Section manual control of a space science of the two-density of two density of two density of two density of the two density of two gains provide the Constant science. The two density of two densi

15.2. Realiance

1.0

If a unservice and nie aldes is methodale inter dat si ne guandian, it should be written to stable size dingen die staar neek damas hav unsel dalle durge propuly. Call an la dijects.

In addition to its upp for anothershing wat generalized in the second for the second s

To minimize the amount of trianmation that must be without to static approximation below automs convert, the Argue system only captes next internation to states, show a state of the second states, it captes accounties objects must be must reach accurately accurately approximately approximately accurate the second state the top. For maters objects, it does anytee monty accurately about the states accurately accurately accurately accurately approximately approximately accurately accurately
15.2 Resilience

changed = proc (m: mutex[T])

is provided for notifying the system that an existing mutex object should be written to stable storage. Calling this operation will cause the object to be written to stable storage (assuming it is accessible) by the time the action that executed the *changed* operation commits to the top. Sometime after the action calls *changed*, and before its top-level ancestor commits, the system will copy the mutex object to stable storage. *Changed* must be called from a process running an action.

Mutex objects also define how much information must be written to stable storage. Copying a mutex object involves copying the contained object. By choosing the proper granularity of mutex objects the user can control how much data must be written to stable storage at a time. For example, a large data base can be broken into partitions that are written to stable storage independently by dividing it among several mutex objects. Such a division can be used to limit the amount of data written to stable storage by calling *changed* only for those partitions actually modified by a committing action.

In copying a mutex object, the system will copy all objects reachable from it, excluding other mutex or built-in atomic objects. A contained mutex or built-in atomic object will be copied only if necessary; that is, only if it is:

- a mutex object for which (a descendant of) the completing action called the changed operation,
- a built-in atomic object that was modified by the action, or
- a newly accessible object for which no stable copy exists.

Furthermore, the component is copied independently of the containing mutex object; they may be copied in either order (or simultaneously), subject to the constraint that the system cannot copy a mutex object without first gaining possession of it.

Finally, mutex objects can be used to ensure that information is in a consistent state when it is written to stable storage. The system will gain possession of a mutex object before writing it to stable storage. By making all modifications to mutex objects inside **selze** statements, the user's code can prevent the system from copying a mutex object when it is in an inconsistent state.

Some details of the effect of *changed* are important for atomic types that are implemented as multiple mutexes. These details are presented in Section 15.6.

15.3. Guidelines

This section discusses some guidelines to be followed when implementing atomic types. There are additional guidelines to follow when multiple mutexes are used to implement an atomic type; those guidelines are discussed in Section 15.6.

An important concept for describing the resilience of user-defined atomic types is synchrony. An object is *synchronous* if it is not possible to observe that any portion of the object is copied to stable storage at a different time from any other portion. For example, an object of type array[mutex[Int]] would not be

synchronous, because elements of the array can be copied at different times. A type is synchronous if all of its objects are synchronous. Whether a type is synchronous or not is an important property of its behavior and should be stated in its specification. The built-in atomic types are synchronous; user-defined types must also be synchronous if they are to be atomic.

To ensure the resilience and serializability of a user-defined atomic type independently of how it is used, the form of the rep for an atomic type should be one of the following possibilities.

- 1. The rep is itself atomic. Note that mutex is not an atomic type.
- 2. The rep is mutex[f] where t is a synchronous type. For example, t could be atomic, or it could be the representation of an atomic type, if the operations on the this fictilious atomic type are coded in-line so that the entire type behaves atomically.
- 3. The rep is an atomic collection of mutex types containing synchronous types.
- 4. The rep is a mutable collection of synchronous types, and objects of the representation type are never modified after they are initialized. That is, mutation may be used to create the initial state of such an object, but once this has been done the object must never be modified.

When using mutex objects, there are a few rules to remember. First, *changed* must be called after the last modification (on behalf of some action) to the contained object. This is true because the Argus system is free to copy the mutex to stable storage as soon as *changed* has been called.

In addition, *changed* should be called even if the object is not accessible from the stable variables of a guardian. In part this rule is just an example of separation of concerns: the implementation of the atomic type should be done independently of any assumptions about how the object will be used. Therefore the type should be implemented as if its objects were accessible from the stable variables of some guardian. However, in addition, if this rule is not followed, it is possible that stable storage will not be updated properly. This situation can occur if an object was accessible, then becomes inaccessible, and later becomes accessible again. The system guarantees that no problems arise if *changed* is always called after the last modification to the object.

Mutex objects should not share data with one another, unless the shared data is atomic or mutex. One reason for this rule is that in copying mutex objects to stable storage Argus does not preserve this kind of sharing.

A final point about mutex objects is that it is unwise to do any activity that is likely to take a long time inside a seize statement. For example, a handler call should not be done from inside a seize statement if possible. Also, it is unwise to wait for a lock inside a seize unless the programmer can be certain that the lock is available or will be soon. Otherwise, a deadlock may occur. An example of where waiting for a lock in a nested seize statement is safe is where all processes seize the two mutex objects in the same order.

15.4. A Prescription for Atomicity

In this section, we discuss how to decide how much concurrency is possible in implementing an atomic type. In writing specifications for atomic types, we have found it helpful to pin down the behavior of the operations, initially assuming no concurrency and no failures, and to deal with concurrency and failures later. In other words, we imagine that the objects will exist in an environment in which all actions are executed sequentially, and in which actions never abort.

Although a sequential specification of this sort does not say anything explicit about permissible concurrency, it does impose limits on how much concurrency can be provided. Implementations can differ in how much concurrency is provided, but no implementation can exceed these limits. Therefore, it is important to understand what the limits are.

This section and the following section together provide a precise definition of permissible concurrency for an atomic type. This definition is based on two facts about Argus and the way it supports implementations of atomic type. First, in implementing an atomic type, it is only necessary to be concerned about active actions. Once an action has committed to the top, it is not possible for it to be aborted later, and its changes to atomic objects become visible to other actions. So, for example, an implementation of an atomic type needs to prevent one action from observing the modifications of other actions that are still active, but it does not have to prevent an action from observing modifications by actions that have already committed. Second, the only method available to an atomic type for controlling the activities of actions is to delay actions while they are executing operations of the type. An atomic type cannot prevent an action from calling an operation, although it can prevent that call from proceeding. Also, an atomic type cannot prevent an action that previously finished a call of an operation from completing either by committing or by aborting.

Given the sequential specification of the operations of a type, these facts lead to two constraints on the concurrency permitted among actions using the type. While an implementation can allow no more concurrency than permitted by these constraints, some implementations, like that for the built-in type generator atomic___array (see Section II.10), may allow less concurrency than permitted by their sequential specifications and our concurrency constraints.

The first constraint is that

 an action can observe the effects of other actions only if those actions committed relative to the first action.

This constraint implies that the results returned by operations executed by one action can reflect changes made by operations executed by other actions only if those actions committed relative to the first action. For example, in an atomic array *a*, if one action performs a *store*(*a*, 3, 7), a second (unrelated) action can receive the answer "7" from a call of *fetch*(*a*, 3) only if the first action committed to the top. If the first action is still active, the second action must be delayed until the first action completes. This first constraint supports recoverability since it ensures that effects of aborted actions cannot be observed by other actions. It also supports serializability, since it prevents concurrent actions from observing one another's changes.

However, more is needed for serializability. Thus, we have our second constraint:

 operations executed by one action cannot invalidate the results of operations executed by a concurrent action.

For example, suppose an action A executes the *size* operation on an atomic array object, receiving n as the result. Now suppose another action B is permitted to execute addh. The addh operation will increase the size of the array to n + 1, invalidating the results of the *size* operation executed by A. Since A observed the state of the array before B executed addh, A must precede B in any sequential execution of the actions (since sequential executions must be consistent with the sequential specifications of the objects). Now suppose that B commits. By assumption, A cannot be prevented from seeing the effects of B. If A observes any effect of B, it will have to follow B in any sequential execution. Since A cannot both precede and follow B in a sequential execution, serializability would be violated. Thus, once A executes *size*, an action that calls addh must be delayed until A completes.

15.5. Commuting Operations

To state our requirements more precisely, consider a simple situation involving two concurrent actions each executing a single operation on a shared atomic object X. (The actions may be executing operations on other shared objects also, but in Argus each object must individually ensure the atomicity of the actions using it, so we focus on the operations involving a single object.) A fairly simple condition that guarantees serializability is the following. Suppose X is an object of type T. X has a current state determined by the operations performed by previously committed actions. Suppose O_1 and O_2 are two executions of operations on X in its current state. (O_1 and O_2 might be executions of the same operation or different operations.) If O_1 has been executed by an action A and A has not yet committed or aborted, O_2 can be performed by the sequential specification of T) of performing O_1 on X followed by O_2 is the same as performing O_2 on X followed by O_1 . It is important to realize that when we say "effect" we include both the results returned and any modifications to the state of X.

The intuitive explanation of why the above condition works is as follows. Suppose O_1 and O_2 are performed by concurrent actions A and B at X. If O_1 and O_2 commute, then the order in which A and B are serialized globally does not matter at X. If A is serialized before B, then the local effect at X is as if O_1 were performed before O_2 , while if B is serialized before A, the local effect is as if O_2 were performed before O_1 . But these two effects are the same since O_1 and O_2 commute.

The common method of dividing operations into readers and writers and using read/write locking works because it allows operations to be executed by concurrent actions only when the operations commute. More concurrency is possible with our commutativity condition than with readers/writers because the meaning of the individual operations and the arguments of the calls can be considered. For example, calls of the atomic array operation addh always commute with calls of addi, yet both these operations are writers. As another example, store(X, i, e_1) and store(X, j, e_2) commute if $i \neq j$.

We require only that O_1 and O_2 commute when they are executed starting in the current state.

15.5 Commuting Operations

Consider a bank account object, with operations to deposit a sum of money, to withdraw a sum of money (with the possible result that it signals *insufficient funds* if the current balance is less than the sum requested), and to examine the current balance. Two withdraw operations, say for amounts m and n, do not commute when the current balance is the maximum of m and n: either operation when executed in this state will succeed in withdrawing the requested sum, but the other operation must signal *insufficient funds* if executed in the resulting state. They do commute whenever the current balance is at least the sum of m and n. Thus if one action has executed a withdraw operation, our condition allows a second action to execute another withdraw operation while the first action is still active as long as there are sufficient funds to satisfy both withdrawal requests.

Our condition must be extended to cover two additional cases. First, there may be more than two concurrent actions at a time. Suppose $A_1,...,A_n$ are concurrent actions, each performing a single operation execution $O_1,...,O_n$ respectively, on X. (As before, the concurrent actions may be sharing other objects as well.) Since $A_1,...,A_n$ are permitted to be concurrent at X, there is no local control over the order in which they may appear to occur. Therefore, all possible orders must have the same effect at X. This is true provided that all permutations of $O_1,...,O_n$ have the same effect when executed in the current state, where effect includes both results obtained and modifications to X.

The second extension acknowledges that actions can perform sequences of operation executions. Consider concurrent actions $A_1, ..., A_n$ each performing a sequence $S_1, ..., S_n$, respectively, of operation executions. This is permissible if all sequences $S_{j1}, ..., S_{jn}$ obtained by concatenating the sequences $S_1, ..., S_n$, in some order, produce the same effect. For example, suppose action A executed addh followed by remh on an array. This sequence of operations has no net effect on the array. It is then permissible to allow a concurrent action B to execute size on the same array, provided the answer returned is the size of the array before A executed addh or after it executed remh.

Note that in requiring certain sequences of operations to have the same effect, we are considering the effect of the operations as described by the specification of the type. Thus we are concerned with the abstract state of X, and not with the concrete state of its storage representation. Therefore, we may allow two operations (or sequences of operations) that do commute in terms of their effect on the abstract state of X to be performed by concurrent actions, even though they do not commute in terms of their effect on the representation of X. This distinction between an abstraction and its implementation is crucial in achieving reasonable performance.

It is important to realize that the constraints that are imposed by atomicity based on the sequential specification of a type are only an upper bound on the concurrency that an implementation may provide. A specification may contain additional constraints that further constrain implementations; these constraints may be essential for showing that actions using the type do not deadlock, or for showing other kinds of termination properties. For example, the specification of the built-in atomic types explicitly describes the locking rules used by their implementations; users of these types are guaranteed that the built-in atomic types will not permit more concurrency than allowed by these rules (for instance, actions writing different components of an array, or different fields of a record, cannot do so concurrently).

15.6. Multiple Mutexes

Section 15.2 grouphed a discussion of copying mater alignets to statute storage. That documeters is adequate for simple intelementations that use just one expland at housener. I is desirable to use more than one makes alight in appreciation at t for example. A او عداد partitioned data base would be implemented this way. By some dutalls that could be ignored when just one sta R n h perioder, the implementer must united and the d of a subset obtains and some problems that any when because papping in the 10000

The writing of mutan objects to stable damage is measurable for a ingention at pack grantion: either all materies modiliad by an action at a gran are. That is, if an action modified many burn and ma Nyr a graph githur all these objects will be recovered, or name of them will be. The on it contact to appearance be analysis and consistency among multiple makes eligente. Meaning restance is readilized regy all be responsible to a any new versions are incompany, of all them will be. dista. I is such to that now versions will be included at some of the second

Although materia objects meetiled by a single active are builded as about stores as a group, the copies are made one at a time. Increased and the databated are defined to be according to a time, the taxe and the databated are defined and the databated of the material of the material of the databated of the data

rup - struct [ited, excend: combined that the class of the two posteriors in the same. How express the opsiem is bandling the typical quantity of some college if the same institution bath continues contained in the deallo-quark, and while the is baggering a surface fraction if is modifying these contained. Then it is possible that when the featurally and possible theory is dealed in the container it contains for changes, but when the second containing is written its contain its contains if the dealer is the second container is written its contains in the dealer is the second container is written in the second container is another in the second container is another in the second container is another the second container is another is container in the second container is another in the second container in the second container is another in the second container is another in the second container is another in the second container in the second container in the second container in the second container is another in the second container in t

However, the representation invariant of the double-taxable maily is authining, for the following section. First rate that the information in stable storage is only all allowed allow granted. The suppose there is a creat. Now there are two possibilities:

104

¹⁴Boo Welhi, W. and Lishav, B., "Implementation of Paciflant, Abanin Data Types," ACM Transmotions on Programming Languages and Systems, values 7, number 2 (April 1989), pages 264-268.

- 1. Before that crash, *B* also committed to the top. In this case the data read back from stable storage is, in fact, consistent, since it must reflect *B*'s changes to both the *first* and *second* semiqueues.
- 2. B aborted or had not yet committed before the crash. In either case, B aborts. Therefore, the changes made to the *first* semiqueue by B will be hidden by the semiqueue implementation: at the abstract level, the two semiqueues do have the same state.

The point of the above example is that if the objects being written to stable storage are atomic, then the fact that they are written incrementally causes no problems.

On the other hand, when an atomic type is implemented with a representation consisting of several mutex objects, the programmer must be aware that these objects are written to stable storage incrementally, and care must be taken to ensure that the representation invariant is still preserved and that information is not lost in spite of incremental writing. If the implementation of a type requires that one mutex object (call it M1) be written to stable storage before another (call it M2), then the write of M1 must be contained in an action that commits to the top before the action that writes M2 is run.

I Syntax



We use an extended BNF grammer to define the system. The gameral form of a production is

nonterminal

	alternative
1	alternative
İ	t
İ	alternative

The following extensions are used:

8	a list of one or more an accorded by commune: "af or "a, a" or "a, a, a", et	6
{a}	a sequence of zero or more as: " "er "e" er "e e", elo.	
[a]	an estimat a: " or "a".	

Nonterminal symbols appear in normal face. Reserved wonte appear in bald face. All other terminal symbols are nonalphabetic and appear in normal face.

module		equate) equates equate) generation equate) generation equate) generator equate) develor equate) chaster	
equates	;:æ	ich - equates [parties [where]] equate (equate } end ich	
guardian	::=	icin - guandian (porms) is idn , . { expanse } { state_deci } { reconvertinity-and } { humbgrowind-body-and } { operation } creater { spe ord idn	[where]
cluster	:: 2	idn okunter { pannie } is opidn ; { scipanie } cop type:_con { comver } routine { routine } end idn	

Syntax

and a standard where and a standard free free provides the free to the stand of the standard free to a standard

operation	::= 	creator handler routine
routine	::= 	procedure iterator
procedure	::=	idn = proc [parms] args [returns] [signals] [where] routine_body end idn
iterator	::=	idn = Iter [parms] args [yields] [signals] [where] routine_body end idn
creator	::=	idn = creator args [returns] [signals] routine_body end idn
handler	::=	idn = handler a rgs [returns] [signals] routine_body end idn
routine_body	::=	{ equate } { own_var } { statement }
parms	::=	[parm ,]
parm	::= 	idn , : type idn , : type_spec
args	::=	([decl ,])
decl	::=	idn , : type_spec
returns	::=	returns (type_spec ,)
yields	::=	yleids (type_spec ,)
signals	::=	signals (exception,)
exception	::=	name [(type_spec ,)]

108

I Syntax

opidn	::= 	idn t ransmit
where	::=	where restriction ,
restriction	::= 	idn h as oper_decl , idn In type_set
type_set	::= 	{ idn idn has oper_decl , { equate } } idn reference \$ name
oper_decl	=::= 	name , : type_spec transmit
constant	:::= 	expression type_spec
state_deci	::= 	[stable] decl [stable] idn : type_spec := expression [stable] decl , := call
equate	::= 	idn = constant idn = type_set idn = reference
own_var	::= 	own deci own idn : type_spec := expression own deci , := call [@ primary]

109

statement

::= decl

idn : type spec := expression decl , := call [@ primary] idn , := call [@ primary] idn , := expression , primary . name := expression primary [expression] := expression call [@ primary] fork call seize expression do body end Dause terminate enter_stmt coenter coarm { coarm } end [abort] leave while expression do body end for stmt if stmt tagcase_stmt tagtest stmt tagwait stmt [abort] return [(expression,)] yield [(expression , ...)] [abort] signal name [(expression , ...)] [abort] exit name [(expression , ...)] abort break [abort] continue begin body end statement [abort] resignal name , { when_handler } statement except [others_handler] end

enter_stmt

112

I

enter topaction body end enter action body end

•,

I Syntax

coarm	::=	armtag [foreach deci , in call] body
armtag	::= 	action topaction process
for_stmt	::= 	for [deci ,] in call do body end for [idn ,] in call do body end
if_stmt	::=	If expression then body { elself expression then body } [else body] end
tagcase_stmt	::=	tagcase expression tag_arm { tag_arm } [others : body] end
tagtest_stmt	::=	tagtest expression atag_arm { atag_arm } [others : body] end
tagwait_stmt	8 - ant 2 - ant	tagwait expression atag_arm { atag_arm } end
tag_arm	::=	tag name , [(idn : type_spec)] : body
atag_arm		tag_kind name , [(idn : type_spec)] : body
tag_kind	=:: 	tag wtag
when_handler	::= 	when name , [(deci ,)] : body when name , (*) : body
others_handler	::=	others [(idn : type_spec)] : body
body	::=	{ equate } { statement }

111

and the second s

1.000 1.000

小学、这些情况,这些一个问题情况,不可能是我的问题都能够是我的问题,我们的"

type_spec	::= n	luli
		node
	1	lood
		Int
		rea!
		char
		string
		any
		image
		rep
		cvt
		sequence [type_actual]
		array [type_actual]
		atomic_array [type_actual]
		struct [field_spec,]
	Į	record [field_spec ,]
		atomic_record [field_spec ,]
		oneof [field_spec ,]
		variant [field_spec ,]
		atomic_variant [field_spec]
	ļ	proctype ([type_spec ,]) [returns] [signals]
	ļ	itertype ([type_spec ,]) [yields] [signals]
	1	creatortype ([type_spec ,]) [returns] [signals]
	ļ	handlertype ([type_spec ,]) [returns] [signals]
	ļ	mutex [type_actual]
	l	reference
field_spec	::=	name,: type_actual
reference	::=	idn
		idn [actual_parm ,]
	İ	reference \$ name
actual_parm	::=	constant
	ł	type_actual
type_actual	::=	type_spec [with { where opbinding , }]
opbinding	::= n	ame , : primary

I Syntax

Instant of parts

expression	::=	primary		
		call @ primary		
		(expression)		
		~ expression	%6	(precedence)
		- expression	%6	
	1	expression ** expression	%	5
		expression // expression	%	4
		expression / expression	%	4
		expression * expression	%	4
		expression expression	%	3
		expression + expression	%	3
		expression - expression	%	3
		expression < expression	%	2
		expression <= expression	%	2
		expression = expression	%	2
		expression >= expression	%	2
		expression > expression	%	2
		expression ~< expression	%	2
		expression ~<= expression	%	2
		expression ~= expression	%	2
		expression ~>= expression	%	2
		expression ~> expression	%	2
		expression & expression	%	1
	1	expression cand expression	%	1
		expression expression	%	0
	1	expression cor expression	%	0
primary	::=	entity		
		call		
		primary . name		
		primary [expression]		

call ::= primary ([expression, ...])

entity	::=	nil
		true
	1	false
		int_literal
		real_literal
		char_literal
		string_literal
		self
		reference
		entity - name
		entity [expression]
		<pre>bind entity ([bind_arg ,])</pre>
		type_spec \$ { field , }
		type_spec \$ [[expression :] [expression ,]]
		type_spec \$ name [[actual_parm ,]]
	1	up (expression)
	1	down (expression)
field	::=	name , : expression
bind ara	::=	*
		expression
	•	1

I Syntax

44.17

Comment: a sequence of characters that begins with a percent sign (%), ends with a newline character, and contains only printing ASCII characters and horizontal tabs in between.

Separator: a blank character (space, vertical tab, horizontal tab, carriage return, newline, form feed) or a comment. Zero or more separators may appear between any two tokens, except that at least one separator is required between any two adjacent non-self-terminating tokens: reserved words, identifiers, integer literals, and real literals.

Reserved word: one of the identifiers appearing in **bold** face in the syntax. Upper and lower case letters are not distinguished in reserved words.

Name, idn: a sequence of letters, digits, and underscores that begins with a letter or underscore, and that is not a reserved word. Upper and lower case letters are not distinguished in names and idns.

Int_literal: a sequence of one or more decimal digits (0-9) or a backslash (1) followed by any number of octal digits (0-7) or a backslash and a sharp sign (1/4) followed by any number of hexadecimal digits (0-9, A-F in upper or lower case).

Real_literal: a mantissa with an (optional) exponent. A mantissa is either a sequence of one or more decimal digits, or two sequences (one of which may be empty) joined by a period. The mantissa must contain at least one digit. An exponent is 'E' or 'e', optionally followed by '+' or '-', followed by one or more decimal digits. An exponent is required if the mantissa does not contain a period.

Char_literal: a character representation other than single quote, enclosed in single quotes. A character representation is either a printing ASCII character (octal value 40 through 176) other than backslash, or an escape sequence consisting of a backslash () followed one to three printing characters as shown in Table 6-1 or Table I-1 below.

String_literal: a sequence of zero or more character representations other than double quote, enclosed in double quotes.

Table I-1 shows most of the character literals supported by Argus, except for the higher numbered octal escape sequences. For each character, the corresponding octal literal, hexadecimal literal, and normal literal(s) are shown. Upper or lower case letters may be used in escape sequences of the form \#**, \^*, \!*, \b, \t, \n, \v, \p, and \r. Note that an implementation need not support 256 characters, in which case only a subset of the literals listed will be legal.

13.1734

'\000' '\#00' '\^@'	'\100' '\ #4 0' '@'	<u>`\200' '\#90' '\}@'</u>	`\300' '\#C0' '\&@'
'\001' '\#01' '\^A'	101' 1441' A'	1201' 1481' 1\A	1301' '\#C1' '\#A'
1002' 1402' 1AB'	1102' 1442' 'B'	1202' 1402' 11R'	1302' 14C2' 14B'
1003' 1403' 140'	1103' 1443' 'C'	1203' 1483' 110'	1303' '\#C3' \&C'
1004' 1#04' 1AD'	1104' 1444' 'D'	1204' 1494' 140'	1904' 14C4' 18D'
	100. 1440. 1		1306 WC6 VAF
1007' 1 # 07' 1^G'	'\107' '\#47' 'G'	1207' 1 #8 7' 11G'	1307" 1#C7" 1&G'
∿010' \#08' '\^H' '\b'	'\110' \#4 8' 'H'	1210' 1 #88' 11H'	1310' 1WC8' 1&H'
1011' 1#09' 1^l' '\r	1111' \#49' '#	1211' 1489' 11	1311' 'MC9' 'Al'
1012' \#0A' 14.I' 1n'	112' \#4A' '.P	1212' 1#RA' 1LI'	1312' 18CA' 14.1
1013' MOB' 14K' 1V'	113' \#48' 'K'	1213' 1498' 116	1313' MCR' MK'
1014' 1#0C' 10L' 10c'	114' 144C' 'L'	1210 WOD 111	1214' 14CC' 181'
			1916 WCE VAN
	1117 WHAT 'O'	1217 1 #81- 1410	
'\020' '\#10' '\^P'	'\120' '\#50' 'P'	`\220' `\#9 0' '\ ! P'	1320' 1#D0' '\&P'
1021' 1#11' 1AO'	\121' \#51' 'Q'	1221' 1491' 1KO'	1321' 'MD1' '\&O'
1022' 1#12' 1AR'	122' 1#52' 'R'	1222' 1482' 18'	1322' 14D2' 14R'
1023' 1412' 146'	1122' 1453' 'S'	1222' 1402' 115'	1222' 1402' 145'
1024' 141 4' 1AT	1120 WOO 0	10041 34041 3 ET	
V26 1#16 1*V	126. 1 856. V	1226 1998 11	7358. MD8. MV.
1027' 1#17' 1 ^₩ '	127' 1 857' 'W '	`\227' '\#97' '\!₩'	`\327' '\#D7' '\&W'
`\030' `\# 18' '\^X'	130' \#58' 'X '	'\230' '\#98' '\!X'	1330' '\#D8' '\&X'
`\031' \#19' '\^Y'	`\131' \#59' 'Y '	1231' 1499' 1Y	1331' 1#D9' 1&Y'
1032' 1#1A' 1^7'	132' 1#5A' '7'	1232' \#9A' \IZ'	1332' 140A' 187'
1033' 1#1B' 14"	133' 1458' 1	1233' 1408' 11'	1333' 1408' 141'
1034' 1#10' 141'	1134' 3450' 11'	1234' 140C' 18'	1924' 1#00' 14\'
3025' 3#1D' 3AI'	1104 WOO 11		
000 W10 [*]			
1037 WHEN 1	137 401-	1237 1 89 11_	1337 WDF 14
\040' \#20' ' '	`\1 40' `\#6 0' '"	'\240' \#A 0' \& '	1340' 1#E0' 14"
\041' \#21' '!'	`\141' `\#6 1' 'a'	1241' 1#A1' 1&!	1341' 1#E1' 1&a'
`\042' `\#22 ' ' '' '\"'	1142' 1462' 'b'	1242' 1442' 14"	1342' 1#E2' 14b'
1043' 1#23' '#'	1143' 1463' 'c'	1243' 1#43' 18#'	1343' 1#F3' \&c'
1044' 1#24' 'S'	144' 1464' 'd'	1244' 1844' 185'	1344' 1#F4' 18d'
1045' 1#25' '%'	145' 1465' 'a'	1945! 1#45! 144	1245' 1455' 120'
1046' 1426' '&'	11-60 WOO U		
1047' 1407' N"		\∠400 \#7400 Vator	
VU4/ V#2/ \	147 WO7 g	\24/ \#A/ \G	V34/ WE/ Vag
`\050' `\#28' '('	'\150' '\#68' 'h'	`\250' `\#A8 ' '\&('	1350' 1#E8' 14h'
\051' '\#29' ') '	`\151' '\#69' 'i'	1251' 1#A9' 1&)'	1351' 1#E9' 1&I'
1052' 1#2A' ""	152' 1#6A' 'i'	1252' \#AA' '\A*'	1352' 1#EA' 14!
1053' 1#2B' '+'	153' \#6B' 'k'	1253' 1#AR' 14+'	1353' \#FB' \&k'
1054' 1#2C' ' '	1154' \#6C' '!'	1254' 1#AC' 18 '	1354' 14FC' 14!
1055' 1#2D' '-'	155' \#6D' 'm'	1255' 1#AD' 18_'	1355' 1#FD' 12m'
1056' 1#2E' ' '	156' \#8E' 'n'	1068' 1#AE''12 '	1258' 1455' 12 mil
	100 WOE II 1671 140E 1-1		NOU WEE WIT
WU/ WEF /	VID/ WHOF D	VCD/ VF/NE VOV	VOO/ WEET VOLO

Table I-1: Character Escape Sequences

化二氯化合物 化二氟化物 化合物 化合物 网络拉拉拉拉拉拉拉拉拉拉拉

e presidente de la composición de la composición de la composición de la composición de la composición de la c

1080' 1499' 10	1100' MITO' 'p'	1207 140	F 165	1980	100 Vac
1061' W91' 'T'	1161' W71' W		TMT	1281	WPT'Net
1082 1082 2	1162 W72 Y	1000	-	-	-
1067 1097 3	1165' 1675' 'Y	and a second second			1000 Yes'
1084' 1434' '4'	1164 1074 1				1000
1085' 1025' '5'	1100 W75'W				
1000' '490' '5'	1100 N7V V	1		-	WIN YOU
1067 1837 7	1167 WTT W		747	1987	WF7 10W
1070 1038 18	1170 W/W X	1019 10		1070	1075 14x
1071' W90' '9'	171' W78' Y	1271' 10	2.07	1971	100 W.
1072 WBA' '	178 W7A'Y	1			
1073' 1498' '	1173 WAR T			1012	MIN W
1074' WBC' '2	1176 1070 T				WFC W
1075' 1080' '-'	1178 1070 1				1007 101
1076' WSE' 5'	1178 WRE			100	
1077 188F '?	1177 WIP WY		707	1077	W

Appendix II Built-in Types and Type Generators

The following sections specify the built-in types and the types produced by the built-in type generators of Argus. For each type and for each instance of each type generator, the objects of the type are characterized, and all of the operations of the type are defined. (An implementation may provide additional operations on the built in types, as long as these are operations that could be implemented in terms of those described in this section.)

All the built-in types (except for any) are transmissible. All instances of the built-in type generators (except for proctype and itertype) are transmissible if all their type parameters are transmissible. Transmission of the built-in types preserves value equality, except for objects of type real. However, in a homogeneous environment, reals can be transmitted without approximations. In a homogeneous environment, the only possible encode or decode failures are exceeding the representation limits of an Image, mutating the size of an array or atomic__array while it is being encoded or decoded, and improper decoding of cyclic objects (see Section 14.4).

All operations are indivisible except at calls to subsidiary operations (such as Int\$similar within array[Int]\$similar), at yields, and while waiting for locks.

The specifications given below are informal and are adapted from the book Abstraction and Specification in Program Development (Liskov, B. and Guttag, J., MiT Press, 1986). A specification starts out by giving a list of the operations and declarations of any formal parameters for the type. This is followed by an overview, which gives an introduction to the type and if necessary defines a way of describing the type's objects and their values. Following this the individual operations are described. For each operation there is a heading and a statement of the operation's effects. In the heading, the return values may be given names. The effects section describes the normal and exceptional behavior of the operation. The effects given are abstract, that is they are described using the vocabulary (or model) defined in the overview section. For example, objects of type int are described using mathematical integers. Thus arithmetic expressions and comparisons used in defining int operations are to be computed over the domain of mathematical integers.

An operation that (abstractly) mutates one of its arguments lists the arguments that it mutates in the clause following the word modifies. An operation is not allowed to mutate any objects, except for those listed in the modifies clause. (For the built-in mutable atomic type generators, modification only refers to the sequential state; it does not refer to changes in the locking information kept for each object.) When an argument, say *a*, is mutated, it is often necessary to describe its state at the start of the call as well as its final state at the end of the call. We use the notation a_{pre} for *a*'s state at the start of the call and the notation a_{post} for its state at the end of the call.

Some operations of the built in type generators are only defined if the type generator is passed appropriate actual routine parameters (see Section 12.6). For example, the copy operation of the array

type generator, is only defined if there is an actual parameter passed (explicitly or implicitly) for the type parameter's *copy* operation. Thus **array[Int]\$copy** is defined but **array[any]\$copy** is not defined. These requirements are stated in a **requires** clause that precedes the description of the operation's effect. The type of the expected routine is also described; remember that the actual operation parameter can have fewer signals (see Section 6.1 and Section 12.6).

By convention, the order in which exceptions are listed in the operation type is the order in which the various conditions are checked.

Operations with the same semantics (for example, **null\$**equal and **null\$**similar) or that can be described in the same way (for example, **Int\$**add and **Int\$**sub) are grouped together to save space.

In defining the built-in types, we do not depend on users satisfying any constraints beyond those that can be type-checked. This decision leads to more complicated specifications. For example, the behavior of the *elements* iterator for arrays is defined even when the loop modifies the array.

II.1. Null

null = data type is copy, equal, similar, transmit

Overview

The type null has exactly one, immutable, atomic object, represented by the literal nil. Nil is generally used as a place holder in type definitions using oneofs or variants.

Operations

equal = proc (n1, n2: null) returns (bool) similar = proc (n1, n2: null) returns (bool) effects Returns true.

copy = proc (n: null) returns (null) transmit = proc (n: null) returns (null) effects Returns nil.

II.2. Nodes

node = data type is here, copy, equal, similar, transmit

Overview

Objects of type **node** are immutable and atomic, and stand for physical nodes. Implementations should provide some mechanism for translating a node "address" into a **node** object and vice versa. (However, these do not have to be operations of type **node**.)

Operations

here = proc () returns (node) effects Returns the node object for the caller's node.

equal = proc (n1, n2: node) returns (bool) similar = proc (n1, n2: node) returns (bool) effects Returns true if and only if n1 and n2 are the same node. copy = proc (n: node) seturne (ne transmit - pres (n: htt let mineres (no allacts Returns n.

II.3. Booleans

bool - date type is and, or, not, equal, similar, copy, transmit

Overview

The two immutable, atomic objects of type logal, with thereis true and false, represent logical truth Values.

The language also provides the operators saind and day for conditional evaluation of boolean expressions, see Section 9.15.

Operations

```
and - prov (b1, b2; boat) returns (boat)
      effects Returns true I bt and bit are both true; returns false otherwise.
```

or - proc (b1, 52: basil automa (basil) effects fishane true I allow of a bit is true; returns tilles atterwise.

not - proc (b: bool) relative (knol) effecto Relative tables I à la tras; relative II à la table.

equal - proc (51, bit: basis antonia (basis) similar - proc (54, bit: Basis antonia (basis) aliante Paturne anto 2 of and at are both thus or bath lates; otherwise returns false.

copy - pros (b: band) estimates (b) 1FILES

ellegis Rotema à.

II.4. Integers

int - date type to add, sub, mul, minue, div, med, power, abe, from to, by, from to, max, min, partse, unpartse, it, is, ge, gt, atput, atmiter, capp, atminuelt

Overview

Objects of type int are immutable and atas the annual section and the subrange of the en. The e ical inte in and can very maine and the second 800000 de a dini characters --- and estimate

Operatione

- add = proc (x, y: int) returns (int) algebia (se
- sub prac (x, y: lat) and read for the
- mul philo (x, y; his officials The and the second and an and the second
 - cation operations. it. and in The second second second second second second second second second second second second second second second s and a second second They shared and interest.

minus - proc (x: int) returns (int) algorith (overflow)

effects Returns the res native of x signals overflow if the result would lie outside the represented interval

div = proc (x, y: int) returns (q: int) alignate (zero, divide, overflow) effects Signals zero, divide if y = 0. Glimmung the integer quotient of dividing x by y; that is, $x = y \cdot q + t$; for some integer r such that $0 \le r < |y|$. Signate overflow if q would lie cutside the represented interval.

mod = proc (x, y: int) setures (r: int) elyminis (xere, childs, evention) effects Signals zero, childs if y=0. Character cancer the integer remainder of childing x by y; that is, r is such that $0 \le r < |y|$, for some integer $q: x = y \cdot q + r$. Signals overflow if r the state in the to Signals overflow if r would be cutately the remainded inserval.

power - proc (x, y: Int) returns (int) signate (resulting expectant, a officite Signals require (returned if y < 0. Country of ecto Signale auguitos, consi reast would be automate the a e si; signals overflow if the Stand Street Street Contraction of the second L Pattrel 100

abs - proc (x: int) esturns (int) algorite (everilen) effects Fisture the algorithe value of x; signals overlian if the next would is outside the represented interval.

from to by - Ner (from, to, by: init) yields (init)

Accuracy, ..., An end to be a second effects Yields the integens from from to is, becamenting by by each time, that is, yields from, formvby, ..., democracy, where the two because produce trager such that from only < is.

- from to = Ner (New, to: Init) yields (Init) effects The effect is identified to from to by(from, to, 1).
- max = proc (x, y: ini) returns (ini) effecte If x ≥ y, then returns x, otherwise returns y.
- min = proc (x, y: int) returns (int) effects if $x \le y$, then returns x, otherwise returns y.

parse - proc (a: string) returns (bit) algorith (bot) iso effects S must be at integral (bits) into a

entional leading plus or minus este S mest he an atomic Black and Aller and sign; I s is not al this term, denote soit corresponding to s; signals souther I the s . inte the inte ne pet at its outside the represented interval.

unpares - proc (x: ini) returne (string)

lects Produces the stating representing the integer value of x in decimal notation, preceded by a minus sign if x < 6. Leading zeros are approximate, and there is no leading plus sign for positive integers.

- It = proc (x, y; int) returns (beet)
- gt proc (x, y: Int) returns (b
- le = prec (x, y: 100) an
- ge proc (x, y: init) in 1
 - indext patients relations. white Through and the d

equal - proc (x, y: int) returns (incl) similar - proc (x, y: int) estimate (basis) alliable Paturns tage if x and y are the same integer; returns taken otherwise.

CODY = DFDC (x; int) selector (int) offects Returns x.

```
transmit = proc (x: int) returns (y: int) signals(failure(string))
```

effects Returns y such that x = y or signals failure if x cannot be represented in the implementation on the receiving end.

II.5. Reals

real = data type is add, sub, minus, mul, div, power, abs, max, min, exponent, mantissa, i2r, r2i, trunc, parse, unparse, it, ie, ge, gt, equal, similar, copy, transmit

Overview

The type real models a subset of the mathematical numbers. It is used for approximate or floating point arithmetic. Reals are immutable and atomic, and are written as a *mantissa* with an optional *exponent*. See Appendix I for the format of real literals.

Each implementation represents a subset of the real numbers in:

D = {-real max, -real min} U {0} U {real min, real max}

where

0 < real min < 1 < real max

Numbers in D are approximated by the implementation with a precision of p decimal digits such that:

∀r∈D	Approx(r) ∈ Real
∀r ∈ Real	Approx(r) = r
∀r∈ D – {0}	$ (Approx(r) - r)/r < 10^{1-p}$
∀r,s∈ D	$r \leq s \Rightarrow Approx(r) \leq Approx(s)$
∀r∈D	Approx(-r) = -Approx(r)

We define Max_width and Exp_width to be the smallest integers such that every nonzero element of real can be represented in "standard" form (exactly one digit, not zero, before the decimal point) with no more than Max_width digits of mantissa and no more than Exp_width digits of exponent.

Real operations signal an exception if the result of a computation lies outside of D; overflow occurs if the magnitude exceeds real_max, and underflow occurs if the magnitude is less than real_min.

Operations

div = proc (x, y: real) returns (real) signals (zero_divide, overflow, underflow)
effects If y = 0, signals zero_divide. Otherwise returns approx(x/y); signals overflow or
underflow if x/y is outside of D.

power = proc (x, y: real) returns (real) signals (zero divide, complex result, overflow, underflow) effects if x = 0 and y < 0, signals zero divide. If x < 0 and y is nonintegral, signals complex result. Otherwise returns an approximation to x^{μ} , good to p significant digits; signals overflow or underflow if x^V is outside of D. abs = proc (x: real) returns (real) effects Returns the absolute value of x. max = proc (x, y: real) returns (real) effects if $x \ge y$, then returns x, otherwise returns y. min = proc (x, y: real) returns (real) effects if $x \le y$, then returns x, otherwise returns y. exponent = proc (x: real) returns (int) signals (undefined) effects if x = 0, signals undefined. Otherwise returns the exponent that would be used in representing x as a literal in standard form, that is, returns $max(\{i \mid abs(x) \ge 10^i\})$ mantissa - proc (x: real) returns (real) effects Returns the mantissa of x when represented in standard form, that is, returns approx(x/10°), where e = exponent(x). If x = 0.0, returns 0.0. i2r = proc (i: int) returns (real) signals (overflow) effects Returns approx(); signals overflow if / is not in D. r2i = proc (x: real) returns (int) signals (overflow) effects Rounds x to the nearest integer and toward zero in case of a tie. Signals overflow if the result lies outside the represented range of integers. trunc = proc (x: real) returns (int) signals (overflow) effects Truncates x toward zero; signals overflow if the result would be outside the represented range of integers. parse = proc (s: string) returns (real) signals (bad format, overflow, underflow)

and and a state of the second of the second state of the second st

effects Returns approx(z), where z is the value represented by the string s (see Appendix I). S must represent a real or integer literal with an optional leading plus or minus sign; otherwise signals bad_format. Signals underflow or overflow if z is not in D.

unparse = proc (x: real) returns (string)

effects Returns a real literal such that parse(unparse(x)) = x. The general form of the literal is:

[-] *i_field.f_field* [*e* ± *x_field*]

Leading zeros in <u>i</u>field and trailing zeros in <u>f</u> field are suppressed. If x is integral and within the range of represented integers, then <u>f</u> field and the exponent are not present. If x can be represented by a mantissa of no more than Max width digits and no exponent (that is, if $-1 \le exponent(arg1) < Max_width$), then the exponent is not present. Otherwise the literal is in standard form, with Exp_width digits of exponent.

it = proc (x, y: real) returns (bool)

le = proc (x, y: real) returns (bool)

ge = proc (x, y: real) returns (bool)

gt = proc (x, y: real) returns (bool)

effects These are the standard ordering relations.

equal = proc (x, y: real) returns (bool)

similar - proc (x, y: real) returns (bool)

effects Returns true if x and y are the same number; returns false otherwise.

copy - proc (x: neet) returns (real) die Roberts x.

tranomit - proc (x: noil) relume (mai) algoats altunto finiume accuration (x) within function for the receiving and one this secondaria and.

1.6. Characters

char - data type to i2c, c2i, it, ie, ge, gt, equal, similar, copy, transmit

Overview

class are immetable and alomic, and Type oher provides the alphabet for test mentalation. Ch form an ordered set. Every inclusion of the set characters; the first till characters are the fill of the set and the but so sure than 512, é andiar.

Operations /2c and c2 convert balances into and along for ASCII coding for the first 128 charactery). The constant character constants is and characters are reacted and committee up to start the the start constants in the terminate character. The numbering

Printing ABCII charactere (estat 46 through social 176), other than single quote or backshoch, can be writtening that descends a sector of character through and tables of allocations and tables of allocations and tables.

Operations

- i2c proc (x: Int) returns (altar) signals (lings) shut) offices Picture ing strandstrangeneiting is ing to an elements illeged other if x is not in the range (0, oher had.
- c2i proc (c: char) manne (ini) offecte Roturne the integ
 - der consepondies to c fusing the ABCII coding if c is an ASCII character).
- k proc (c1, c2: ohar) returne (
- Øř le - proc (ci, ci: ch
- 00 pies (c1. c2: e
- gi pres (ci, cit: e

1009 Sec. 19 Signal Co initia Thans inc. Nambaring al ala Tayle -

equal - pres (c1, c2; char), alterna (and), similar - pres (c1, c2; char), alterna (and), eliuse Record and at as the same character, i.e., returns (c2(c1) - c2(c2)).

copy - proc (c1: glust) setures (char) ellente Pieturne cf.

tranend - proc (c1: char) subsche (char) signalis frances no notation on the implementation on the receiving and.

II.7. Strings

string = data type is c2s, concat, append, substr, rest, size, empty, fetch, chars, indexs, indexc, s2ac, ac2s, s2sc, sc2s, it, ie, ge, gt, equal, similar, copy, transmit

Overview

Type string is used for representing text. A string is an immutable and atomic tuple of zero or more characters. The characters of a string are indexed sequentially starting from one. Strings are lexicographically ordered based on the ordering for characters.

والمراجعة المحمول والعربية وتعريجه

Sec. P.

A string literal is written as a sequence of zero or more character representations enclosed in double quotes. See Appendix I for a description of the character escape sequences that can be used within string literals. No string can have a size greater than *int_max*; however, an implementation may restrict string lengths to a value less than *int_max*. If the result of a string operation would be a string containing more than the maximum number of characters, the operation signals *limits*.

Operations

c2s = proc (c: char) returns (string) effects Returns a string containing c as its only character.

concat = proc (s1, s2: string) returns (r: string) signals (limits)

effects Returns the concatenation of s1 and s2. That is, /[/j=s1[/] for / an index of s1 and /[size(s1)+i]=s2[/] for / an index of s2. Signals limits if r would be too large for the implementation.

append = proc (s: string, c: char) returns (r: string) signals (limits)

effects Returns a new string having the characters of s in order followed by c. That is, r[size(s)+1] = c. Signals limits if the new string would be too large for the implementation.

substr = proc (s: string, at: int, cnt: int) returns (string) signals (bounds, negative_size)

effects if cnt < 0, signals negative_size. If at < 1 or at > size(s)+1, signals bounds. Otherwise returns a string having the characters s[a], s[a+1], ... in that order; the new string contains min(cnt, size-at+1) characters. For example,

> substr ("abcdef", 2, 3) = "bcd" substr ("abcdef", 2, 7) = "bcdef" substr ("abcdef", 7, 1) = ""

Note that if min(cnt, size-at+1) = 0, substr returns the empty string.

- rest = proc (s: string, i: int) returns (r: string) signals (bounds) effects Signals bounds if i < 0 or i > size(s) + 1; otherwise returns a string whose first
 - character is s[i], whose second is s[i+1], ..., and whose size(i)th character is s[size(s)]. Note that if i = size(s)+1, rest returns the empty string.
- size = proc (s: string) returns (int) effects Returns the number of characters in s.
- empty = proc (s: string) returns (bool) effects Returns true if s is empty (contains no characters); otherwise returns false.
- fetch = proc (s: string, i: int) returns (char) signals (bounds) effects Signals bounds if i < 0 or i > size(s); otherwise returns the ith character of s.
- chars = iter (s: string) yields (char) effects Yields, in order, each character of s (i.e., s[1], s[2], ...).

indexs = proc (s1, s2: string) returns (int) effects if s1 occurs as a substring in s2, returns the least index at which s1 occurs. Returns 0 if s1 does not occur in s2, and 1 if s1 is the empty string. For example, indexs("abc", "abcbc") = 1 indexs("bc", "abcbc") = 2 indexs("", "abcde") = 1 indexs("bcb", "abcde") = 0 indexc = proc (c: char, s: string) returns (int) effects if c occurs in s, returns the least index at which c occurs; returns 0 if c does not occur in s. s2ac = proc (s: string) returns (array[char]) effects Stores the characters of a as elements of a new array of characters, a. The low bound of the array is 1, the size is size(s), and the ith element of the array is the ith character of s, for $1 \le i \le size(s)$. ac2s = proc (a: array[char]) returns (string) effects This is the inverse of s2ac. The result is a string with characters in the same order as in a. That is, the ith character of the string is the (i+arravichariSlow(a)-1)th element of a. s2sc = proc (s: string) returns (sequence(char)) effects Transforms a string into a sequence of characters. The size of the sequence is size(s). The th element of the sequence is the th character of s, for $1 \le i \le size(s)$. sc2s = proc (s: sequence(char)) returns (string) effects This is the inverse of a2sc. The result is a string with characters in the same order as in s. That is, the Ah character of the string is the Ah element of s. It = proc (s1, s2: string) returns (bool) le = proc (s1, s2: string) returns (bool) ge = proc (s1, s2: string) returns (bool) gt = proc (s1, s2: string) returns (bool) effects These are the usual lexicographic ordering relations on strings, based on the ordering of characters. For example, "abc" < "aca" "abc" < "abca" equal = proc (s1, s2: string) returns (bool) similar = proc (s1, s2: string) returns (bool) effects Returns true if s1 and s2 are the same string; otherwise returns false. copy = proc (s1: string) returns (string) effects Returns s1. transmit = proc (s1: string) returns (string) signals (failure(string))

effects Returns s1. Signals failure only if s1 is not representable on the receiving end.

II.8. Sequences

sequence = data type [t: type] is new, e2s, fill, fill_copy, replace, addh, addl, remh, reml, concat, subseq, size, empty, fetch, bottom, top, elements, indexes, a2s, s2a, equal, similar, copy, transmit

Overview

Sequences represent immutable tuples of objects of type *t*. The elements of the sequence can be indexed sequentially from 1 up to the size of the sequence. Although a sequence is immutable, the elements of the sequence can be mutable objects. The state of such mutable elements may change; thus, a sequence object is atomic only if its elements are also atomic.

Sequences can be created by calling sequence operations and by means of the sequence constructor, see Section 6.2.8.

Any operation call that attempts to access a sequence with an index that is not within the defined range terminates with the *bounds* exception. The size of a sequence can be no larger than the largest positive int (*int_max*), but an implementation may restrict sequences to a smaller upper bound. An attempt to construct a sequence which is too large results in a *limits* exception.

Operations

- new = proc () returns (sequence[t]) effects Returns the empty sequence.
- e2s = proc (elem: t) returns (sequence[t]) effects Returns a one-element sequence having elem as its only element.
- fill = proc (cnt: Int, elem: t) returns (sequence[1]) signals (negative_size, limits) effects if cnt < 0, signals negative_size. If cnt is larger than the maximum sequence size supported by the implementation, signals limits. Otherwise returns a sequence having cnt elements each of which is elem.
- fill copy = proc (cnt: int, elem: t) returns (sequence(t))
 - signals (negative_size, limits, failure(string))
 - requires t has copy: proctype (t) returns (t) signals (failure(string)
 - effects if *cnt* < 0, signals *negative_size*. If *cnt* is bigger than the maximum size of sequences that the implementation supports, signals *limits*. Otherwise returns a new sequence having *cnt* elements each of which is a copy of *elem*, as made by *t\$copy*. Note that *\$copy* is called *cnt* times. Any *failure* signal raised by *t\$copy* is immediately resignalled. This operation does not originate any *failure* signals by itself.

replace = proc (s: sequence[t], i: int, elem: t) returns (sequence[t]) signals (bounds)

effects if i < 1 or i > high(s), signals bounds. Otherwise returns a sequence with the same elements as s, except that elem is in the th position. For example, replace(sequence[int]\$[2,5], 1, 6) = sequence[int]\$[6, 5]

addh = proc (s: sequence[t], elem: t) returns (r: sequence[t]) signals (limits)

effects Returns a sequence with the same elements as s followed by one additional element, elem. That is, r[i]=s[i] for i an index of s, and r[size(s)+1]=elem. If the resulting sequence would be larger than the implementation supports, signals limits.

addl = proc (s: sequence[t], elem: t) returns (r: sequence[t]) signals (limits)

effects Returns a sequence having elem as the first element followed by the elements of s in order. That is, f[1]=elem and f[i]=a[i-1] for i = 2, ..., size(r). If the resulting sequence would be larger than the implementation supports, signals limits.

remh = proc (s: sequence[t]) returns (r: sequence[t]) signals (bounds)

effects if s is empty, signals bounds. Otherwise returns a sequence having all elements of s in order, except the last one. That is, size(r)=size(s)-1 and f(i)=s(i) for i = 1, ..., size(s)-1.

remi - proc (s: eequence)@ returne (r: entremant)@ elevate (munit) effecte if a is empty, signals branch. Charles anothe every a accuence containing all elements of s in order, except the first one. That is, gauget if far i= 1, ..., size(s)-1.

concat - proc (s1, s2: sequence(ii) returns (r: securitized) elements (finite) effects Returns the constationation of s1 and all which is a security the elements of s1 tollowed by the elements of s2. That is, develop for / an index of s1 and false(s1)+(-align for / an index of s2. That is made if the resulting sequence would be Apiza(st)-A-all) for / as to larger than the traplamantat

subseq = proc (s: sequence(i, st, cat: ini) secures (sining) eligentic formatic, respective, start

effects If and < 0, day I at a 1 or at > alcolul+1, signals bounds. the stands date date it ... in that order: the Otherwise setures a s But I min(ant, also-al+1) -New sequence contains and 0, autore citure the empty of

size = prod (s: experimentity returns (int) effects Returns the Runteer of elements in a

- emply print (s: segmental); relarite (bont) ellecte Relarite true il s containe ne elements; ellecture relarite false.
- felch prop (s: companying), i: hill publicate Al adapted a state of the second ante the Ah element of a

bottom - proc (a: engranning), internatio (a) standing for effects if a is empty, signate breakly. Consume in 11. 60

top = proc (s: enquerent) (stands & dependent) clients II s is empty agends councils. Silverse . In such as a state of the

elements - Rer (s: experiment) yields (t) ellects Visits the elements of a in order (i.e., e(1), e(2), ...).

indexes - Her (a: cogninably yields (kit) allede Vields the indiana of allow 1 to altra(a).

s2s - proc (a: altrug@) initiates (annumed) climite finitume a sequence hardeg if elements of a in the same order as in a.

s2a - proc (s: anguanting) entering (array)(b) effects Platents a new array with few based 1 and having the elements of s in the same order as in s.

equal - pres (s1, s2: engrennent) where the state of the sectory)

ndraine / hans an the effect of and the second make an an andre stategyd a

لده فصل

end:

POLUM (LINE)

-

similar - prec (s1, s2: entrance(D) antegra (book) (double (bollow)) Rogense (frances) Antegra (frances) Antegra (frances) Vierte in the contemport (contemport) Similar

transmit = proc (s: sequence[t]) returns (sequence[t]) signals (failure(string)) requires t has transmit

effects Returns a sequence having as elements transmitted copies of the elements of s in the same order. Sharing among elements is preserved. Signals failure if this cannot be represented on the receiving end and also resignals any failures from Stransmit.

II.9. Arrays

array = data type [t: type] is create, new, predict, fill, fill_copy, addh, addi, remh, remi, set_low, trim, store, fetch, bottom, top, empty, size, low, high, elements, indexes, equal, similar1, copy, copy1, transmit

Overview

Arrays are mutable objects that represent tuples of elements of type t that can grow and shrink dynamically. Each array's state consists of this tuple of elements and a low bound (or index). The elements are indexed sequentially, starting from the low bound. Each array also has an identity as an object.

Arrays can be created by calling array operations *create, new, fill, fill_copy,* and *predict.* They can also be created by means of the array constructor, which specifies the array low bound, and an arbitrary number of initial elements, see Section 6.2.9.

Operations low, high, and size return the current low and high bounds and size of the array. For array a, size(a) is the number of elements in a, which is zero if a is empty. These are related by the equation: high(a) = low(a) + size(a) - 1.

For any index *i* between the low and high bound of an array, there is a defined element, *a*[*i*]. The bounds exception is raised when an attempt is made to access an element outside the defined range. Any array must have a low bound, a high bound, and a size which are all legal integers. An implementation may restrict these to some smaller range of integers. A call that would lead to an array whose low or high bound or size is outside the defined range terminates with a *limits* exception.

Operations

create = proc (ib: int) returns (array[t]) signals (limits)

effects Returns a new, empty array with low bound *lb. Limits* occurs if the resulting array would not be supported by the implementation.

new = proc () returns (array[t]) effects Returns a new, empty array with low bound 1. Equivalent to create(1).

predict = proc (ib, cnt: http://www.com

The shashute value of out is a state of the same series. If our in Ra to sto n of these status a > 0. adulta ana a NERSON'S CONSIDER A SERVICE a de ange sud COLOR OF COLOR

fill - proc (ib, cnt: int, einer () mit

sector bound it and alco official I and a C formation 315-A-1

fill_copy = proc (ib, cut; ind, chunk & second damage

a de la companya de la companya de la companya de la companya de la companya de la companya de la companya de l e Angen - Herrie Angemer, int 11. EB

า **นั้งสุดต้าน**การสุการการสุการการสุด ala an an San State the state of the state . and and the second second second second second second second second second second second second second second s أتدخر تتله

addh - proc (a: annytt, class: & algebra dhaha

10.2

is if extending a on the Mats and answer the Mats lawed or size of a to be autoide the noise extends a by 1 in the physical jet] - size. in the staff states

add - prot (s: armyll, done) algodia (initia)

3 900

nd or also of a to be eviate the Survey and the sky 1 in the tow day

remh - proc (a: array(2) valuene () algunde fanades

and a high shares and the numerical shares and .) - 1.

remi - prez (a: emplo manna di ciando disent

34

If a is enough, algoride instruct. Otherschup shifts a star connecting its low element and many the instruction descent many the instruction of the low element and

set_low - pape (a: anappi), is: but aligned (index)

.

alles the low and black hours of a day particulation of a lo do and the new

```
store - proc (a: array[t], i: int, elem: t) signals (bounds)
      modifies a.
      effects if i < low(a) or i > high(a), signals bounds; otherwise makes elem the element of a
         with index i
fetch = proc (a: array[t], i: int) returns (t) signals (bounds)
      effects if i < low(a) or i > high(a), signals bounds; otherwise returns the element of a with
          index i.
bottom = proc (a: arraviti) returns (t) signals (bounds)
      effects if a is empty, signals bounds; otherwise returns a low(a)].
top = proc (a: array[t]) returns (t) signals (bounds)
      effects if a is empty, signals bounds; otherwise returns a high(a)].
empty = proc (a: arrav(t)) returns (beoi)
      effects Returns true if a contains no elements; otherwise returns false.
size = proc (a: array[t]) returns (int)
      effects Returns a count of the number of elements of a.
low = proc (a: arraviti) returns (int)
      effects Returns the low bound of a.
high = proc (a: arraviti) returns (int)
      effects Returns the high bound of a.
elements = iter (a: array[t]) yields (t) signals (failure(string))
      effects Yields the elements of a, exactly once for each index, from the low bound to the high
          bound (i.e., bottom(apre), ..., top(apre)). The elements are fetched one at a time, using
          the indexes that were legal at the start of the call. If, during the iteration, a is modified so
          that fetching at a previously legal index signals bounds, then the iterator signals failure
          with the string "bounds". The iterator is divisible at vields.
indexes = iter (a: array[t]) yields (int)
      effects Yields the indexes of a from the low bound of a_{pre} to the high bound of a_{pre}. Note
          that indexes is unaffected by any modifications done by the loop body. It is divisible at
          vields.
equal = proc (a1, a2: arraviti) returns (bool)
       effects Returns true if a1 and a2 refer to the same array object; otherwise returns false.
similar = proc (a1, a2: array[t]) returns (bool) signals (failure(string))
       requires t has similar: proctype (t, t) returns (bool) signals (failure(string))
       effects Returns true if a1 and a2 have the same low and high bounds and if their elements
          are pairwise similar as determined by Ssimilar. This effect of this operation is equivalent
          to the following procedure body (except that this operation is only divisible at calls to
          f$similar\:
                  at = array[t]
                  if at$iow(a1) ~= at$iow(a2) cor at$size(a1) ~= at$size(a2)
                       then return (faise)
                        and
                  for i: int in at$indexes(a1) do
                        If ~t$similar(a1[i], a2[i]) then return (false) end
                           regional failure
                           except when bounds: signal failure("bounds") end
                        end
                  return (true)
```

similar1 = proc (a1, a2: array[t]) returns (bool) signals (failure(string))

requires t has equal: proctype (t, t) returns (bool) signals (failure(string))

effects Returns true if a1 and a2 have the same low and high bounds and if their elements are pairwise equal as determined by Sequel. This operation works the same way as similar, except that Sequel is used instead of Seimilar.

Ale and a straight of the

copy = proc (a: array[t]) returns (b: array[t]) signals (failure(string))

requires t has copy: proctype (I) returns (I) signals (failure(string))

effects Returns a new array b with the same low and high bounds as a and such that each element b[i] contains \$copy(a[i]). The effect of this operation is equivalent to the following body (except that it is only divisible at calls to \$copy):

b: array[t] := array[t]\$copy1(a)

for i: int in array[1]\$indexes(a) do

b[i] := t\$copy(a[i])

resignal failure

except when bounds: signal failure("bounds") end

end

return (b)

copy1 = proc (a: array[t]) returns (b: array[t])

effects Returns a new array b with the same low and high bounds as a and such that each element b[l] contains the same element as a[l].

transmit = proc (a: array[t]) returns (b: array[t]) signals (failure(string))

requires t has transmit

effects Returns a new array b with the same low and high bounds as a and such that each element b[i] contains a transmitted copy of a[i]. Sharing among the elements of a is preserved in b. Signals failure if b cannot be represented on the receiving end or if fetching an element at a legal index of a_{pre} causes a bounds exception and resignals any failure signals raised by **stransmit**.

II.10. Atomic Arrays

atomic_array = data type [t: type] is create, new, predict, fill, fill_copy, addh, addl, remh, remi,

set_low, trim, store, fetch, bottom, top, empty, size, low, high, elements, indexes,

aa2a, a2aa, equal, similar, similar1, copy, copy1, tranemit,

test and read, test and write, can read, can write, read lock, write lock

Overview

Atomic_arrays are mutable atomic objects that represent tuples of elements of type *t* that can grow and shrink dynamically. Each atomic_array's (sequential) state consists of this tuple of elements and a low bound (or index). The elements are indexed sequentially, starting from the low bound. Each atomic_array also has an identity as an object.

Atomic_arrays can be created by calling atomic_array operations create, new, fill, fill_copy, and predict. They can also be created by means of the atomic_array constructor, which specifies the array low bound, and an arbitrary number of initial elements, see Section 6.2.9.

Operations low, high, and size return the current low and high bounds and size of the atomic_array. For an atomic_array a, size(a) is the number of elements in a, which is zero if a is empty. These are related by the equation: high(a) = low(a) + size(a) - 1.

For any index *i* between the low and high bound of an atomic_array, there is a defined element, *a[i]*. The bounds exception is raised when an attempt is made to access an element outside the defined range. Any atomic_array must have a low bound, a high bound, and a size which are all legal integers. An implementation may restrict these to some smaller range of integers. A call that would lead to an atomic_array whose low or high bound or size is outside the defined range terminates with a *limits* exception. *limits* exception.

and the second states and the

同时动物的 出生产行行的 建物

Atomic_arrays use read/write locking to achieve atomicity. The locking rules are described in Section 2.2.2. It is an error if a process that is not in an action attempts to test or obtain a lock; when this happens the guardian running the process will crash. As defined below, the only operation that (in the normal case) does not attempt to test or obtain a lock is the equal operation.

Operations

create = proc (ib: int) returns (a:stomic_array[t]) signals (limits)

effects Returns a new, empty atomic_array a with low bound lb. Limits occurs if the resulting atomic_array would not be supported by the implementation. The caller obtains a read lock on a.

new = proc () returns (atomic_array[i]) effects Equivalent to create(1).

predict = proc (ib, cnt: int) returns (a: stomic_array[i]) signals (limits)

effects Returns a new, empty atomic array a with low bound *lb*. The caller obtains a read lock on *a*. This is essentially the same as create(*ib*), except that the absolute value of *cnt* is a prediction of how many addits or addits are likely to be performed on this new atomic_array. If *cnt* > 0, addits are expected; otherwise addits are expected. These operations may execute faster than if the atomic_array had been produced by calling *create*. *Limits* occurs if the resulting atomic_array would not be supported by the implementation because of its initial low bound (not because of its predicted size or because of the predicted high or low bound).

fill = proc (lb, cnt: int, elem: t) returns (atomic_array[t]) signals (negative_size, limits)

effects if cnt < 0, signals negative_size. Returns a new atomic_array with low bound /b and size cnt, and with elem as each element; if this new atomic_array would not be supported by the implementation, signals limits. The caller obtains a read lock on the result.

fill copy = proc (lb, cnt: int, elem: t) returns (atomic array[t])

signale (negative size, limits, failure(string))

requires t has copy: proctype (t) returns (t) signals (tailure(string))

effects The effect is like fill except that elem is copied ant times. If ant < 0, signals negative_size. Normally returns a new array with low bound ib and size ant and with each element a copy of elem, as produced by ficapy. The caller obtains a read lock on the result. Any failure signal raised by ficapy is immediately resignated. This operation does not originate any failure signals by itself. If the new array cannot be represented by the implementation, signals limits.

addh = proc (a: atomic_array[t], elem: t) signals (ilmits) modifies a.

effects Obtains a write lock on *a*. If extending *a* on the high end would cause the high bound or size of *a* to be outside the range supported by the implementation, then signals *limits*. Otherwise extends *a* by 1 in the high direction, and stores *elem* as the new element. That is, a_{real} *high* $(a_{real})+1$ = *elem*.
A STATE AND AND A STATE AND A STATE AND A STATE AND A STATE AND A STATE AND A STATE AND A STATE AND A STATE AND

addi = proc (a: atomic arrayit), elem: t) signals (limits)

modifies a.

effects Obtains a write lock on a. If extending a on the low end would causes the low bound or size of a to be outside the range supported by the implementation, then signals limits. Otherwise extends a by 1 in the low direction, and stores elem as the new element. That is, $a_{\text{num}}[low(a_{\text{num}})-1] = olem.$

remh = proc (a: atomic arraviti) returns (t) algebra (bounds)

modifies a.

- effects Obtains a write lock on a. If a is empty, signals bounds. Otherwise shrinks a by removing its high element, and returns the removed element. That is, high(a_{non}) = high(a____) - 1.
- remi = proc (a: atomic array(t)) returns (t) signals (bounds)

modifies a.

- effects Obtains a write lock on a. If a is empty, signals bounds. Otherwise stwinks a by removing its low element, and returns the removed element. That is, $low(a_{max}) =$ $low(a_{nm}) + 1.$
- set low = proc (a: atomic array[t], b: int) signals (limits)

modifies a.

effects Obtains a write lock on a. If the new low (or high) bound would not be supported by the implementation, then signals limits, Otherwise, modilies the low and high bounds of a; the new low bound of a is 10 and the new high bound is $high(a_{max}) =$ high(ann)+10-10w(ann).

trim = proc (a: atomic array(t), ib, cnt: int) signals (negative size, bounds)

modifies a.

- effects if ant < 0, signals negative size and does not obtain any locks. Otherwise obtains a write lock on a. If b < low(a) or b > high(a)+1, signals bounds. Otherwise, modilies a by removing all elements with index < b or greater than or equal to b+cnt; the new low bound is b. For example, if a = atomic emerginities array[int]0[1,2,3,4,5], then: trim(a, 2, 2) results in a having value atomic energy[int]0[2:2, 3]
 - trimita, 2, 2) results in a naving value stemic_array[int]\$[4: 4, 5] trim(a, 4, 3) results in a having value stemic_array[int]\$[4: 4, 5]

store = proc (a: atomic array[1], i: int, elem: t) signale (bounds)

modifies a.

- effects Obtains a write lock on a. If i < low(a) or i > high(a), signals bounds; otherwise makes alem the element of a with index /.
- fetch = proc (a: atomic array(t), i: int) returne (t) signale (bounds) effects if / < low(a) or i > high(a), signals bounds; otherwise returns the element of a with

index /. Always obtains a read lock on a.

bottom = proc (a: atomic_array[t]) returns (t) signals (bounds) effects if a is empty, signals bounds; otherwise returns allow(a)]. Always obtains a read lock on a.

- top = proc (a: atomic array[t]) returns (t) signals (bounds) effects if a is empty, signals bounds; otherwise returns a[high(a)]. Always obtains a read lock on a.
- empty = proc (a: atomic array[1]) returns (bool) effects Returns true if a contains no elements, returns false otherwise. In either case obtains a read lock on a.

size = proc (a: atomic arrayf() returns (int) effects Returns a count of the number of elements of a obtains a read lock on a. low = proc (a: atomic_array[t]) returns (int)

effects Returns the low bound of a, obtains a read lock on a

high = proc (a: atomic_array[t]) returns (int)

effects Returns the high bound of a, obtains a read lock on a.

- elements = iter (a: atomic_array[t]) yields (t) signals (failure(string))
 - effects Obtains a read lock on a and yields the elements of a, each exactly once for each index, from the low bound to the high bound (i.e., bottom(apre), ..., top(apre)). The elements are fetched one at a time, using the indexes that were legal at the start of the call. If, during the iteration, a is modified so that fetching at a previously legal index signals bounds, then the iterator signals failure with the string "bounds". The iterator is divisible at yields.
- indexes = iter (a: atomic_array[t]) yields (int)

effects Obtains a read lock on a, then yields the indexes of a from the low bound of apre to the high bound of apre. Note that *indexes* is unaffected by any modifications done by the loop body. It is divisible at yields.

- aa2a = proc (aa: atomic_array[t]) returns (array[t]) effects Obtains a read lock on as and returns an array a with the same (sequential) state.
- a2aa = proc (array[t]) returns (aa: atomic_array[t]) effects Returns an atomic array as with the same state as a. Obtains a read lock on as.
- equal = proc (a1, a2: atomic_array[t]) returns (bool) effects Returns true if a1 and a2 refer to the same atomic_array object; otherwise returns false. No locks are obtained.
- similar = proc (a1, a2: atomic_array[t]) returns (bool) signals (failure(string))
 - requires t has similar: proctype (t, t) returne (bool) signals (failure(string))
 - effects Returns true if a1 and a2 have the same low and high bounds and if their elements are pairwise similar as determined by **Baimilar**. See the description of the *aimilar* operation of array for an equivalent body of code. This operation is divisible at calls to *\$\$similar*. Read locks are obtained on a1 and a2, in that order.

similar1 = proc (a1, a2: atomic_array[t]) returns (bool) signals (failure(string))

requires t has equal: proctype (t, t) returns (beel) signals (failure(string))

effects Returns true if a1 and a2 have the same low and high bounds and if their elements are pairwise equal as determined by (Sequal. This operation works the same way as similar, except that (Sequal is used instead of (Seimilar. Read locks are obtained on a1 and a2, in that order.

copy = proc (a: atomic_array[t]) returns (b: atomic_array[t]) signals (failure(string))

requires t has copy: proctype (1) returns (1) signals (failure(string))

effects Returns a new atomic array b with the same low and high bounds as a and such that each element b[i] contains #copy(a[i]). See the description of the copy operation of array for an equivalent body of code. This operation is divisible at calls to #copy, and obtains read locks on a and b.

copy1 = proc (a: atomic_array[t]) returns (b: atomic_array[t])

effects Returns a new atomic_array b with the same low and high bounds as a and such that each element b[4] contains the same element as a[4]. Read locks are obtained on a and b.

transmit = proc (a: atomic_array[t]) returns (b: atomic_array[t]) signals (failure(string))
requires t has transmit

effects Returns a new array b with the same low and high bounds as a and such that each element b[i] contains a transmitted copy of a[i]. Read locks are obtained on a and b. Sharing among the elements of a is preserved in b. Signals failure if b cannot be represented on the receiving end or if fetching an element at a legal index of apre causes a bounds exception and resignals any failure signals raised by Stransmit.

医白色 医静脉炎 化氟基酸酸酯医稀加丁酸医香油 化合金

test_and_read = proc (aa: atomic_array[t]) returns (bool)

- effects Tries to obtain a read lock on aa. If the lock is obtained, returns true; otherwise no lock is obtained and the operation returns false. The operation does not "wait" for a lock. Even if the executing action "knows" that a lock could be obtained, false may be returned. Even if false is returned, a subsequent attempt to obtain a read lock might succeed without waiting.
- test and write proc (aa: stomic_array[t]) returns (bool)
 - effects Tries to obtain a write lock on aa. If the lock is obtained, returns true; otherwise no lock is obtained and the operation returns false. The operation does not "wait" for a lock. Even if the executing action "knows" that a lock could be obtained, false may be returned. Even if false is returned, a subsequent attempt to obtain a write lock might succeed without waiting.
- can read = proc (aa: atomic array[t]) returns (bool)
 - effects Returns true if a read lock could be obtained on *aa* without waiting, otherwise returns false. No lock is actually obtained. Even if the executing action "knows" that a lock could be obtained, false may be returned. Since some concurrent action may obtain or release a lock on an atomic_array at any time, the information returned is unreliable: even if true is returned, a subsequent attempt to obtain the lock may require waiting; and even if false is returned, a subsequent attempt to obtain a read lock might succeed without waiting.

can_write = proc (aa: atomic_array[t]) returns (bool)

- effects Returns true if a write lock could be obtained on *aa* without waiting, otherwise returns false. No lock is actually obtained. Even if the executing action "knows" that a lock could be obtained, false may be returned. Since some concurrent action may obtain or release a lock on an atomic_array at any time, the information returned is unreliable: even if true is returned, a subsequent attempt to obtain the lock may require waiting; and even if false is returned, a subsequent attempt to obtain a write lock might succeed without waiting.
- read_lock = proc (aa: atomic_array[t]) effects Obtains a read lock on *aa*.
- write_lock = proc (aa: atomic_array[t]) effects Obtains a write lock on aa.

IL11. Structs

 $\mathbf{t} = \text{data type}\left\{\mathbf{r}_{i}; t_{1}, ..., \mathbf{r}_{k}; t_{k}\right\} \text{is replace } \mathbf{r}_{i}$ r, 181 uqual, civilar, acqu, i

A struct (sh	the "winethern")				tinte. The
					r
where					
		Raine Caracter and			
	and an an and a second	ristadouges, outras esté mentionales anno services		and the second second second second second second second second second second second second second second second	and the second second
type:					
A struct is an					
For purposed					
	and a second second second second second second second second second second second second second second second		al de la compara de color de la competencia de la competencia de la competencia de la competencia de la compete Competencia de la competencia de la comp		
Much as all					finition, a
in the follows					
	알, 제 - 10 1881) 1997년 - 1997년 - 1997년 - 1997년 -				
	a ang ang ang ang ang ang ang ang ang an				A Records
-					
- Andrewski - Andr	e danger Handse dan s		in the second second second		

1.475 41 -36 - 1 Ref. 1 Tolkeday - ----and a second second second second second second second second second second second second second second second **and head of the second s**

ate des. 31.4 1 1 1 . Étaise des a**stan**t elenera di s and the second states in the s

similar = proc (s1, s2: st) returns (bool) signals (failure(string))

requires each t_i has similar: proctype (t_i, t_i) returns (booi) signais (failure(string))

and the second for the second second

effects Returns true if s1 and s2 contain similar objects for each component as determined by the *t\$similar* operations. Any *failure* signal is immediately resignalled. This operation does not itself originate any *failure* signal. The comparison is done in lexicographic order of the selectors; if any comparison returns **false**, **false** is returned immediately.

copy = proc (s: st) returns (st) signals (failure(string))

requires each t, has copy: proctype (t,) returns (t,) signals (failure(string))

effects Returns a struct containing a copy of each component of s; copies are obtained by calling the *tfscopy* operations. Any *failure* signal is immediately resignalled. This operation does not itself originate any *failure* signal. Copying is done in lexicographic order of the selectors.

transmit = proc (s: st) returns (st) signals (failure(string))

requires each ti has transmit

effects Returns a struct containing a transmitted copy of each component of s. Sharing is preserved among the components of s. Any failure signal from (\$transmit is immediately resignated. This operation does not itself originate any failure signal.

II.12. Records

record = data type [n₁: t₁, ..., n_k: t_k] is r_gets_r, r_gets_s, set_n₁, ..., set_n_k, get_n₁, ..., get_n_k, equal,similar, similar1, copy, copy1, transmit

Overview

A record is a mutable collection of one or more named objects. The names are called *selectors*, and the objects are called *components*. Different components may have different types. A record also has an identity as an object.

An instantiation of record has the form:

record [field spec, ...]

where

field spec ::= name, ... : type actual

(see Appendix I). Selectors must be unique within an instantiation (Ignoring capitalization), but the ordering and grouping of selectors is unimportant. For example, the following name the same type:

record[last, first, middle: string, age: int] record[last: string, age: int, first, middle: string]

A record is created using a record constructor, see Section 6.2.11.

For purposes of the certain operations, the the names of the selectors are ordered lexicographically. Lexicographic ordering of the selectors is the alphabetic ordering of the selector names written in lower case (based on the ASCII ordering of characters).

In the following definitions of record operations, let $\mathbf{rt} = \mathbf{record}[\mathbf{n}_1; \mathbf{t}_1, ..., \mathbf{n}_k; \mathbf{t}_k]$.

Operations

r_gets_r = proc (r1, r2: rt)

modifies r1.

effects Sets each component of r1 to be the corresponding component of r2.

r gets s = proc (r: rt, s: st)

modil 10 1.

effects Here of is a struct type whose components have the same selectors and types as /t. Sets each component of r to be the contemporative contemporat of a.

set_n; = proc (r: n; e: t;)

100 /.

effects tildilles r by making the component whose selector is n be e. There is a set_ operation for each selector.

get n = proc (r: rt) returne (t.)

effects Returns the component of r whose selector is n. There is a get operation for each selector.

equal = proc (r1, r2; rt) returns (boot)

offecto Pieturne taxe # r1 and r2 are the same record object; otherwise returns false.

similar - proc (ct, r2: rt) sultante (bealt aligned Same services and the services of

no pach (has die (grinte)ended)) ate Rotore true I it and it The second an share was street as the second nit as determin by the Galatie general sen else reasons de And the second second does not lead on ic enter ant was de of the palestary. I gay came

similar1 - proc (r1, r2: r1) returns (and allo

requires each (has equit (e daibere (abrine)) and a state of a state of a in Rinkuma (man il // and de nd by 2000

close not load and Sector Second dana dia man kingala d of the actoriant. Carst and

copy = proc (r: it) returns (it) at anti-

regulates cash (dans and gang Maria Patisma di ang Ang with and than replacing each Ś. szagista i sz A Sugar de nd by ord with a cap GOMBON The 1.00 5 calling the disease equation constant of the second and shows and to dome in Indecarachic order of the seland

copy1 - pres (r: it) returns (rt)

effects Returns a new record containing the components of r as its components.

tranemit - proc (r: n) netwine (n) elevate (tellere(entreg))

te and the bas

And the second s teren industria partition the second in the second

II.13. Atomic Records

atomic_record = data type [n₁ : t₁, ..., n_k: t_k] is ar_gets_ar, set_n₁, ..., set_n_k, get_n₁, ..., get_n_k, ar2r, r2ar, equal,similar, similar1, copy, copy1, transmit,

test_and_read, test_and_write, can_read, can_write, read_lock, write_lock

Overview

An atomic_record is a mutable atomic collection of one or more named objects. The names are called *selectors*, and the objects are called *components*. Different components may have different types. An atomic_record also has an identity as an object.

An instantiation of atomic_record has the form:

atomic_record [field_spec , ...]

where

field_spec ::= name, ... : type_spec

(see Appendix I). Selectors must be unique within an instantiation (ignoring capitalization), but the ordering and grouping of selectors is unimportant. For example, the following name the same type:

atomic_record[iast, first, middle: string, age: int]
atomic_record[last: string, age: int, first, middle: string]

An atomic_record is created using a atomic_record constructor, see Section 6.2.11.

For purposes of the certain operations, the the names of the selectors are ordered lexicographically. Lexicographic ordering of the selectors is the alphabetic ordering of the selector names written in lower case (based on the ASCII ordering of characters).

Atomic_records use read/write locking to achieve atomicity. The locking rules are described in Section 2.2.2. It is an error if a process that is not in an action attempts to test or obtain a lock; when this happens the guardian running the process will crash. As defined below, the only operation that (in the normal case) does not attempt to test or obtain a lock is the equal operation.

In the following, let art = atomic_record[n1: t1, ..., nk: tk].

Operations

ar_gets_ar = proc (r1, r2: art)

modifies r1.

effects Obtains a write lock on r1 and a read lock on r2, then sets each component of r1 to be the corresponding component of r2.

 $get_{n_i} = proc (r: art) returns (t_i)$

effects Obtains a read lock on r and returns the component of r whose selector is n_j . There is a get operation for each selector.

set_ $n_i = proc$ (r: art, e: t_i)

modifies r.

effects Obtains a write lock on *r* and modifies *r* by making the component whose selector is *n_i* be *e*. There is a *set_* operation for each selector.

ar2r = proc (ar: art) returns (r: art)

effects Obtains a read lock on ar and returns a record r with the same state.

r2ar = proc (r: art) returns (ar: art)

effects returns an atomic record ar with the same state as r. Obtains a read lock on ar.

equal = proc (r1, r2: art) returns (bool)

effects Returns true if r1 and r2 are the very same atomic_record object; otherwise returns false. No locks are obtained.

similar = proc (r1, r2: art) returne (bool) signals (failure(string))

- requires each t_i has similar: proctype (t_i, t_i) returns (bool) signals (failure(string)) effects Obtains a read lock on r1, then a read lock on r2; then compares corresponding
 - components from r1 and r2 using the tSeimilar operations. Any failure signal is immediately resignalled. This operation does not itself originate any failure signal. The comparison is done in lexicographic order of the selectors; if any comparison returns false, false is returned immediately. If all comparisons return true, returns true.
- similar1 = proc (r1, r2: art) returns (bool) signals (failure(string))
 - requires each t_i has equal: proctype (t_i, t_i) returns (bool) signals (failure(string))
 - effects This operation is the same as similar, except that t\$equal is used instead of t\$similar.
- copy = proc (r: art) returns (res: art) signals (failure(string))
 - requires each t, has copy: proctype (t,) returns (t,) signale (failure(string))
 - effects Obtains a read lock on *r*, then returns a new atomic record *res* obtained by performing *copy1(r)* and then replacing each component with a copy of the corresponding component of *r*. Copies are obtained by calling the *tScopy* operations. Any *failure* signal is immediately resignalled. This operation does not itself originate any *failure* signal. Copying is done in lexicographic order of the selectors. A read lock is also obtained on the new atomic record *res*.

copy1 = proc (r: art) returns (res: art)

effects Obtains a read lock on r, then returns a new atomic_record res containing the components of r as its components. A read lock is also obtained on the new atomic_record res.

transmit = proc (ar: art) returns (art) signals (failure(string))

requires each t, has transmit

effects Returns a new atomic_record containing a transmitted copy of each component of ar. Sharing is preserved among the components of ar. A read lock is obtained on ar and the new atomic_array. Any failure signal from t\$transmit is immediately resignalled. This operation does not itself originate any failure signal.

test_and_read = proc (ar: art) returns (bool)

effects Tries to obtain a read lock on *ar*. If the lock is obtained, returns true; otherwise no lock is obtained and the operation returns false. The operation does not "wait" for a lock. Even if the executing action "knows" that a lock could be obtained, false may be returned. Even if false is returned, a subsequent attempt to obtain a read lock might succeed without waiting.

test_and_write = proc (ar: art) returns (bool)

effects Tries to obtain a write lock on *ar.* If the lock is obtained, returns true; otherwise no lock is obtained and the operation returns false. The operation does not "wait" for a lock. Even if the executing action "knows" that a lock could be obtained, false may be returned. Even if false is returned, a subsequent attempt to obtain a write lock might succeed without waiting.

can_read -- proc (ar: art) returns (back) effects Returns the if a read lick could be obtained an arwithout waiting, otherwise returns article "lucume" that a lock tales. No lock is actually obtained. Even I the assouting action "knows" that a lock could be obtained, false any be ask nin or 1. 🗖 i e in allow obtain roleans a took on an applie net itie: Martin 100000 oven if true is estimated, a subsequent statement even if false is subsequent, a subsequent subsequent without waiting. e and had might succeed

can write - proc (ar: art) relains (bool)

te Pletane true il e unite lock could be obtained on ar utilheut walling, otherwise returns a haan dagaa da haan ga Agaa dagaa sagaa sagaa Agaa daga sagaa sagaa sa ي ا 1 in. No lock to a inf that a lock and the second s could be alterned. releases a lock on up disails, des and deress is universe even il true le retain k and beecous thank i without we

read lock - pres (ar: art) effects Obtains a read lock on ar.

write lock - proc (ar: art) effects Obtains a write lock on ar.

II.14. Oneofe

 $one of = data type(n_1; t_1, ..., n_k; t_1) to make n_1, ..., make n_k, is n_1, ..., is n_k, value n_1, ..., value n_k, and a state n_k, and$ ozv. vzo. equal, similar, capy, wanamit

Overview

A oneof is a tagged, deprivated union; that is, a labeled shipst, to be thought of as "one of" a set of alternatives. The label is called the taggest and the algost is called the value (or date part). set of alternatives. The label is called the taggent, and

An instantiation of enget has the form:

creat (lisid_spec ,]

where (as for records)

field apec 228 mme, ... : type_actual (see Appendix 1). Tage must be unique within an instantiation (ignoring capitalization), but the ordering and grouping of tags is uningestight.

Although there are ensel operations for decomposing ensel objects, they are usually decomposed via the tagence statement, which is document in floating 10.14.

A one of is immutable but may contain a mutable chief; therefore, a energy is atomic only if all of the types of its data parts are alemic.

in the following, let at = anealing: type, n.: 1.].

Operations

make n = proc (e: 1) reliance (ot)

effects Returns a oneof object with tag //, and value #. There is a make operation for each selector.

is n = proc (o: ot) returne (bool)

effects Returns true if the tag of o is n, also returns false. There is an is, operation for each salastor.

value_n; = proc (o: ot) returns (t;) signals (wrong_tag)

effects if the tag of o is n_i , returns the value of o; otherwise signals wrong_tag. There is a value_operation for each selector.

o2v = proc (o: ot) returns (vt)

effects Here vt is a variant type with the same selectors and types as of. Returns a new variant object with the same tag and value as o.

v2o = proc (v: vt) returns (ot)

effects Here vt is a variant type with the same selectors and types as ot. Returns a one of object with the same tag and value as v.

equal = proc (o1, o2: ot) returns (bool) signals (failure(string))

requires each t, has equal: proctype (t, t) returns (bool) signals (failure(string))

effects Returns true if o1 and o2 have the same tag and equal values as determined by the equal operation of their data part's type. Any failure signal is immediately resignalled. This operation does not itself originate any failure signal. This operation is divisible at the call of t\$equal.

similar = proc (o1, o2: ot) returns (bool) signals (failure(string))

requires each t, has similar: proctype (t, t) returns (bool) signals (failure(string))

effects Returns true if 01 and 02 have the same tag and similar values as determined by the *similar* operation of their value's type. Any failure signal is immediately resignalled. This operation does not itself originate any failure signal. This operation is divisible at the call of t\$similar.

copy = proc (o: ot) returns (ot) signals (failure(string))

requires each t_i has copy: proctype (t_i) returns (t_i) signals (failure(string))

effects Returns a oneof object with the same tag as *o* and containing as a value a copy of *o*'s value; the copy is made using the copy operation of the value's type. Any failure signal is immediately resignalled. This operation does not itself originate any failure signal. This operation is divisible at the call of *t*\$copy.

transmit = proc (o: ot) returns (ot) signals (failure(string))

requires each t; has transmit

effects Returns a oneof object with the same tag as *o* and containing as a value a transmitted copy of *o*'s value. Any *failure* signal is immediately resignalled. This operation does not itself originate any *failure* signal.

II.15. Variants

variant = data type [n₁: t₁, ..., n_k: t_k] is make_n₁, ..., make_n_k, change_n₁, ..., change_n_k, is_n₁, ..., is_n_k, value_n₁, ..., value_n_k, v_gets_v, v_gets_o, equal, similar, similar1, copy, copy1, transmit

Overview

A variant is a mutable, tagged, discriminated union. Its state is a oneof, that is, a labeled object, to be thought of as "one of" a set of alternatives. The label is called the *tag part*, and the object is called the *value* (or data part). A variant also has an identity as an object.

An instantiation of variant has the form:

variant [field_spec , ...]

where

field_spec ::= name, ... : type actual

(see Appendix I). Tags must be unique within an instantiation (ignoring capitalization), but the ordering and grouping of tags is unimportant.

Although there are variant operations for decomposing variant objects, they are usually decomposed via the tagcase statement, which is discussed in Section 10.14.

In the following let $vt = variant[n_1: t_1, ..., n_k: t_k]$.

Operations

 $make_{n_i} = proc (e: t_i) returns (vt)$

effects Returns a new variant object with tag n_j and value e. There is a make_ operation for each selector.

change_ $n_i = proc$ (v: vt, e: t_i)

modifies v.

effects Modifies v to have tag n_i and value e. There is a change operation for each selector.

 $is_n = proc (v: vt) returns (bool)$

effects Returns true if the tag of v is n; otherwise returns false. There is an is_ operation for each selector.

value_n; = proc (v: vt) returns (t_i) signals (wrong_tag)

effects if the tag of v is n; returns the value of v; otherwise signals wrong_tag. There is a value operation for each selector.

v gets v = proc (v1, v2: vt)

modifies v1.

effects Modifies v1 to contain the same tag and value as v2.

v_gets_o = proc (v: vt, o: ot)

modifies v.

effects Here ot is the oneof type with the same selectors and types as vt. Modifies v to contain the same tag and value as o.

equal = proc (v1, v2: vt) returns (bool)

effects Returns true if v1 and v2 are the same variant object.

similar = proc (v1, v2: vt) returns (bool) signals (failure(string))

requires each t has similar: proctype (t, t) returns (boof) signals (failure(string))

effects Returns true if v1 and v2 have the same tag and similar values as determined by the similar operation of their value's type. Any failure signal is immediately resignalled. This operation does not itself originate any failure signal. This operation is divisible at the call of t\$similar.

similar1 = proc (v1, v2: vt) returns (bool) signals (failure(string))

requires each t_i has equal: proctype (t_i, t_i) returns (booi) signals (failure(string)) effects Same as similar, except that t\$equal is used instead of t\$similar.

copy = proc (v: vt) returns (vt) signals (failure(string))

requires each t_i has copy: proctype (t_i) returns (t_i) signals (failure(string))

effects Returns a variant object with the same tag as v and containing as a value a copy of v's value; the copy is made using the copy operation of the value's type. Any failure signal is immediately resignalled. This operation does not itself originate any failure signal. This operation is divisible at the call of *tScopy*.

copy1 = proc (v: vt) returns (vt)

effects Returns a new variant object with the same tag as v and containing v's value as its value.

transmit = proc (v: vt) returns (vt) signals (failure(string))

requires each t_i has transmit

effects Returns a variant object with the same tag as v and containing as a value a transmitted copy of v's value. Any *failure* signal is immediately resignalled. This operation does not itself originate any *failure* signal.

والأحلاق والمترجع والمراجع

II.16. Atomic Variants

atomic_variant = data type [n₁: t₁, ..., n_k: t_k] is make_n₁, ..., make_n_k, change_n₁, ..., change_n_k, av_gets_av, is_n₁, ..., is_n_k, value_n₁, ..., value_n_k, av2v, v2av, equal, similar, similar1, copy, copy1, transmit, test_and_read, test_and_write, can_read, can_write, read_lock, write_lock

Overview

An atomic_variant is a mutable, atomic, tagged, discriminated union. Its state is a oneof, that is, a labeled object, to be thought of as "one of" a set of alternatives. The label is called the *tag part*, and the object is called the *value* (or data part). An atomic_variant also has an identity as an object.

An instantiation of atomic variant has the form:

atomic_variant [field_spec , ...]

where

field_spec ::= name, ... : type_actual

(see Appendix I). Tags must be unique within an instantiation (ignoring capitalization), but the ordering and grouping of tags is unimportant.

Although there are atomic_variant operations for decomposing atomic_variant objects, they are usually decomposed via the tagtest statement or tagwalt statement, which are discussed in Section 10.15.

In the following, let avt = atomic_variant[n1: t1, ..., nk: tk].

Operations

make_ $n_i = proc$ (e: t_i) returns (av: avt)

effects Returns a new atomic_variant object av with tag n_i and value e. Obtains a read lock on av. There is a make_operation for each selector.

change_ $n_i = proc$ (v: avt, e: t_i)

modifies v.

effects Obtains a write lock on v, then modifies v to have tag n_i and value e. There is a change_operation for each selector.

av_gets_av = proc (v1, v2: avt)

modifies v1.

effects Obtains a read lock on v2 and then a write lock on v1, then modifies v1 to contain the same tag and value as v2.

is_n_i = proc (v: avt) returns (bool)

effects Obtains a read lock on v, then returns true if the tag of v is n; otherwise returns false. There is an *is* operation for each selector.

value_n; = proc (v: avt) returns (t;) signals (wrong_tag)

effects Obtains a read lock on v. Then, if the tag of v is n_j, returns the value of v; otherwise signals wrong_tag. There is a value_operation for each selector.

av2v = proc (av: avt) returns (v: vt)

effects Here vt is a variant type with the same selectors and types as avt. Obtains a read lock on av and returns a variant v with the same state.

v2av = proc (v: vt) returns (av: avt)

effects Here vt is a variant type with the same selectors and types as avt. Returns an atomic variant av with the same state as v. Obtains a read lock on av.

equal = proc (v1, v2: avt) returns (bool)

effects Returns true if v1 and v2 are the same atomic_variant object. No locks are obtained.

similar = proc (v1, v2: avt) returns (bool) signals (failure(string))

requires each t_i has similar: proctype (t_i, t_i) returns (bool) signals (failure(string))

effects Obtains read locks on v1 and v2, in order, and then compares the objects; returns true if v1 and v2 have the same tag and similar values as determined by the similar operation of their type. Any failure signal is immediately resignaled. This operation does not itself originate any failure signal. This operation is divisible at the call of t\$similar.

similar1 = proc (v1, v2: avt) returns (bool) signale (failure(string)) requires each t_i has equal: proctype (t_i, t_i) returns (bool) signals (failure(string)) effects Same as similar, except that tSequal is used instead of tSsimilar.

copy = proc (v: avt) returns (avt) signals (failure(string))

requires each t_i has copy: proctype (t_i) returns (t_i) signals (failure(string))

effects Obtains a read lock on v, then returns an atomic_variant object with the same tag as v and containing as a value a copy of v's value; the copy is made using the copy operation of the value's type. Any failure signal is immediately resignalled. This operation does not itself originate any failure signal. This operation is divisible at the call of t\$copy. A read lock is obtained on the result.

copy1 = proc (v: avt) returns (avt)

effects Obtains a read lock on v, then returns a new atomic_variant object with the same tag as v and containing v's value as its value. A read lock is obtained on the result.

transmit = proc (v: avt) returns (avt) signals (failure(string))

requires each t, has transmit

effects Returns an atomic_variant object with the same tag as v and containing as a value a transmitted copy of v's value. Obtains a read lock on v. Any failure signal is immediately resignalled. This operation does not itself originate any failure signal.

test_and_read = proc (av: avt) returns (bool)

effects Tries to obtain a read lock on *av.* If the lock is obtained, returns true; otherwise no lock is obtained and the operation returns false. The operation does not "wait" for a lock. Even if the executing action "knows" that a lock could be obtained, false may be returned. Even if false is returned, a subsequent attempt to obtain a read lock might succeed without waiting.

test_and_write = proc (av: avt) returns (bool)

effects Tries to obtain a write lock on av. If the lock is obtained, returns true; otherwise no lock is obtained and the operation returns false. The operation does not "wait" for a lock. Even if the executing action "knows" that a lock could be obtained, false may be returned. Even if false is returned, a subsequent attempt to obtain a write lock might succeed without waiting.

```
can read = proc (av: avt) returns (bool)
```

effects Returns true if a read lock could be obtained on av without waiting, otherwise returns false. No lock is actually obtained. Even if the executing action "knows" that a lock could be obtained, false may be returned. Since some concurrent action may obtain or release a lock on an atomic_variant at any time, the intermation returned is unreliable: even if true is returned, a subsequent attempt to obtain the lock may require waiting; and even if false is returned, a subsequent attempt to obtain a read lock might succeed without waiting.

and the second second second second second second second second second second second second second second second

can write - proc (av: avi) returns (bool)

effects Returns true if a write lock could be obtained on *av* without waiting, otherwise returns false. No lock is actually obtained. Even if the executing action "knows" that a lock could be obtained, false may be returned. Since some concurrent action may obtain or release a lock on an atomic variant at any time, the intermation returned is unreliable: even if true is returned, a subsequent attempt to obtain the lock may require waiting; and even if false is returned, a subsequent attempt to obtain a write lock might succeed without waiting.

read_lock = proc (av: avt) effects Obtains a read lock on av.

write_lock - proc (av: avt) effects Obtains a write lock on av.

II.17. Procedures and Iterators

proctype - data type is equal, similar, copy Itertype - data type is equal, similar, copy

Overview

Procedures and iterators are objects created by the Argus system. The type specification for a procedure or iterator contains most of the information stated in a procedure or iterator heading; a procedure type specification has the form:

proctype ([type_spec , ...]) [returns] [signals]

and an iterator type specification has the form:

itertype ([type_spec , ...]) [yields] [signals]

where

 returns
 ::=
 returns (type_spec , ...)

 yields
 ::=
 yields (type_spec , ...)

 signals
 ::=
 signale (exception , ...)

 exception
 ::=
 name [(type_spec , ...)]

(see Appendix I). The first list of type specifications describes the number, types, and order of arguments. The returns or yields clause gives the number, types, and order of the objects to be returned or yielded. The signals clause liels the exceptions mixed by the procedure or iterator; for each exception name, the number, types, and order of the objects to be returned are also given. All names used in a signals clause must be unique. The ordering of exceptions is not important. For example, both of the following type specifications name the procedure type for stringSaubstr:

proctype (string, int, int) returns (string) signals (bounds, negative_size) proctype (string, int, int) returns (string) signals (negative_size, bounds) Procedure and Iterator objects are created by compiling modules (and by the bind expression, see Section 9.8). Procedure and iterator types are not transmissible and are considered to be immutable and atomic in normal use. However, some uses of own data (see Section 12.7) in procedures and iterators can violate this assumption.

In the following operation descriptions, t stands for a proctype or itertype.

Operations

```
equal = proc (x, y: t) returns (bool)
similar = proc (x, y: t) returns (bool)
```

effects These operations return true if and only if x and y are the same implementation of the same abstraction, with the same parameters (see Section 12.6).

copy - proc (x: t) returns (t) effects Returns x.

II.18. Handlers and Creators

handlertype - data type is equal, similar, copy, transmit creatortype - data type is equal, similar, copy, transmit

Overview

Handlers and creators are created by the Argus system. The type specification for a handler or creator contains most of the information stated in a handler or creator heading; a handler type specification has the form:

handlertype ([type_spec, ...])[returns][signals]

and a creator type specification has the form:

creatortype ([type spec, ...]) [returns] [signals]

where

returns	::=	returns (type_spec ,)
signals	::=	signals (exception ,)

exception ::= name [(type_spec , ...)]

(see Appendix I). The first list of type specifications describes the number, types, and order of arguments. The returns clause gives the number, types, and order of the objects to be returned. The signals clause lists the exceptions raised by the handler or creator; for each exception name, the number, types, and order of the objects to be returned are also given. All names used in a signals clause must be unique; none can be unavailable or failure, which have a pre-defined meaning for remote calls (see Section 8.3). The ordering of exceptions is not important.

Creators are created by compiling modules, and handlers are created as a side-effect of guardian creation. Handlers and creators are transmissible and are considered to be immutable and atomic in normal use. Certain uses of own data in handlers can violate this assumption.

In the following operation descriptions, t stands for a handlertype or creatortype.

Operations

equal = proc (x, y: t) returns (bool) similar = proc (x, y: t) returns (bool)

effects These operations return true if and only if x and y are the same object (see Section 12.6 for an exact definition for the case of creators in guardian generators).

copy = proc (x: t) returns (t) transmit = proc (x: t) returns (t) effects Returns x.

II.19. Anys

any - data type is create, force, is_type

Overview

An object of type any contains a type T and an object of type T. Anys are immutable and are not transmissible. Anys are atomic only if their contained object is atomic.

Operations

```
create = proc[T: type] (contents: T) returns (any)
effects Returns an any object containing contents and the type T.
```

- force = proc[T: type] (thing: any) returns (T) signals (wrong_type) effects if thing contains an object of a type included in type 7, then that object is returned; otherwise wrong_type is signalled.
- is_type = proc[T: type] (thing: any) returns (bool) effects if thing contains an object of a type included in type 7, then true is returned;
 - otherwise, false is returned.

II.20. Images

image = data type is create, force, is_type, copy, transmit

Overview

An object of type **Image** is the value of an arbitrary transmissible type. See Section 14 for more details. Images are immutable, atomic, and transmissible.

Operations

create = proc[T: type] (contents: T) returns (image) signals (failurestring)

requires T has transmit

effects Returns an image object obtained from contents via the encode operation of T. Resignals any failure signal raised by T's encode operation.

force = proc[T: type] (thing: image) returns (T) signals (wrong_type, failure(string)) requires T has transmit

effects if thing encodes an object of a type included in type 7, then that object is extracted using the decode operation of 7 and returned. Otherwise wrong__type is signalled. Resignals any failure signal raised by 7's decode operation.

is_type = proc[T: type] (thing: Image) returns (bool)

requires T has transmit

effects if thing encodes an object of a type included in type 7, then true is returned; otherwise, false is returned.

copy = proc (thing: image) returns (image) transmit = proc (thing: image) returns (image) effects Returns thing.

II.21. Mutexes

mutex = data type[t: type] is create, set_value, get_value, changed, equal, similar, copy, transmit

Overview

A mutex is a mutable container for an object of type t. A mutex also has an identity as an object.

 $-i_{\pi} m$

An object of type mutex[t] provides *mutual* exclusion for process synchronization, and allows explicit control over how information contained in the mutex is written to stable storage (see Section 15.1).

The seize statement is used in order to gain possession of a mutex. See section 6.7.

Although mutex objects are mutable, sharing among mutex objects is usually wrong, because the contained object should only be accessible through the mutex. Hence there is no copy1 operation, since this would introduce sharing, and there is no *similar1* operation to check for sharing (see Section 6.7).

Operations

- create proc (thing: t) returns (mutex[t]) effects Returns a new mutex object containing thing.
- set_value = proc (container: mutex[t], contents: t)
 modifies container.
 effects Modifies container by replacing its contained object with contents.
- get_value = proc (container: mutex[t]) returns (t) effects Returns the object contained in container.
- changed = proc (container: mutex[t])

effects informs the Argus system that the calling action requires the contents of *container* to be copied to stable storage by the time the action commits, provided *container* is accessible from a stable variable. It is a programming error if a process that is not running an action calls this operations, and if this is done the guardian will crash.

equal = proc (m1, m2: mutex[t]) returns (bool)

effects Returns true if and only if m1 and m2 are the same object.

similar = proc (m1, m2: mutex[t]) returns (bool) signals (failure(string))

requires t has similar: proctype(t, t) returne(bool) algebras (failure(string))

effects Seizes m1, then seizes m2, and calls seimilar to determine its result; any failure signal is immediately resignalled. Possession of both mutexes is retained until similar terminates.

copy = proc (m1: mutex[t]) returns (m2: mutex[t]) signals (failure(string))

requires t has copy: proctype(t) returns(t) signals (failure(string))

effects Seizes m1, then calls is copy to make a copy which it places in the new mutex object m2. Any failure signal is immediately resignalled. Possession of m1 is retained until is copy terminates.

transmit - proc (m1: mutex[t]) returns (mutex[t]) signals (failure(string)) requires t has transmit

effects Seizes *m1*, and returns a new mutex containing a transmitted copy of the contained object. Any *failure* signal is immediately resignalled. Possession of *m1* is retained until istransmit terminates.

Appendix III Rules and Guidelines for Using Argus

This appendix collects the rules and guidelines that should be followed when programming in Argus. Following these rules makes selze statements meaningful, actions atomic, and so on. In some rare cases there may be valid reasons for violating these guidelines, but doing so greatly increases the difficulty of building, debugging, and running the resulting system.

All of the rules listed in this appendix are based on information appearing elsewhere in the manual. Each rule is followed by a brief rationale, including a reference to the section of the manual from which it is drawn.

III.1. Serializability and Actions

Actions should share only atomic objects.

Rationale: Actions that share non-atomic data are not necessarily serializable. [Section 2.2.2]

 A subaction that aborts should not return any information obtained from data shared with other concurrent actions.

Rationale: Returning such data may violate serializability. [Section 2.2.1]

A nested topaction should be serializable before its parent. This is true if either

- 1. the nested topaction performs a benevolent side effect (a change to the state of the representation that does not affect the abstract state), or
- 2. all communication between the nested topaction and its parent is through atomic objects.

Rationale: Other uses may violate serializability. [Section 2.2.3]

• The creation or destruction of a guardian must be synchronized with the use of that guardian via atomic objects such as the catalog.

Rationale: Otherwise serializability may be violated. [Section 10.18]

III.2. Actions and Exceptions

• If an exception raised by a call should not commit an action, the exception must be handled within that action.

Rationale: If an exception raised within an action body is handled outside the action, the implicit flow of control outside of the action will commit the action. [Section 11.5]

III.3. Stable Variables

• Stable variables should denote resilient data objects.

Rationale: Only data objects that are (reachable from the stable variables and) resilient are written to stable storage when a topaction commits. (This can be ensured by having stable variables only denote objects of an atomic type or objects protected by mutex.) Non-resilient objects stored in stable variables are only written to stable storage when the guardian is created. [Section 13.1]

- If a bound procedure or iterator will be accessible from a stable variable,
 - 1. the procedure or iterator being bound must be atomic and
 - 2. only atomic objects should be bound as arguments.

Rationale: The bound procedure or iterator may be stored in stable storage, and non-atomic data is only written to stable storage once. [Section 9.8]

III.4. Transmission and Transmissibility

• An abstract type's encode and decode operations should not cause side effects.

Rationale: The number of calls to an encode or decode operation is unpredictable, since arguments or results may be encoded and decoded several times as the system tries to establish communication. In addition, verifying the correctness of transmission is easier if encode and decode are simply transformations to and from the external representation. [Section 14.3]

• If the naming relation among objects to be transmitted is cyclic (e.g., a circular list) then encode and decode must be implemented in one of two ways:

- 1. The internal and external representation types must be identical, and encode and decode return their argument without modifying or accessing it, or
- The external representation object must be acyclic.

Rationale: A circular external representation may cause decode to fail. [Section 14.4]

 Objects that share other objects should be bound into a handler or creator in the same bind expression.

Rationale: Sharing is only preserved among objects bound at the same time. [Section 9.8]

III.5. Mutex

Mutual exclusion or atomic data should be used to synchronize access to all shared objects.

Rationale: In the presence of concurrency, any interleaving of indivisible events is possible. Without synchronization mechanisms, this concurrency will be visible to programs, significantly complicating coding and testing. [Section 8]

All modifications to mutex objects should be made inside seize statements.

Rationale: The system will gain possession of a mutex object before writing it to stable storage; thus, seizing a mutex in order to modify it will prevent the system from copying a mutex object when it is in an inconsistent state. This also prevents other processes from seeing inconsistent data [Section 15.2 and Section 15.1]

• Nested seizes should be avoided when pause is used, and pause must be avoided when nested seizes are used.

Rationale: A pause in a nested seize does not actually release possession of the mutex object. [Section 10.17]

 If an object is referred to by a mutex object, it should not be referred to by any other object, nor should it be denoted by a variable except when in possession of the containing mutex.

Rationale: If an object contained in a mutex can be reached by a method other than seizing the mutex, the mutual exclusion property of the mutex is undermined. [Section 6.7]

• No activity that is likely to take a long time should be performed while in a selze statement. In particular, programs should not make handler calls or wait for locks on atomic objects while in possession of a mutex.

Rationale: Waiting for a lock while in a mutex is likely to cause a deadlock with other actions or between the action holding the mutex and the Argus system. [Section 15.3]

Mutex objects should not share data with one another, unless the shared data is atomic or mutex.

Rationale: Sharing of non-atomic objects between mutex objects is not preserved when the mutexes are written to stable storage. [Section 15.3]

• Mutex[/]\$changed must be called after the last modification (on behalf of some action) to the contained object of a mutex.

Rationale: The Argus system is free to copy the mutex to stable storage as soon as mutex[#schanged has been called. Changes after the last call to mutex[#schanged but before topaction commit may not be written to stable storage. [Section 15.3]

• Mutex[/]\$changed should be called even if the mutex object changed is not accessible from the stable variables.

Rationale: In a scenario where the object was accessible, becomes inaccessible, then becomes accessible again, it is possible that stable storage would not be updated properly if this rule were not followed. The system guarantees that no problems with updating stable storage will arise if mutex[*i*]\$changed is always called after the last modification to the object. [Section 15.3]

• An atomic type implemented with a representation consisting of several mutex objects should use separate topactions to ensure that the mutexes are written to stable storage in an order that preserves the correctness of the representation.

Rationale: Mutexes are written to stable storage incrementally. Sometimes, subtle timing problems can be caused by incremental writing if this rule is not followed. [Section 15.3]

III.6. User-Defined Atomic Objects

• If an atomic object X of type T provides operations O_1 and O_2 , and action A has executed O_1 but not yet committed, then operation O_2 can be performed by a concurrent action B only if O_1 and O_2 commute: given the current state of X, the effect (as described by the sequential specification of 7) of performing O_1 , then O_2 is the same as performing O_2 , then O_1 . "Effect" includes both results returned and the (abstract) state modified.

Rationale: There are two concurrency constraints for user-defined atomic objects:

- An action can observe the effects of other actions only if those actions committed relative to the first action.
- Operations executed by one action cannot invalidate the results of operations executed by a concurrent action.

Two operations (or sequences of operations) that commute in their effect on the abstract state of X may be permitted to run concurrently, even if they do not commute in their effect on the representation of X. This distinction between an abstraction and its implementation is crucial in achieving reasonable performance. [Section 15.4]

 If a user-defined atomic object is accessible from the stable variables of some guardian, it should be written to stable storage whenever an action that modifies it commits to the top.

Rationale: A user-defined atomic type that is not written to stable storage on topaction commit will not be resilient. [Section 15.2]

- The form of the rep for a user-defined atomic type should be one of the following possibilities.
 - 1. The rep is itself atomic. Note that mutex is not an atomic type.
 - 2. The rep is mutex[f] where t is a synchronous type. For example, t could be atomic, or it could be the representation of an atomic type, if the operations on the this fictitious atomic type are coded in-line so that the entire type behaves atomically.
 - The rep is an atomic collection of mutex types containing synchronous types.
 - 4. The rep is a mutable collection of synchronous types, and objects of the representation type are never modified after they are initialized. That is, mutation may be used to create the initial state of such an object, but once this has been done the object must never be modified.

Rationale: In any other case it will be impossible to guarantee the resilience or serializability of the type's objects independently of how they are used. [Section 15.3]

III.7. Subordinate Where Clauses

• A where clause requirement on a cluster as a whole should be used whenever the actual parameters make some difference in the abstraction. For example, in a set cluster, the type parameter's equal operation must be required by the cluster as a whole, in order to preserve type safety and the representation invariant.

Rationale: Argus assumes that requirements that are not placed on the cluster as a whole do not affect the semantics of the abstraction or the representation. [Section 12.6]

Appendix IV Changes from CLU

This appendix lists the changes made to Argus that are not upward compatible with CLU, that is, those which are not merely additions to CLU and that would cause a CLU program to be illegal or to run differently.

IV.1. Exception Handling

Unlike CLU, which propagated unhandled exceptions (by turning them into *failure* exceptions) and gave the *failure* exception special status, unhandled exceptions in Argus are considered errors and always cause a crash of the guardian, and *failure* is not given special status. All exceptions signalled in a procedure, iterator, handler, or creator must be declared in the routine's header, and there are no implicit resignals of *failure* exceptions. See Section 11.6 for details.

IV.2. Type Any

The type any is now a type like any other type, with parameterized routines force, create, and is_type. Thus the CLU manual's notion of "type inclusion" is no ionger necessary (but there is a new notion of type inclusion in Argus, see Section 6.1). The any\$force routine only signals "wrong_type" if the any object's underlying type is not *included* in the type parameter given, but the type of the result of any\$force is its type parameter. The any\$is_type routine returns false if the any object's underlying type is not *included* in the CLU reserved word "force" was eliminated from Argus, and the creation of an any object is never implicit in an assignment in Argus.

IV.3. Built-in Types

Several changes to the interfaces of the built-in types were necessitated by the changes to exception handling. Specifically, the following changes were made to the built-in types.

- 1. The string operations concat, append, s2ac, ac2s, s2sc, and sc2s, can now all signal limits. A string literal that would be too large to represent will not be compiled.
- The sequence operations fill, fill_copy, addh, addl, and concat can now all signal limits. A sequence constructor that would be too large to represent will not be compiled.
- 3. The array (and atomic_array) operations create, predict, set_low, fill, fill_copy, addh, and add/ can now all signal limits. An array constructor that cannot be legally represented will either not be compiled (if this can be detected at compile time) or will signal limits.
- 4. The copy operations of the structured built-in type generators, and the *fill_copy* operations of sequence and array (and atomic_array), allow the copy operations of their type parameters to have a *failure*(string) exception. They will resignal such a *failure* exception. (Note that the type inclusion rule allows a type parameter to be used even if its copy operation does not have exceptions.)
- 5. The similar operations of the built-in structured type generators allow the similar operations of their type parameters to have a failure(string) exception. They will resignal such a failure exception.
- 6. The equal operations of the type generators sequence, struct, and oneof, and the similar1

operations of the type generators array, record, and variant (and their atomic counterparts), allow the equal operation of their type parameters to have a failure(string) exception. They will resignal such a failure exception.

7. The elements iterator and the similar and similar1 procedures of the type generator array (and atomic_array) will raise a failure(string) exception if the array argument is mutated in such a way as to cause a bounds exception when an element is fetched.

IV.4. Type inclusion

Type inclusion (the new notion, see Section 6.1) is used in all contexts, including the decis of except and tagcase statements, where CLU had previously required type equality.

IV.5. Where Clauses

CLU had syntax in the where clause (specifically the production for op__name) that allowed one to require an instantiation of a type parameter's generator. This little used feature has been superseded by the mechanism described in Section 12.6.

IV.6. Uninitialized Variables

An uninitialized variable reference error is defined to cause a crash of the guardian, rather than raising a *failure* exception, which could conceivably be caught.

IV.7. Lexical Changes

Several new reserved words were added. In addition, the semicolon (;) was banished from the syntax.

IV.8. Input/Output Changes

The input/output data types (file_name, stream, and istream) and the library procedures described in appendix III of the CLU manual are not furnished by the Argus system. Our current implementation of Argus provides a *keyboard* cluster for input and a *pstream* cluster for output. In addition, most of the built-in types currently have *print* operations defined, for pretty-printing objects onto pstreams. These I/O mechanisms, however, are still experimental, and so are not documented in this reference manual.

Index

Index 24 \$ 47, 48, 79 % 20, 115 & 53 ⁷ 23 (*) 71 53, 55 +, -, etc. 53 27, 58 --- 17 // 53 **::=**, **|**, **{ }**, **[**] 17 := 39, 58 <, >, etc. 53 = 53 @ 44, 51, 57 [] 26, 58 \ 23 | 53 || 53 53 Abort 8, 10, 60, 61, 69, 72, 88, 97 and exception handling 73 of a remote call action 41 of a subaction 9 qualifier 59, 61, 69, 72 Action 8, 59, 88, 97 abortion versus seize statements 60 activation action 41, 43 ancestors 10 and exception handling 73 call action 41 coenter statement 59 deadlock 13 descendants 10 divisible termination of 60 onler statement 59 8 beteen nested topaction 11, 60 orphan 12, 61 parent of 9 subaction 8 termination 60, 69 topaction 9 See also atomic Activation action 41, 43 Actual argument 40 Actual parameter 80, 81 Ancestor 10 Any 22, 24, 32, 150 versus CLU 159 versus image 32 Argument actual 40 versus parameter 80 Array 25, 52, 130 constructor 26 Assignment 4, 39, 40 and concurrency 39 implicit 39 multiple 39 simple 39 statement 39 type checking for 39 Atomic 3, 8, 97

action 8 built-in atomic types 9, 30, 133, 141, 146 object 9 type 9,97 Atomic_array 30, 52, 133 Atomic record 30, 52, 141 Atomic variant 30, 64, 146 Background 8,89 Bind 48 and equates. 50 and routine equality 49 Block 58 Block structure 36 BNF 17, 107 Body 57 Bool 22, 54, 121 Break 63 **Built-in** atomic types 9, 30 type 22, 119 Built-in type versus CLU 159 Call 4, 40, 41, 44, 50, 51, 57 action 41 by sharing 4, 40 by value 4, 12, 41, 93 oreator 44, 51 expression 50 handler 50 local 40 mensage 43 procedure 50 remote 11, 41, 44, 50, 51, 89 semantics of creator call 44 semantics of remote call 43 statement 57 Call action 41, 43, 44 Cand 54 Catalog 15 Char 23, 125 ecopes 115, 23 Ciosure 48 CLU 3, 11, 21, 24, 73, 159 built in types taken from 22 differences from 159 Cluster 77 Coarm 59 controlling 60 Coenter 59 foreach clause 59 Comment 20, 115 Commit 8, 10, 59, 60, 69, 88, 97 and exception handling 73 committed descendent 10 of a remote call action 41 of a subaction 9 to the top 10 two phase commit protocol 8, 60 Concumuncy 8, 33, 39, 59 Constant 38, 47, 81 Constructor 52 array 26, 52 none for user-defined types 52 record 27, 52

sequence 25, 52 struct 27 structure 52 Continue 63 Controlling coarm 60 Cor 54 Crash 8, 85, 89 and own variables 85 recover code 8 recovery 89 Creator 7, 11, 32, 44, 48, 88, 149 bound 49 equality of bound creators 49 type 149 Creator call 44 as expression 51 as statement 57 semantics of 44 Creatortype 32, 149 Critical section 13, 66 Cvt 78 Data abstraction 7,77 Data type 77 Deadlock 13 Declaration 36, 57, 78 as statement 57 simple 36 with initialization 36 Decode 12, 21, 41, 43, 49, 94 Description unit 15,84 Divisible termination 60 Divisible termination 60 Down 55, 78 DU See also description unit Effects 119 E**lse** 62 Elseif 62 Encode 12, 21, 41, 43, 44, 49, 61, 94 with bind 49 Enter 59 Entity 48 Equate 37, 79 Equate module 34, 79 reference 47 Equated identifier 47 Example key-item table 95 replicated data base 60 spooler guardian 90 Except 70 Exception 41, 69 action termination 73 handler 70 handling 70 name 69 raise 70 result 69 unhandled 73 versus CLU 73, 159 Exit 72 Expression 47 conditional 54 forms of 47 External representation type 12, 94

Failure 11, 42, 43, 44, 73 of communications in a remote call 43 versus CLU 73, 159 See also crash False 22, 121 Fetch 51 Floating point See also real For 62 Force See also any Foreach 59 Fork 58 Formal argument 40, 76 parameter 80 Generator 21, 80 instantiation 81 Get 51 Globel object 3, 7 Guardian 5, 7, 15, 31, 41, 44, 87 background code 89 crash 73 creation 15, 44, 88 definition 87 guardian image 15 interlace 31 lifetime 90 permanence 90 recovery 89 spooler example 90 stable state 87 state 87 temporary 90 termination 67,90 type of 31 versus guardian interface 31 Guidelines 153 Handler 7, 32, 89, 149 bound 49 call 41 equality of bound handlers 49 type 149 See also exception Handlertype 32, 149 Hidden routine 78, 90 Identifier 19 equated 47 See also idn, name Idn 35, 115 versus name 35 If 62 image 12, 21, 32, 93, 150 versus any 32 See also guardian image Immutable 3, 21 Indivisibility 39 Indivisible 21 Input/output 160 versus CLU 160 Instance 81 Instantiate 80 Instantiation 81, 160 type checking of 83 int 22, 121

the second second

Index

Iterator 48, 62, 76, 148 bound 48 equality of bound iterators 49 type 148 Itertype 148 Keyboard 160 Leave 61 Lexicographic order 126, 138, 139, 141 Library 15 Literal 20, 47 char 115 int 115 real 115 string 115 Local 3 call 40,50 object 7 Locking 9, 10, 13, 30 deadlock 13 for built-in atomic types 9 table of locking rules 10 Loop 62 Modifies 119 Module 5, 75, 87 instantiation of 80,81 parameterized 80 Mutable 3, 21 versus atomic 22 Mutex 11, 33, 98, 151 changed operation 99 guidelines 99 multiple 104 sharing 100 Name 35, 115 versus idn 35 Nested action 8 Nested topaction 11, 60 NH 22, 120 Node 34, 44, 120 of guardian creation 44 Null 22, 120 Object 3, 21, 77, 78 abstract 78 as value of expression 47 atomic 3, 21, 97 concrete 78 global 3,7 immutable 3, 21 implementation of 77 local 3,7 mutable 3,21 non-atomic 21 references 3 representation 77 sharing 3, 96, 100 stable 3,7 transmissible 3, 12, 21, 93 transmission of cyclic objects 96 versus variable 3 volatile 7 Oneof 63, 143 Opbinding 81 Operation 77

 $= \sum_{i=1}^{n} \left\{ \left\{ (1+i) +$

-

🕈 - en el ser

indivisibility 21, 119 Operator 20 binary 53 inflx 53 precedence 54 prefix 53 unary 53 Optional parameter 82,84 Ophan 12, 44, 61 Overview 119 Own data 49,85 Own variable 85 and crash recovery 85 Parameter 47,80 actual 81 optional 82 versus argument 80 Parameterization 80 Parameterized type 21, 81 instantiation of 81 Parent 9 Pause 66 Post 119 Pragmatics 153 Pre 119 Precedence 54 Principal argument 30 Print 160 Private reutine 78 Procedure 48, 75, 148 bound 48 ciceure 48 equality of bound procedures 49 type 148 Process 8, 59 See also action Proctype 148 Patream 160 Punctuation token 20 Qualifier abort 50, 61, 69 action, topaction 59 Raise 70 Read lock 9 Reader 30 Real 23, 123 Record 52, 139 constructor 27 Recover code 8, 89 Recoverable 8, 97, 98 Recovery 8, 89, 97 Refer 3 Pleference 34, 47 Flemote call 11, 41, 44, 50, 51, 89 semantics of 43 Replicated database example 60 Representation 77 concrete 78 external 12, 94 Required operation 81 Reserved word 19, 115 Resignal 72 Recijence 97, 98 See also recoverable **Fiestriction** 80, 81

index

Result 47 Return 61 Routine 75, 76, 90 equality 83 See also iterator, procedure RPC See also remote call Rules 153 Scope 35, 78 rules 35 unit 35 Seize 66, 98 Selection of component 51 of element 51 Self 48, 88 Separator 19, 20, 115 Sequence 25, 52, 128 constructor 25 Serializable 8, 9, 67, 97 Set_operation 58 Sharing 3 and mutex 103 and transmission 96 Signal 69 See also exception Spooler guardian 90 Stable object 3,7 state 8,87 storage 8, 97 storage and closures 49 storage recovery 89 variable 3,87 See also resilience Statement 57 abort break 63 abort continue 63 abort leave 61 abort prefix 59 abort resignal 72 abort return 61 abort signal 69 assignment 39 block 58 break 63 coenter 59 component update 58 conditional 62 continue 63 control 57 element update 58 enter 59 except 70 exit 72 for 62 fork 58 if 62 iteration 62 leave 61 pause 66 resignal 72 return 61 seize 66 signal 69 tagcase 63 tagtest 64

tagwait 65 terminate 67 update 58 while 62 yield 62 Store operation 58 String 24, 126 See also char escapes Struct 26, 52, 138 constructor 27 Structure See also struct Subaction 8, 10, 41, 59 Synchronization 39, 97 Synchronous 90 Syntax 107 Table example, transmission of 95 Tagcase 63 Tagtest 64 Tagwait 65 Terminate 67 Termination exceptional 69 of a guardian 67, 90 of a routine 40 Then 62 Token 19, 115 Topaction 9, 59 nested 11 Transmissible 3, 12, 21, 93 object 12 Transmit 21, 41, 78, 84, 93 actual 84 for parameterized modules 94 True 22, 121 Two-phase commit 8, 59, 60, 73 Type 3, 4, 15, 21, 39, 77, 81 actual 81 atomic 9,97 built-in 22, 119 built-in atomic types 9 correctness 4 equality 83 external representation 12,94 generator 21, 80, 81 guardian interface 31 implementation of 77 inclusion 4, 22 of a creator 32, 149 of a guardian 31 of a handler 32, 149 of a iterator 148 of a procedure 148 perameter 34,81 parameterized 9, 21, 80 salety 4 set 80 tranomisaible 12, 21, 93 user-defined 34, 52, 77 versus type actual 82 See also cluster, guardian Type checking 15, 39, 83 of an instantiation 83 Type inclusion 4, 22 versus CLU 160 Type_spec 21

ne Berthe

 $= (-1) \frac{1}{2} \frac{1}{$

Mary

```
Unavailable 11, 42, 43, 44, 59, 60
Unhandled exception 73
   versus CLU 159
Uninitialized variable 36
   versus CLU 160
Up 55,78
Update statement 58
Value 47
Variable 3, 36, 47
   own variable 85
   stable 3, 97
   uninitialized 36
   versus object 3
Variant 63, 144
Version
   of an atomic object 9
Volatile
   object 7
   state 8,87
   variable 87
Where clause 80, 160
   subordinate 82
While 62
Write lock 9
Writer 30
Yield 62
```

This blank page was inserted to preserve pagination.

CS-TR Scanning Project

in war a dawar

ز.-

4

Document Centrol Form	Date : $4/26/96$
Report # Las-Ta- 400	
Each of the following should be identified by a constraint of the following should be identified by a constraint of the state of the st	heckmark:
Artificial Intellegence Laboratory (AI) A Laboratory for Computer Science (LCS)	
Document Type:	
Technical Report (TR) Technical Memo (TM)
Document Information Number	e DOD forms, printer intetructions, etc original pages only.
Originals are:	Intended to be printed as :
Double-sided	
Print type: Typewriter Offset Press Laser Print InkJet Printer Kunknown Other:	· · · · · · · · · · · · · · · · · · ·
Check each if included with document:	
DOD Form ()-) Funding Agent Form Spine Other:	 Cover Page Photo negatives
Page Data:	-
Vii Blank Pages (by page number): 6,14,16,18,46,	56 68 AND PAGES FOLLOWING TITLE PAGE
Photographs/Tonal Material (by page number):	
Other (note description/page number): Description : Page Num <u>LINAGE MAP: (1-175) UN</u> <u>1-1</u> (176-187) S-	NOONTBOL, DOQ(2), TOGTS(3)
Scanning Agent Signoff: Date Received: <u>4196196</u> Date Scanned:	4129196 Date Returned: 512196
Scanning Agent Signature: Michael N., G	Rev 9/94 DS/LCS Document Control Form catriorm.ved

	REPORT DOCU	MENTATION	PAGE		
Ia. REPORT SECURITY CLASSIFICATION		16. RESTRICTIVE	MARKINGS		
Unclassified					
2a. SECURITY CLASSIFICATION AUTHORITY		3 ONTRIUTION / ATRIEADILITY OF REPORT			
2b. DECLASSIFICATION / DOWNGRADING SCHEDULE 4. PERFORMING ORGANIZATION REPORT NUMBER(S) MIT/LCS/TR-400		Approved for public release; distribution is unlimited. 5. MONITORING ORGANIZATION REPORT NUMBER(S) NOO014-83-K-0125			
ic ADDRESS (City State and ZIP Code)		7h ADDRESS (C)	ty State and 7	(P Code)	
545 Technology Square Cambridge, MA 02139	Informatic Arlington,	on Systems VA 22217	Program		
Ba. NAME OF FUNDING/SPONSORING ORGANIZATION DARPA/DOD	8b. OFFICE SYMBOL (If applicable)	9. PROCUREMEN	TINSTRUMENT	IDENTIFICATIO	N NUMBER
ADDRESS (City, State, and 7/P Code)		10. SOURCE OF		ERS	
1400 Wilson Blvd. Arlington, VA 22217		PROGRAM ELEMENT NO.	PROJECT NO.	TASK NO.	WORK UNIT ACCESSION NO.
Argus Reference Manual 2. PERSONAL AUTHOR(S) Liskov, Barb (editor); Scheifler, Robert;	ara; Day, Mark; and Weihl, Will:	Herlihy, Mau Lam	irice; John	son, Paul	; Leavens, Gary
Argus Reference Manual 2. PERSONAL AUTHOR(S) Liskov, Barb (editor); Scheifler, Robert; 3a. TYPE OF REPORT Technical 13b. TIME (FROM	ara; Day, Mark; and Weihl, Will: COVERED TO 18. SUBJECT TERMS Distributed	Herlihy, Mau lam 14. DATE OF REPO 1987 Nove Continue on reven systems, fa	urice; John MRT (Year, Monti amber We <i>if necessary a</i> ult-toleran	son, Paul h, Day) 15 P nd identify by nce, neste	; Leavens, Gary AGE COUNT 165 block number) ed transactions
Argus Reference Manual 2. PERSONAL AUTHOR(S) Liskov, Barb (editor); Scheifler, Robert; 3a. TYPE OF REPORT Technical 13b. TIME OF FROM 5. SUPPLEMENTARY NOTATION 7. COSATI CODES FIELD GROUP SUB-GROUP	ara; Day, Mark; and Weihl, Will: COVERED TOTO 18. SUBJECT TERMS (Distributed concurrency	Herlihy, Mau lam 14. DATE OF REPO 1987 Nove Continue on reven systems, fa , concurrenc	wif necessary and ult-tolerany control,	son, Paul h. Day) 15 P nd identify by nce, neste locking,	; Leavens, Gary AGE COUNT 165 block number) ed transactions persistent
Argus Reference Manual 2. PERSONAL AUTHOR(S) Liskov, Barb (editor); Scheifler, Robert; 3a. TYPE OF REPORT Technical 13b. TIME (FROM	ara; Day, Mark; and Weihl, Will: COVERED TO	Herlihy, Mau lam 14. DATE OF REPO 1987 Nove Continue on revers systems, fa , concurrenc stract data	wif necessary and ult-tolerany y control, types, atom	son, Paul h. Day) 15 P nd identify by nce, neste locking, nic_object	; Leavens, Gary AGE COUNT 165 block number) ed transactions persistent s, remote (con
Argus Reference Manual 2. PERSONAL AUTHOR(S) Liskov, Barb (editor): Scheifler, Robert; 3a. TYPE OF REPORT Technical 13b. TIME (FROM	18. SUBJECT TERMS (Distributed concurrency storage. ab and identify by block al language/syst ograms. Argus i efit from being e applications: urrency and hard are not severe. ing systems. It is largely us provides proc d clusters for d and control acce a convenient way at a single modu	Herlihy, Mau 14. DATE OF REPO 1987 Nove Continue on reven systems, fa concurrence stract data number) em designed s intended t implemented they make us ware failure Examples of an extension edures for p lata abstract ess to one or to refer to an defin	wif recessory and amber wif recessory and ult-toleran y control, types, atom to support by a distri- e of on-lin s, and they such appl: of CLU, bu rocedural a ion. In action constants e a class of	son, Paul h, Day) 15 P nd identify by nce, neste locking, mic object the const only a sub ibuted pro- ne data the y provide ications a ut there a abstraction ddition, A urces. An . As in Co	AGE COUNT 165 block number) ad transactions persistent is, remote (con cruction and oset of the ogram. Two nat must remain services under are office are number of on, iterators argus provides igus also LU, modules i abstractions.
Argus Reference Manual 22. PERSONAL AUTHOR(S) Liskov, Barb (editor): Scheifler, Robert; 13a. TYPE OF REPORT Technical 13b. TIME (FROM	ara; Day, Mark; and Weihl, Will: IN IOVERED IO IO IN SUBJECT TERMS (Distributed concurrency storage. ab and identify by block of al language/syst ograms. Argus i efit from being e applications: urrency and hard are not severe. ing systems. It is largely us provides proof and control acce a convenient way at a single modu RPT. DTIC USERS	Herlihy, Mau I. DATE OF REPO 1987 Nove Continue on reven systems, fa concurrence stract data number) em designed s intended t implemented they make us ware failure Examples of an extension edures for p lata abstract ess to one or to refer to cle can defin 21. ABSTRACT SE Unclassif	wif necessary and ult-toleran y control, types. atom to support of by a distri- e of on-lin s, and they such appl: of CLU, by rocedural a ion. In action constants a class of CURITY CLASSIFI	son, Paul h, Day) 15 P nd identify by nce, neste locking, nic object the const ibuted pro- ne data the y provide ications a ut there a abstraction ddition, A urces. Ar . As in C of related	AGE COUNT 165 block number) ad transactions persistent is, remote (con cruction and oset of the ogram. Two nat must remain services under are office are number of on, iterators argus provides igus also ILU, modules i abstractions.

1

1.1.1

-

ţ

 $\tau = - \frac{1}{2} \sum_{i=1}^{n-1} \frac{e^{i \pi i x_i} e^{i \pi i x_i}}{e^{i \pi i x_i} e^{i \pi i x_i}}$

. .

6

18. procedure call, orphans, exception handling.